Enesia Banda Chaponda

The epidemiology of malaria, curable sexually transmitted and reproductive tract infections and their coinfection among pregnant women in a catchment area in Nchelenge District, Zambia

2017

London school of Hygiene and Tropical Medicine, University of London

Disease Control Department Faculty of Infectious and Tropical Diseases

Funded by International Centres of Excellence for Malaria Research and the Southern Africa Consortium for Research Excellence

Declaration by Candidate

I Enesia Banda Chaponda confirm that the work presented in this thesis is my own. Where information has been derived from other sources I confirm that this has been indicated in the thesis.

Signed:----- Date: ------

Full Name: Enesia Banda Chaponda

Table of contents

Declaration by Candidate	2
Table of contents	3
List of Figures	8
List of Tables	9
Scope of work conducted by the PhD candidate	10
Abstract	12
Abbreviations	15
Acknowledgement	16
Summary of content	18
1 Introduction	20
1.1 Background	21
1.2 Risks associated with malaria and curable STIs/RTIs in pregnancy	22
1.2.1 Effects of curable STIs/RTIs on pregnancy outcomes	22
1.2.2 Additional evidence for association of STIs/RTIs with ABO	25
1.2.3 Effects of malaria on pregnancy outcomes	29
1.2.4 Effect of treatment of curable STIs/RTIs in pregnancy	35
1.3 Prevalence of malaria and STIs/RTIs	37
1.3.1 General overview of malaria prevalence in pregnancy	37
1.3.2 General overview of STI/RTI prevalence	
1.4 Prevalence of malaria and STIs/RTIs in pregnancy in Zambia and countries	neighbouring 39
1.4.1 Prevalence of malaria in pregnancy in Zambia and neighbouring coun	tries41
1.4.2 Prevalence of curable STIs/RTIs in pregnancy in Zambia and countries	neighbouring 45
1.4.3 General assessment of prevalence literature	48
1.5 Risk factors for STI/RTI and malaria	50
1.5.1 Risk factors for STIs/RTIs	50
1.5.2 Risk factors for Malaria infection	51
1.5.3 Conceptual frameworks for infection	51
1.6 Prevalence of malaria and STIs/RTIs coinfection	55
1.7 Current antenatal Care package	55
1.7.1 Antenatal care package for malaria in pregnancy	55
1.7.2 Antenatal care package for curable STIs/RTIs in pregnancy	56

1.7.3 Antenatal care in Zambia	57
1.8 Birthing practices and prevalence of adverse pregnancy outcome in Zambia	58
1.8.1 Practices and beliefs about pregnancy and childbirth in Zambia	58
1.8.2 Prevalence of adverse birth outcomes in Zambia	61
1.9 Rationale and objectives	63
1.9.1 Rationale	63
1.9.2 Objectives	64
2 Methods	65
2.1 Chapter introduction	66
2.1.1 Study design and sample size	66
2.1.2 Study design	66
2.1.3 Sample size calculation	68
2.2 Study area and sampling	70
2.2.1 Study area	70
2.2.2 Sampling	73
2.3 Inclusion and exclusion criteria	73
2.3.1 Inclusion criteria	73
2.3.2 Exclusion criteria	73
2.4 Enrolment and antenatal care package	73
2.5 Data and sample collection	75
2.5.1 Data and sample collection at recruitment	75
2.5.2 Collection of global position system coordinates	75
2.5.3 Follow-up at day 28	76
2.5.4 Data and sample collection at delivery	76
2.6 Laboratory testing and quality control	77
2.6.1 Syphilis testing and treatment of seropositives	77
2.6.2 Malaria diagnosis by microscopy	78
2.6.3 Diagnosis of placental malaria by histology	78
2.6.4 Diagnosis of bacterial vaginosis by microscopy	79
2.6.5 Nucleic acid amplification	79
2.6.6 Quality control of PCR methods	82
2.6.7 Quality assurance of data and sample collection	83
2.7 Ethical consideration	83
2.8 Data Processing and Analysis	85
3 Results (1): Maternal characteristics and prevalence estimates of malaria, STIs/RTI their coinfection	s and 88

3.1 Chapter introduction	89
3.2 Participation distribution	89
3.3 Socio-demographic characteristics of participants	91
3.4 Malaria and HIV Burden	95
3.4.1 The Prevalence of HIV infection	95
3.4.2 The prevalence of peripheral malaria	95
3.4.3 Placental malaria infection	96
3.4.4 Monthly peripheral malaria infection among first ANC attendees	98
3.5 Reproductive tract and curable sexually transmitted infections	99
3.6 Malaria and curable STI/RTI coinfection	101
3.6.1 Malaria and STI/RTI coinfection according to parity	101
3.7 Quality control assessment results	106
4 Results (2): Risk factors for malaria, STI/RTI and their coinfection	107
4.1 Chapter introduction	108
4.2 Risk factors for malaria infection	108
4.3 Risk factors for curable STIs/RTIs	111
4.3.1 Risk factors for STIs	111
4.3.2 Risk factors for BV	111
4.4 Risk factors for malaria and curable STI/RTI coinfection	116
5 Results (3): The prevalence of DHFR and DHPS mutations associated with S and <i>in vivo</i> efficacy, parasitological failure and prophylactic failure of SP Zambian women	SP resistance in pregnant 119
5.1 Chapter introduction	120
5.2 The prevalence of DHFR and DHPS mutations associated with SP resistanc	e120
5.3 Therapeutic efficacy of IPTp-SP over a 28-day period	124
5.4 Classification of parasitological failure cases based on MSP-2 genotyping	124
6 Results (4) The prevalence of adverse birth outcomes among delivering part predictors of adverse birth outcomes	ticipants and 128
6.1 Chapter introduction	129
6.2 Maternal and infant characteristics	129
6.2.1 Baseline characteristics of delivering participants	129
6.2.2 Characteristics of delivering women and infants	132
6.3 Prevalence of adverse birth outcomes	136
6.3.1 Prevalence of adverse birth outcome according to parity	136
6.3.2 Prevalence of adverse birth outcome among infected and uninfected	women138
6.4 Placental malaria burden	140

6.4.1 Frequency of adverse birth outcome in placental malaria histopathology classes
6.5 Prevalence of adverse birth outcome according to antimalarial and antibiotic treatment
6.6 Protective and risk factors for composite adverse birth outcomes146
6.7 Risk factors for PD, LBW and IUGR151
7 Discussion157
7.1 Losses to follow-up158
7.2 Prevalence of malaria, STI/RTI and their coinfection159
7.2.1 The malaria burden159
7.2.2 Prevalence of curable STIs/RTIs162
7.2.3 Prevalence of malaria and STI/RTI coinfection165
7.3 Risk factors for malaria, curable STIs/RTIs and their coinfection
7.3.1 Risk factors for malaria infection165
7.3.2 Risk factors for curable STIs/RTIs infection166
7.3.3 Risk factors for malaria and STI/RTI coinfection167
7.4 Prevalence of point mutations associated with parasite resistance to SP168
7.5 Therapeutic and prophylactic failure of sulphadoxine-pyrimethamine169
7.6 Characteristics of delivering women, Prevalence of adverse birth outcomes and risk factors for adverse birth outcomes
7.6.1 Characteristics of women according to follow-up170
7.6.2 Prevalence of adverse birth outcomes170
7.6.3 Risk factors for adverse birth outcome171
7.7 Potential study biases, limitations, Conclusions and recommendations177
7.7.1 Potential study biases and limitations177
7.7.2 Conclusions
7.7.3 Recommendations180
7. 8 Future work
7.9 Reflexivity
8.0 References
9.0 Appendices217
Appendix 1.1a: Search terms for studies reporting malaria and/or STIs/RTIs and birth outcomes217
Appendix 1.1b: Search terms for studies reporting point prevalence for malaria and STIs/RTIs219
Appendix 1.2a: Peripheral malaria point prevalence estimates among pregnant women in
Zambia and surrounding countries220

Appendix 1.2b: Placental malaria point prevalence estimates among pregnant women in Zambia and surrounding countries
Appendix 1.2c: Syphilis point prevalence estimates among pregnant women in Zambia and surrounding countries
Appendix 1.2d: <i>Chlamydia trachomatis</i> point prevalence estimates among pregnant women in Zambia and surrounding countries
Appendix 1.2e: <i>Neisseria gonorrhoeae</i> point prevalence estimates among pregnant women in Zambia and surrounding countries
Appendix 1.2f: <i>Trichomonas vaginalis</i> point prevalence estimates among pregnant women in Zambia and surrounding countries
Appendix 1.2g: Bacterial vaginosis point prevalence estimates among pregnant women in Zambia and surrounding countries
Appendix 1.3: Diagnostic Methods and published Sensitivity and Specificity Measures228
Appendix 1.4: Risk factors for malaria and STIs/RTIs229
Appendix 2.1: Information Sheet237
Appendix 2.2: Eligibility screening form
Appendix 2.3: Consent Form241
Appendix 2.4: Questionnaire242
Appendix 2.5: Study completion or drop out record form248
Appendix 2.6: Laboratory Record/Reporting forms249
F1: Blood smear record form249
F2: Record form for Chlamydia, trichomoniasis and gonorrhoea test results249
F3: Record form for bacterial vaginosis diagnosis249
F4: Placental Histology Results Reporting Form249
Appendix 2.7: Names and sequences of primers that were used for detection of chlamydia, gonorrhoea, trichomoniasis and malaria infection250
Appendix 2.8: Polymerase chain reaction parameters
Appendix 2.9: Primer names and sequences for detection of DHFR and DHPS mutations
Appendix 2.10: Restriction fragment length polymorphisms analysis of DHFR and DHPS polymorphisms253
Appendix 2.11: Sequences of oligonucleotide primers that were employed for the differentiation of recrudescence from re-infection
Appendix 2.12: Letter of notification of Ethical Clearance

List of Figures

Figure 1. 1: Identification and screening of studies	26
Figure 1. 2: Flow chart	41
Figure 1. 3: Conceptual framework for STIs/RTIs	52
Figure 1. 4: Conceptual framework for malaria infection	52
Figure 1. 5: Conceptual frame work for malaria and STI/RTI coinfection	54
Figure 1. 6: Conceptual framework for adverse birth outcomes	62
Figure 2. 1 Flow-chart of research activities	67
Figure 2. 2: Map showing the position of Nchelenge district in Zambia and the study ar	[.] ea72
Figure 3. 1: Participation flow-chart	90
Figure 3. 2: Prevalence of malaria in pregnant women of Nchelenge District and total	
monthly rainfall during the study recruitment period	98
Figure 5. 1: Prevalence of P. falciparum DHFR mutations in pregnant women of Nchele	enge
district	120
Figure 5. 2: Prevalence of <i>P. falciparum</i> DHPS point mutations in pregnant women of	
Nchelenge district	121
Figure 5. 3: Classification of recurrent parasitaemia among participating women at day	/ 28
post-IPTp-SP in reference to day 0.	127

List of Tables

Table1- 1: Summary of studies associating curable STIs/RTIs and malaria with adverse
pregnancy outcomes31
Table 2-1: Sample size calculation for determination of risk factors for adverse birth
outcomes69
Table 3-1: Baseline characteristics of first ANC attendees of Nchelenge District, Zambia,
according to recruitment site92
Table 3- 2: Prevalence of HIV and <i>Plasmodium falciparum</i> malaria infection at first antenatal
attendance and at delivery among women of Nchelenge district according to
gravidity/parity97
Table 3- 3: Prevalence of curable STIs/RTIs among first ANC attendees of Nchelenge district
according to HIV status100
Table 3- 4: Prevalence of malaria and STI/RTI coinfection according to gravidity102
Table 3- 5: Prevalence of individual curable STI/RTI and malaria coinfection among first
antenatal care attendees of Nchelenge District, Zambia104
Table 4- 1: Predictors of malaria infection in pregnant women of Nchelenge District109
Table 4- 2: Univariate and multivariate analyses of risk factors for STIs
Table 4- 3: Univariate and multivariate analyses of risk factors for BV
Table 4- 4: Risk factors for malaria and curable STI/RTI coinfection in pregnant women of
Nchelenge District117
Table 5- 1: Antifolate multiple mutants among first ANC attendees of Nchelenge District,
Zambia122
Table 5- 2: Results of DHFR and DHPS genotyping of samples from 96 first antenatal care
attendees of Nchelenge District123
Table 5- 3: Frequency of prophylactic and treatment failure of sulphadoxine-pyrimethamine
among HIV negative first ANC attendees of Nchelenge District, Zambia

Scope of work conducted by the PhD candidate

This thesis is the work of the principle investigator, Enesia Banda Chaponda. This investigator designed the study with the help of the main and the co-supervisors. All the standard operation procedure, the questionnaire and all study forms were developed by this investigator. The translation of the questionnaire and consent forms from English to Bemba and back translation was done by personnel from the Zambia National Broadcasting Network with input from the principle investigator who ensured that what was communicated in the local language, Bemba, was what was intended. This investigator applied for ethical clearance from the London school of Hygiene and Tropical Medicine Observational/Interventions Research Ethics committee.

After obtaining ethical clearance this investigator also sought and obtained written permission to conduct the study from the Zambia Ministry of Health. Upon being granted permission by the ministry of health this investigator wrote letters (and personally delivered them) to the district health office, the office of the district commissioner, the administrators of St. Paul's Mission Hospital (only hospital in Nchelenge District) and administrators of the two health centres (recruitment sites for the study) to inform them about the study before commencement. This investigator paid a courtesy call on the chief in the area and organised and held prestudy meetings with community leaders (village headmen). With the help of members of the Tropical Disease Research centre study team members who were conducting research in the area, this investigator recruited her study team members, which included 4 full-time team members and those who were already working in the health system. Those who were already in the system were recruited to help out with the project activities during routine duty. These included midwives, general nurses, traditional birth attendants/counsellors, counsellors who were involved in the routine running of the ANC clinic in the two the health centres, a radiographer and laboratory technicians.

This investigator also trained team members on the general aspects of the project and specifics of their role in the project. This investigator managed the day to day running of the project, coordinated study team members and conducted weekly

feedback meetings with team members. This investigator was also involved in some of study activities including screening eligibility of participants, obtaining written consent and administration of questionnaires while coordinating the rest of the activities during the enrolment and sample collection process.

All the molecular work was conducted by this investigator. Molecular work included deoxy-ribonucleic acid extraction from dried blood samples and cervico-vaginal swabs for diagnosis of malaria and three sexually transmitted infections and diagnoses of malaria, Chlamydia, gonorrhoea and trichomoniasis infection by standard polymerase chain reaction. Over 75% of slides for diagnosis of bacterial vaginosis were stained by the principle investigator each week during the recruitment period. At the beginning of the project, slide staining for diagnosis of bacterial vaginosis was carried out by a technician who later left for school. Reading of slides for diagnosis of bacterial vaginosis of bacterial vaginosis was done by this investigator at the University of Zambia at intervals. Data was mainly entered by a data clerk; this investigator coordinated data entry, occasionally entered data and conducted regular data entry checks. Data cleaning, development of the analysis plan, data processing, data analysis and writing up of the thesis were all done by this investigator.

Activities coordinated by the candidate but conducted by hired individuals

Some study activities were conducted by individuals who were part of the team or those hired for a specific function. These included preparation of blood slides for diagnosis of malaria and slides reading; processing of placental tissue, slide preparation and reading of slides for diagnosis of placental malaria; rapid plasma reagin (RPR) and Treponema pallidum haemagglutination (TPHA) assay testing and gestational age measurement by ultra-sound. All of these functions were conducted by individuals with training and experience in performing these procedures.

Abstract

Introduction

Malaria and curable sexually transmitted and reproductive tract infections (STIs/RTIs) are important causes of adverse birth outcomes (ABO) and are both prevalent in most parts of sub-Saharan Africa. From a public health perspective, control of these infections requires interventions that are part of an integrated antenatal care package. The extent to which there may be coinfection increases the importance of such an integrated approach to reduce ABO.

A systematic review and meta-analysis published in 2012 showed that the prevalence of malaria and curable STIs/RTIs among antenatal attendees in sub-Saharan Africa is considerable. However, the prevalence of malaria and curable STI/RTI coinfection has not been reported in any epidemiological setting.

The primary objective of this thesis is to address this knowledge gap by estimating the prevalence of malaria, curable STIs/RTIs and their coinfection and to highlight the importance of an integrated approach to control malaria and STIs/RTIs in pregnancy. Secondary objectives of the study were to:

(1) determine risk factors for malaria, curable STIs/RTIs and their coinfection;

(2) estimate the prevalence of ABO and identify risk factors for ABO;

(3) measure the *in vivo* efficacy and the prophylactic effect of sulphadoxinepyrimethamine (SP) in pregnant women, and

(4) characterise the molecular markers associated with parasite resistance to SP among pregnant women.

Methods

A prospective cohort study of 1,086 antenatal attendees was conducted in Nchelenge District, Zambia. Consenting women visiting two health centres for their first antenatal care (ANC) visit were screened for malaria and curable STIs/RTIs (Chlamydia, gonorrhoea, trichomoniasis, bacterial vaginosis [BV] and syphilis). Sociodemographic data and maternal characteristics were also collected at enrolment. Sulphadoxine-pyrimethamine was administrated as intermittent preventive treatment to eligible women and they were followed up at day 28 for a second

malaria screening to determine the therapeutic and prophylactic failure of SP. At delivery participants were screened for placental malaria and data on birth outcomes were recorded. Univariate and multivariate analyses were conducted to determine the association between the potential risk factors for infection and ABO.

Results

Of the 1086 women recruited 729 were successfully followed to delivery. The prevalence of malaria infection measured by microscopy was 31.8% (95% CI, 29.1-34.6) and by PCR was 57.8% (95% CI, 54.9-60.8). The risk of malaria infection was higher among pregnant women recruited from Nchelenge health centre compared to those attending the Kashikishi health centre (adjusted odds ratio [aOR] = 1.81; 95% CI, 1.38-2.37, P < 0.001), and HIV-infected women across health centres had a greater risk of malaria infection compared to HIV-uninfected women (aOR = 1.46; 95%, 1.00-2.13, P = 0.045).

Infection with at least one STI/RTI was observed in 64.8% (95% CI, 61-67.4) of the participating women. With the exclusion of BV the prevalence of infection with at least one curable STI was 34.5% (95% CI, 31.7-37.4). Infection with at least one STI was associated with BV. In comparison to uninfected women, women infected with BV were at a higher risk of being infected with at least one STI (aOR 1.44; 95% CI, 1.08-1.92, P = 0.012). HIV-infected women had a higher risk of infection with BV than HIV-uninfected women (aOR 1.87; 95% CI, 1.24-2.83, P = 0.003) and women infected with at least one STI had a higher risk of BV (aOR 1.40; 95% CI (1.07 -1.84, P = 0.01).

Among participants with complete results (n=1071), 38.7% (95% CI,35.7-41.6) were coinfected with malaria parasites and at least one STI/RTI; 18.9% (95% CI, 16.5-21.2) were infected with malaria parasites only; 26.0% (95% CI, 23.5-28.8) were infected with at least one STI/RTI but no malaria parasites, and 16.4% (95% CI, 14.1-18.6) had no infection. The risk of malaria and curable STI/RTI coinfection was higher among HIV infected women than HIV-uninfected women (OR; 3.59 [95% CI, 1.73-7.48], P < 0.001).

The prevalence of composite ABO was 35.1%. Women shorter than 1.5m were at a higher risk of experiencing at least one ABO (aOR 1.55; 95% Cl, 1.10-2.18, P = 0.02). The risk of having ABO among para II was less than half of the risk observed in

primiparous women (aOR 0.41; 95% CI, 0.27-0.61, P < 0.001) and much lower among multiparous women (aOR 0.32; 95% CI, 0.22-0.48, P < 0.001). Having taken two or more doses of SP during pregnancy was protective against ABO (aOR 0.47; 95% CI, 0.31-0.72, P = 0.001). None of the infections (malaria, curable STIs/RTIs and their coinfection) diagnosed at first ANC were associated with ABO.

The prevalence of highly resistant quintuple mutant was 68.8% among first ANC attendees. Despite the moderate prevalence of the quintuple mutant among pregnant women, SP cleared parasitaemia in 86% of the asymptomatic malaria cases among HIV-negative women

Conclusion

The prevalence of malaria, STI/RTI and their coinfection at first ANC in this study population was considerable. However, no association was found between ABO and infection with malaria or STI/RTI or their coinfection. This lack of association is partially a result of interventions within the ANC package including treatment of some STI/RTI, intermittent preventive treatment in pregnancy with SP and iron and folic acid supplementation. Sulphadoxine-pyrimethamine retains partial efficacy against *P. falciparum* malaria in this area with moderate prevalence of the quintuple mutant. While continuing the policy of offering intermittent preventive treatment between the against both malaria and STIs/RTIs needs to be considered.

Abbreviations

ABO	Adverse birth outcome
ANC	Antenatal clinic
BV	Bacterial vaginosis
CI	Confidence interval
DHFR	Dihydrofolate reductase
DHPS	Dihydropteroate synthase
Hb	Haemoglobin
HIV	Human immunodeficiency virus
HPTN	HIV Prevention Trials Network
ID	Identification (number)
ITN	Insecticide treated net
IUGR	Intra-uterine growth retardation
IPTp-SP	Intermittent preventive treatment in pregnancy with sulphadoxine-
	pyrimethamine
LBW	Low birth weight
MSP	Merozoite surface protein
OR	Odds ratio
PD	Preterm delivery
PCR	Polymerase chain reaction
RPR	Rapid plasma reagin
RR	Risk ratio
RTI	Reproductive tract infection
SP	Sulphadoxine-pyrimethamine
STI	Sexually transmitted infection
STI/RTI	Sexually transmitted and reproductive tract infection
ТРНА	Triponema pallidum Haemagglutination assay
ZDHS	Zambia Demographic and Health Survey
ZEPRS	Zambia Electronic Perinatal Record System

Acknowledgement

I would like to thank my supervisor Professor Daniel Chandramohan for the challenges and sound advice throughout the programme. Thank you for being there whenever I needed direction. I thank the co-supervisors, Matthew Chico, Professor Charles Michelo and Dr. Bellington Vwalika. You made my journey interesting and much more bearable. I would like to thank Professor Philippe Mayaud for being part of my advisory panel. I would like to express my gratitude to Jane Bruce for offering guidance with the data analysis. Many thanks to the International Centres of Excellence for Malaria Research (ICEMR) in Southern Africa, and the US National Institutes of Health for the financial support from NIH/NIAID grant U19AI089680 (Malaria Transmission And The Impact Of Control Efforts In Southern Africa) with John Hopkins University Bloomberg School of Public Health and the University Of Zambia School Of Medicine (UNZA-SoM). I would also like to thank my ICEMR mentor Dr. Sungano Mharakurwa for laboratory technical support. I am grateful for the financial support I received from the Research Support Centre at the UNZA-SoM through the Southern African Consortium for Research Excellence (SACORE), which is part of the African Institutions Initiative grant of the Wellcome Trust (company no. 2711000), a charity (no. 210183) registered in England, as well as the Commonwealth Scholarship Commission in the UK. Many thanks go to the administrators of the National Malaria Control Centre for the provision of some reagents and consumables.

I am grateful to my research assistants James Lupiya, Joseph Mpundu, Doreen Muyayi, Vincent Chishimba, Martha Chibwe, Beauty Mwenya and Delgracious Makwaya. I would also like to thank the midwives who were working at St. Paul's Hospital labour ward and at Kashikishi and Kabwali health centres for their assistance in collection of samples at delivery. Many thanks to the Nchelenge District Health Management team, community leaders and administrators of Kashikishi and Nchelenge health Centres for their warmth and support to the project. I am grateful to the participants and the ANC staff at Nchelenge and Kashikishi health centres.

I appreciate the professionalism of the management and staff of the National Sexually Transmitted Infections Reference Laboratory and the Pathology and Microbiology Department at the University Teaching Hospital. I would also like to thank Professor Paul Kelly and the Tropical Gastroenterology and Nutrition Group Laboratory management for granting usage of their laboratory for the molecular work.

Many thanks to my dear friend, Kangwa Chintu, for her prayers and constant encouragement. Many thanks to Anita Ngulube for taking excellent care of my children, Mphatso Ngulube and Israel Munashe Ngulube, in my absence. To my husband, Joseph, I would like say thank you for taking care of the family during my time away from home and for being a pillar of strength, you are simply the best. And last but not least, I will always be thankful to God Almighty, besides whom there is no other, for His provision, for connecting me to many wonderful people and seeing me through the challenges that I have faced in my life and education.

Summary of content

This thesis aims to address a knowledge gap by measuring malaria and STIs/RTIs and their coinfection. Many studies have reported the prevalence of malaria and that of STI/RTI in pregnant women but the prevalence of their coinfection has not been estimated in an epidemiological setting. Secondary objectives were to: identify risk factors for malaria, curable STIs/RTIs and their coinfection; estimate the prevalence of ABO and identify risk factors; characterise the mutations that are associated with parasite resistance to SP and assess the *in vivo* efficacy, prophylactic and therapeutic failure of SP among pregnant women.

The thesis is organised as follows:

- Chapter 1 gives the background to the study by highlighting the burden of malaria, curable STIs/RTIs in pregnant women and their effect on birth outcome and the antenatal care package for their management. The background also highlights the gap in knowledge on malaria and curable STI/RTI coinfection and points out the weaknesses of the intermittent preventive treatment in pregnancy with SP (IPTp-SP) strategy for the alleviation of the consequences of malaria infection in pregnancy in the face of emerging parasite resistance to SP. The background also points out the inaccuracy of the syndromic approach for the management of STIs/RTIs. The chapter also gives the rationale for the study and the objectives.
- Chapter 2 describes the methods employed including the study design, a description of the study area, identification and recruitment of participant; and tools used in the collection of data and samples, detection of infections, data processing and analyses.

- Chapter 3 describes the characteristics of study participants according to recruitment sites and the prevalence of malaria, curable STIs/RTIs and their coinfection among participants.
- Chapter 4 describes the risk factors for malaria, curable STIs/RTIs and their coinfection in the study population.
- Chapter 5 describes results on the estimates of dihydrofolate reductase (DHFR) and dihydropteroate synthase (DHPS) mutations associated with *P*. *falciparum* resistance to SP and gives estimates of the *in vivo* efficacy and the therapeutic and prophylactic failure of IPTp-SP. The idea behind this chapter is to give details of interest that is give an estimation of the level of SP resistance and the efficacy of SP in the study population.
- Chapter 6 gives estimates of individual and composite adverse birth outcome and describes risk factors for ABO.
- Chapter 7 discusses all the results and gives the limitations of the study,

conclusions, policy recommendations and reflexivity.

Note

1. The data on malaria prevalence and risk factors for infection have been published in the *Malaria Journal (2015)* 14:380 DOI 10.1186/s12936-015-0866-1

Chaponda EB, Chandramohan D, Michelo C, Mharakurwa S, Chipeta J, Chico RM, 2015. High burden of malaria infection in pregnant women in a rural district of Zambia: a cross-sectional study. Malar J 14: 380.

2. The data on malaria and STI/RTI coinfection has been published in the American Journal of Tropical Medicine and Hygiene.

Chaponda EB, Chico RM, Bruce J, Michelo C, Vwalika B, Mharakurwa S, et al. Malarial Infection and Curable Sexually Transmitted and Reproductive Tract Infections Among Pregnant Women in a Rural District of Zambia. Am J Trop Med Hyg. 2016;95(5):1069-76.

1 Introduction

1.1 Background

Maternal malaria and curable STIs/RTIs remain important causes of adverse birth outcomes (ABO). One of the interventions recommended by the World Health Organisation (WHO) for reducing the burden of malaria in pregnancy is intermittent preventive treatment in pregnancy with sulphadoxine-pyrimethamine (IPTp-SP). However, the effectiveness of IPTp-SP has been undermined by the emergence of SP resistant parasites [1, 2]. Results from a systematic review and meta-analysis of trials showed that administration of three doses of IPTp-SP appear to be beneficial even in areas with high SP resistance. However, this does not imply that it may be so in areas with very high SP resistance [3]. Nevertheless, there is urgent need to identify alternative preventive therapy for SP.

Testing women for curable STIs/RTIs during ANC visits and providing appropriate care has been a public health challenge for decades in resource-poor settings. The WHO has developed syndrome-based algorithms for the detection of curable STIs/RTIs to assist countries with limited resources [4]. However, syndrome-based algorithm for vaginal discharge has poor sensitivity in detecting cervical infections [5]. Therefore, a considerable burden of STIs/RTIs in pregnancy remains undetected and untreated.

A systematic review and meta-analysis of studies reporting malaria or curable STIs/RTIs in pregnant women attending antenatal clinics in sub-Saharan Africa published in 2012 showed that the prevalence of malaria and curable STIs/RTIs during pregnancy is considerable [6]. The review also highlighted the paucity of data on coinfection of malaria and curable STIs/RTIs during pregnancy. In view of the challenges in the current approach, one of the possible ways forward is to find an initiative that integrates the prevention and control of malaria and STI/RTI in pregnancy. However, initiative for an integrated solution is hampered by a lack of information about the burden of their coinfection. This thesis aims to address this gap in knowledge and to highlight the importance of an integrated approach to control malaria and STIs/RTIs in pregnancy.

Malaria and curable STIs/RTIs are both prevalent in most parts of sub-Saharan Africa. The overlapping prevalence of malaria and curable STIs/RTIs has important public health implications especially in pregnant women.

Despite the fact that malaria and curable STIs/RTIs are important risk factors for ABO, no study identified in the systematic review by Chico *et al.* had reported the prevalence of malaria and STIs/RTIs coinfection, with the exception of one study which reported that 48.3% of women who tested positive to rapid plasma reagin (RPR) had placental malaria [6].

The primary objective of the current study was to estimate the prevalence of malaria, STIs/RTIs, and their coinfection among pregnant women in an area with overlapping prevalence of malaria and STIs/RTIs. Secondary objectives were to:

- 1. determine risk factors for malaria, curable STIs/RTIs and their coinfection;
- 2. estimate the prevalence of ABO and identify risk factors for ABO;
- 3. measure the *in vivo* efficacy and the prophylactic effect of sulphadoxinepyrimethamine (SP) in pregnant women, and
- characterise the molecular markers associated with parasite resistance to SP among pregnant women.

1.2 Risks associated with malaria and curable STIs/RTIs in pregnancy

1.2.1 Effects of curable STIs/RTIs on pregnancy outcomes

Curable STIs/RTIs are a major public health problem, especially in developing countries [7, 8]. In 2004 Mullick and colleagues reviewed the prevalence of STIs/RTIs and the relationship between these infections and pregnancy outcomes [7]. The paper reviewed evidence from both developed and developing countries. Although there were few studies from developing countries at the time of the review, the authors concluded that developing countries have a high prevalence of STIs/RTIs which have a significant impact on birth outcomes.

1.2.1.1 Syphilis

Mullick *et al.* identified two studies that looked at the impact of syphilis in pregnancy at a population level in Malawi and Tanzania [9, 10]. In Malawi women with active

syphilis were 11 times more likely to experience a stillbirth (adjusted odds ratio [aOR] 10.89; 95% CI, 6.61-17.93) and 26% of stillbirths were attributed to active syphilis [9]. In Tanzania, 17% of all adverse pregnancy outcomes were attributed to syphilis. Syphilis was associated with increased risk of stillbirth (adjusted risk ratio [aRR] 18.1; 95% CI, 5.5-59.60), low birth weight (LBW) in live born infants (aRR 3.3; 95% CI, 2.0-5.4), preterm delivery (PD) (aRR 6.1; 95% CI, 2.5-15.3) and intra-uterine growth retardation (IUGR) (aRR 2.1; 95% CI, 1.0-4.2) [10].

1.2.1.2 Chlamydia trachomatis

Two prospective studies from the United States (US) included in the same review showed that Chlamydia trachomatis has significant impact on pregnancy outcomes. In the first study researchers found that *C. trachomatis* infection was associated with both IUGR (aOR 2.4; 95% CI, 1.3-4.2) and PD (aOR 1.6; 95% CI, 1.0-4.2) [11]. Authors of the second study found that infection with C. trachomatis was associated with LBW (aOR 2.7; 95% CI, 1.3 to 5.7), premature rupture of membranes (PROM) (OR 2.4; 95% CI 1.7 to 5.4) and < 34 weeks preterm labour (aOR 4.0; 95% CI, 1.7-9.2) [12]. However, a third study (also done in the US) showed no such association between infection with C. trachomatis and ABO [13]. This lack of association could be explained by the fact that some of the infected women received prenatal antibiotic treatment. It is highly unlikely that Chlamydial-infected women in the first study mentioned above received treatment as this is not mentioned or referred to at all [11]. In the second study mentioned above that associated infection with C. trachomatis with LBW, PROM and preterm labour, women received treatment for gonococcal infections but Chlamydial infections were not treated until after the onset of labour or rupture of membranes [12].

In two studies from Kenya, infection with C. *trachomatis* was associated with postpartum endometritis [14] and 31% of 181 cases of neonatal conjunctivitis were found to have been caused by *C. trachomatis* [15].

1.2.1.3 Neisseria gonorrhoea

Three studies on the impact of *N. gonorrhoea* infection on pregnancy outcome were noted in the review by Mullick *et al.* 2005. One was based on case reports in the United Kingdom and documented sequelae of untreated gonorrhoea in pregnancy

which included PD, PROM, LBW, postpartum endometritis and gonococcal ophthalmia neonatorum [16]. A case control study of women at delivery in Kenya found an association between *N. gonorrhoeae* infection and LBW (aOR 2.9; 95% CI 1.2 - 7.2) and it was concluded that *N. gonorrhoeae* was responsible for 14% of LBW cases in the study population [17]. In South Africa a prospective study of 167 women looked at gonorrhoea diagnosed at first ANC visit and pregnancy outcome [18]. It was found that women with gonorrhoea delivered significantly smaller babies (mean weight 2252g versus 2970, *P* < 0.005).

1.2.1.4 *Trichomonas vaginalis*

Among the four studies of *Trichomonas vaginalis* in pregnancy included in this review, only one study was done in Africa [19]. The studies associated infection with *T. vaginalis* in pregnancy with PD and LBW [19-22]. In the Democratic Republic of Congo, trichomoniasis infection in pregnancy was associated with LBW in infants born to infected mothers (aOR 2.4; 95% Cl, 1.2- 4.5) [19]. In the three studies in the US, one found that infection with *T. vaginalis* was associated with LBW among adolescents [22]; in the second study, women infected with *T. vaginalis* were significantly more likely to have PROM (P < 0.03) [21]. Trichomoniasis was associated with LBW and PD of a LBW infant (aOR 1.3; 95% Cl, 1.1 to 1.5), PD (aOR 1.3; 95% Cl, 1.1 -1.4) and PD of a

1.2.1.5 Bacterial vaginosis

Studies on the effect of bacterial vaginosis (BV) on birth outcomes identified in the review by Mullick *et al.* included a prospective, controlled treatment trial in the US. Bacterial vaginosis was associated with increased risk of both PD (aRR 1.9; 95% CI, 1.2-3.0) and PROM (aRR 3.5; 95% CI, 1.4-8.9) in women in the observation phase of the study [23]. A cohort study in the US found an association between BV and PD of LBW infants (aOR 1.4; 95 % CI, 1.1- 1.8) [24] and a prospective study in Indonesia associated BV diagnosed early, between 16 and 20 weeks gestation, with PD (OR 2.0; 95% CI, 1.0-3.9) but not with BV diagnosed at 28 to 32 weeks gestation (aOR 1.5, 95% CI, 0.7-3.0) [25].

1.2.2 Additional evidence for association of STIs/RTIs with ABO

To explore additional evidence for the effects of STIs/RTIs on pregnancy outcome, a literature search was undertaken. The last search was conducted in February 2016. The Pubmed, Medline, Embase and Global Health databases were searched using combinations of the following key words: "Pregnancy", "antenatal/prenatal", disease", "sexually transmitted "syphilis/*Treponema* pallidum", "Chlamydia/Chlamydia trachomatis", "Gonorrhoea/Neisseria "trichomoniasis/Trichomonas vaginalis", "bacterial *gonorrhoeae*/gonorrhoea", vaginosis" and "birth/pregnancy outcome". The words "Malaria/Plasmodium falciparum" were also used to identify and include studies reporting the effect of malaria on pregnancy outcome in the search. The Medline, Embase and Global Health databases were searched via Ovid[®]. Appendix 1.1a shows search terms that were used for studies reporting infection with malaria or STI/RTI in pregnancy and birth outcomes. There was no restriction of the time when the studies were conducted. Titles and abstracts were screened for relevance and full texts were screened for those that were relevant to the subject. Articles with full texts published in non-English language were excluded. Studies that reported investigating the association between malaria or STI/RTI and ABO were identified. Figure 1.1 shows a flow chart of the literature search and identification of articles.

Figure 1. 1: Identification and screening of studies



Overall, 32 studies associating STIs/RTIs with ABO (including those that were identified in the review by Mullick *et al.*) and seven studies on malaria and ABO were identified. Below are brief details of 16 of the 32 studies on STIs/RTIs and ABO that were published after the review by Mullick *et al.*, 2005.

1.2.2.1 Syphilis

In a multiple centre record based retrospective study of 368,151 births in Latin America and the Caribbean, syphilis was associated with stillbirth (aOR 1.88; 95% CI, 1.25-2.83) [26].

A review and meta-analysis of studies reporting estimates of ABO among untreated 1,715 women with syphilis and 22,515 women without syphilis, pooled estimates showed that among untreated pregnant women with syphilis, fetal loss and stillbirth were 21% more frequent, neonatal deaths were 9.3% more frequent and PD or LBW were 5.8% more frequent than among women without syphilis [27].

1.2.2.2 Chlamydia trachomatis

In a study of 343 pregnant women in South Africa, 36 (10.5%) delivered before 37 weeks gestation and C. trachomatis was found in 8 (22.2%) of women who had PD in contrast to 32 (10.4%) of women who had full-term deliveries (P = 0.037) [28]. Results from an unmatched case-control study conducted in Uganda showed no association between C. trachomatis infection and PROM (aOR; 2.05, 95% CI, 0.37-11.49) [29]. In a study of 3,913 pregnant women in the Netherlands C. trachomatis infection was associated with PD occurring before 32 weeks (aOR 4.35; 95% CI, 1.3 -15.2) and 35 weeks gestation (aOR: 2.66; 95% Cl, 1.1-6.5), but not before 37 weeks (aOR 1.17; 95%Cl, 0.6-2.4) and not with LBW (P = 0.25) [30]. In a case control study in Switzerland, C. trachomatis was associated with PD (aOR; 7.93, 95% CI, 1.34-46.76) [31]. A record-based retrospective cohort study conducted in Australia involving 354,217 participants, showed that women with Chlamydia had increased risk of PD (aOR 1.17, 95% CI, 1.01-1.37) and stillbirth (aOR 1.40, 95% CI, 1.00 -1.96) but no association was found between infection with Chlamydia and small for gestational age (aOR 0.99, 95% CI, 0.89-1.09) [32]. In a retrospective study in the US, electronic medical records from a sexually transmitted disease clinic were matched with state birth records for 730 women [33] Chlamydia was associated with LBW (aOR 2.07;

95% CI: 1.01 - 4.24). A population based retrospective cohort study in the US found that Chlamydial-infected women were at increased risk of PD (aRR 1.4; 95% CI: 1.08 - 1.99) and PROM (aRR 1.5; 95% CI 1.03 -2.17) compared with uninfected women [34]. No increased risk of LBW was observed in this study (aRR 1.12; 0.61-1.68). Another study done in the US found no association between *C. trachomatis* infection and PD (aOR 0.7; 95% CI 0.4–1.4) [35]. This could be attributed to the fact that 66% of the positive diagnoses were done early and treatment was administered before 20weeks gestation.

1.2.2.3 Neisseria gonorrhoeae

In a record-based retrospective cohort study in Australia involving 354,217 participants, no association was found between infection with gonorrhoea and stillbirth (aOR 2.35, 95% CI, 0.58-9.56) or gonorrhoea and small for gestational age (aOR 0.98, 95% CI, 0.58-1.68) [32]. Women infected with gonorrhoea had a higher risk of experiencing PD (OR 2.50, 95% CI, 1.39-4.50) [32]. Similarly, gonorrhoea was associated with PD (aOR 2.01; 95% CI 1.02 - 3.97), particularly when diagnosed in the first trimester (aOR 2.95; 95% CI: 1.30 - 6.70) in a study of 730 women in the US [33].

1.2.2.4 Trichomonas vaginalis

No addition studies on the effect of *T. vaginalis* on ABO were identified in the literature search. The only studies that were identified were included in the review by Mullick *et al.*, 2005 and are already described above.

1.2.2.4 Bacterial vaginosis

In a prospective study of 1,536 Tanzanian women designed to determine risk factors for poor birth outcomes, no association was found between treated STIs/RTIs and ABO except untreated BV which was associated with PD (aOR; 2.9 95% Cl, 1.3 - 6.6) [36]. In a prospective cohort study of 3,262 women in Denmark, BV was associated with LBW (aOR 1.95; 95% Cl, 1.3-2.9), PD of a LBW infant (aOR 2.5; 95% Cl, 1.6-3.9) and clinical chorioamnionitis (aOR 2.4; (95% Cl, 1.4-4.1) [37]. In the United Kingdom, women with BV were not at a higher risk of experiencing PD (aOR 0.9, 95% Cl, 0.4-2.2) but had increased risk of experiencing spontaneous abortion (aOR 4.0, 95% Cl, 1.3-12.1) [38].

In a meta-analysis aimed at evaluating BV as a risk factor for PD, BV was associated with PD (aOR 2.2; 95% Cl, 1.5-3.1) and spontaneous abortion (aOR 9.9; 95% Cl, 2.0-49.3) [39]. The meta-analysis was updated after about four years and BV was still associated with PD (aOR 2.2; 95% Cl, 1.6-3.9) and spontaneous abortion (aOR 6.3; 95% Cl, 3.7-10.9) [40]. In Switzerland a study of 1,197 pregnant women found that BV was associated with preterm delivery (aOR 2.19; 95% Cl, 1.2-4.0) [41].

1.2.3 Effects of malaria on pregnancy outcomes

Pregnant women in malaria endemic areas are a high risk group for *Plasmodium falciparum* infection and related consequences [42]. Pregnancy results in a transient depression of cell-mediated immunity which improves the likelihood of foetal allograft retention but also interferes with immune responses to various infectious diseases [43]. Red blood cells that have been infected by *P. falciparum* sequester in vascular spaces in the placenta mediated by adhesion to chondroitin sulphate A [44] leading to placental inflammation [45] and complications associated with malaria in pregnancy [46]. Consequences of malaria in pregnancy include maternal anaemia, IUGR [36, 47], PD [36, 48], stillbirth [49, 50] and LBW [36, 42]. Low birth weight is associated with a marked increase in neonatal death [51-54].

Studies that investigated the effect of malaria on pregnancy outcome identified in the literature search mentioned above were all conducted in Africa. In Tanzania a study of the effectiveness of syphilis screening and treatment conducted from 1997 to 2000, women with malaria had an increased risk of experiencing PD (aOR 3.2; 95% CI, 1.9-5.2), LBW (aOR 5.4; 95% CI, 3.1-9.5) and IUGR (aOR 2.8; 95% CI, 1.2-6.7) [36]. In another study designed to identify risk factors for stillbirth conducted in November and December of 2006 in Ghana, women with malaria had increased risk of stillbirth (aOR 1.9; 95% CI, 1.2-9.3) [50]. In a hospital based study also aimed at identifying risk factors for stillbirth, conducted between November 2007 to February 2008 in Sudan, infection with malaria was associated with increased risk of stillbirth (aOR 3.0; 95% CI, 1.0-8.9) [49].

Among 1,766 women in Malawi, placental blood *P. falciparum* infection was independently associated with LBW (aOR: 1.71) [55]. Secondary analysis of data for

2,149 women included in a community based randomised placebo controlled trial for the prevention of preterm birth. Results showed that persistent malaria (despite malaria prophylaxis) increased the risk of late preterm birth (aOR 1.99, 95% Cl, 1.05-3.79) [56]. In a case-control study in Sudan conducted in 2010, a combination of histologically determined and sub-microscopic infections was significantly associated with LBW (aOR 2.45; 95% Cl, 1.2-4.9) [57]. No association was found between malaria and LBW (aOR 1.0; 95% Cl, 0.4-2.5), PD (aOR 1.7; 95% Cl, 0.7-3.9), stillbirth (aOR 1.0; 95% Cl, 0.3-3.0) and any ABO (aOR 1.3; 95% Cl, 0.66-2.38) in a study in Sudan [58] . This lack of association was attributed to high IPTp-SP uptake in the study population.

Table 1-1 shows a summary of studies associating curable STIs/RTIs and malaria with ABO.

Organism/ infection	LBW	IUGR	PD	Stillbirth	PROM	Spontaneous abortion	Country	year	Author, publication Year
Plasmodium falcinarum	5.4(3.1-9.5)	2.83 (1.2-6.7)	3.2 (1.9-5.2)	NA	NR	NR	Tanzania	1997-2000	Watson-Jones et al., 2007[36]
jaro.param	NR	NR	NR	3.0 (1.0-8.9)	NR	NR	Sudan	2007-2008	Bader et al., 2010 [49]
	NR	NR	NR	1.9 (1.2-9.3)	NR	NR	Ghana	2006	Yatichi <i>et al.,</i> 2010 [50]
	1.71, P < 0.05	NR	NR	NR	NR	NR	Malawi	1987-1990	Steketee <i>et al.,</i> 1996 [55]
	2.45 (1.2-4.9)	NR	NR	NR	NR	NR	Sudan	2010	Muhammed et al., 2013 [57]
	NR	NR	2.0 (1.1-3.8)	NR	NR	NR	Malawi	2004-2005	Van der Brook <i>et al.,</i> 2014 [56]
	1.0 (0.4-2.5)	NR	1.7 (0.7-3.9)	1.0 (0.3-3.0)	NR	NR	Ghana	2011	Asundep <i>et al.,</i> 2014 [58]
Treponema	NR	NR	NR	NR	NR	42% of cases	Zambia	1982	Ratnam et al., NR [59]
pallidum	NR	NR	1.4 (0.5-4.1)	NR	NR	NR	Kenya	1985	Elliot <i>et al.,</i> 1990 [17]
	NR	NR	33%, 5 of 15 cases	NR	NR	NR	South Africa	1988	Donders <i>et al.,</i> 1993 [18]
	NR	NR	NR	11.0 (6.6-17.9)	NR	NR	Malawi	1987-1990	McDermott <i>et al.,</i> 1993 [9]
	4.01, <i>P</i> < 0.001	NR	NR	3.3 ^a , <i>P</i> = 0.028	NR	NR	Kenya	1991	Temmerman <i>et al.,</i> 1995 [60]
	3.3 ^a (2.0-5.4)	2.1 ^ª (1.0-4.2)	6.1 ^ª (2.5-15.3)	18.1 ^ª (5.5-59.6)	NR	NR	Tanzania	1998-2000	Watson-Jones <i>et al.</i> , 2002 [61]
	NR	NR	NR	1.9 (1.3-2.8)	NR	NR	Multiple	2009-2012	Arnesen <i>et al.,</i> 2015[26]
Neisseria aonorrhoeae	2.9(1.2-7.2)	NR	NR	NR	NR	NR	Kenya	1985	Elliot <i>et al.</i> , 1990 [17]
gonormocue	<i>P</i> = 0.005 [‡]	NR	56%, 5 of 9 cases	NR	NR	NR	South Africa	1988	Donders <i>et al.,</i> 1993 [18]
	0.8 (0.3-2.3)	NR	2.0 (1.0-4.0)	NR	NR	NR	USA	1996-2002	Johnson <i>et al.,</i> 2011 [33]
	NR	NR	2.5 (1.4-4.5)	NR	NR	NR	Australia	1999-2008	Liu <i>et al.,</i> 2013 [32]

Table1- 1: Summary of studies associating curable STIs/RTIs and malaria with adverse pregnancy outcomes

Organism/ infection	LBW	IUGR	PD	Stillbirth	PROM	Spontaneous abortion	Country	year	Author, publication Year
Chlamydia	2.7(1.3-5.7)	NR	NR	NR	2.4 (1.7-5.4)	NR	USA	1983	Gravett <i>et al.,</i> 1986 [12]
trachomatis	NR	NR	19.4% versus	NR	19.4% versus	NR	USA	NR	Sweet et al., 1987 [62]
			8.0%, <i>P</i> = 0.03		8.0%, <i>P</i> = 0.03				
	NR	2.4 (1.3-4.2)	NR	NR	NR	NR	USA	1983-1985	IJH, 1989 [11]
	NR	NR	NR	NR	NR	1.8 (0.3-10.7)	USA	1995-1996	Sozio et al., 1998 [63]
	15.5 versus 13.2	7.3 versus 5.8%	NR	20 versus 21%	NR	NR	Hungary	1994-1995	Kovacs et al., 1998[64]
	<i>P</i> > 0.05	<i>P</i> > 0.05		<i>P</i> > 0.05					
	NR	NR	22.2% 8 of 36 cases versus 10.4% 32 of 307 cases P = 0.037	NR	NR	NR	South Africa	2002-2003	Odendaal <i>et al.,</i> 2006 [28]
	1.1 (0.7-1.70) ^a	NR	1.5 (1.1-2.0) ^a	NR	1.5 (1.0-2.2) ^a	NR	USA	2003	Blas et al., 2007 [34]
	0.19 (0.1-1.5)	0.97 (0.1-7.9)	0.59 (0.2-2.0)	NR	NR	NR	Tanzania	1997-2000	Watson-Jones et al., 2007 [36]
	0.3 (0.1-1.2) NR	3.65 (1.1-12.0)	1.15 (0.5-2.6)	NR	NR	NR		2005-2008	Silveira et al. 2009 [35]
			0.7 (0.4 1.4)				03/(2003 2000	
	2.07 (1.01-4.24)	NR	1.0 (0.6-2.0)	NR	NR	NR	USA	1996-2002	Johnson <i>et al.,</i> 2011 [33]
	NA	NR	4.4 (1.3-15.2) ^b 2.7 (1.6-6.5) ^c	NR	NR	NR	Netherlands	2003-2005	Rours <i>et al.,</i> 2011 [30]
	NR	NR	7.9 (1.3-46.9)	NR	NR	NR	Switzerland	2006-2009	Baud et al., 2014 [31]
	NR	NR	NR	NR	2.1 (0.4-11.5)	NR	Uganda	2013	Nakubulwa <i>et al.,</i> 2015 [29]

Table 1-1: Summary of studies associating curable STIs/RTIs and malaria with adverse pregnancy outcomes continued

Organism/ infection	LBW	IUGR	PD	Stillbirth	PROM	Spontaneous abortion	Country	year	Author, publication Year
Trichomonas vaginalis	NR	NR	NR	NR	P < 0.03	NR	USA	NR	Minkoff <i>et al.,</i> 1984 [21]
vaginans	1.3(1.1-1.5)	NR	1.3 (1.1-1.4)	NR	NR	NR	USA	1984-1990	Cotch <i>et al.,</i> 1997 [20]
	2.1(1.0-4.2.)	NR	NR	NR	NR	NR	Congo DR	1989-1990	Sutton <i>et al.,</i> 1999 [19]
	1.11 (0.6-1.9)	0.61 (0.2-1.5)	1.27 (0.8-2.0)	2.32 (1.0-5.7)	NR	NR	NR	1997-2000	Watson-Jones <i>et al.,</i> 2007 [36] [¥]
	NR	NR	2.38 (0.5-12.1)	5.57 (0.5-66.1)	NR	NR	NR		
	1.5 (0.9-2.6)	NR	1.4 (0.7-2.8)	NR	NR	NR	USA	1996-2002	Johnson <i>et al.,</i> 2011[33]
Bacterial vaginosis	NA	NR	NR	NR	2.0 (1.1-3.7)	NR	USA	1983	Gravett et al., 1986 [12]
VuBinosis	NR	NR	2.0 (1.0-3.9) ^e	NR	NR	NR	Indonesia		Riduan <i>et al.,</i> 1993 [25]
	NR	NR	1.9 (1.2-3.0) ^a	NR	3.5 (1.4-8.9) ^a	NR	USA	1991-1992	Mcgregor <i>et al.,</i> 1995 [23]
			1.5 (0.7-3.0) ^f						
	NR	NR	1.4 (0.9-2.05) Week 24, 1.8 (1.2-3.0) Weeks 28	NR	NR	NR	Multiple	Multiple	Meis <i>et al,</i> 1995 [65]
	1.5 (1.2-1.7)	NR	1.4 (1.1-1.8)	NR	1.1 (0.8-1.6)	NR	USA	1984-1989	Hillier et al., 1995 [24]
	NR	NR	0.9 (0.4-2.2)	NR	NR	4.0 (1.3-12.1	UK	1998-2000	Oakeshott et al., 2004 [38]
	NR	NR	2.16 (0.9-3.6) ^a	NR	NR	NR	Sweden	1990-1991	Jacobson <i>et al.,</i> 2002 [66]
	NR	NR	2.2 (1.5-3.1)	NR	NR	9.9 (2.0-49.3)	Multiple	Multiple	Leitich <i>et al.,</i> 2003 [39]
	2.0 (1.3-2.9)	NR	2.5 (1.6-3.9)*	NR	NR	NR	Denmark	1998-2002	Svare et al., 2006 [37]
	NR	NR	2.2 (1.6-3.0)	NR	NR	6.3 (3.7-10.9)	Multiple	Multiple	Leitich <i>et al.,</i> 2007 [40]
	NR	NR	2.19 (1.2-4.0)	NR	NR	NR	Switzerland	NR	Daskalakis et al. , 2006 [41]
	1.08 (0.7-1.9) 2.02 (0.7-5.9)	1.09 (0.6-2.2) NR	0.91 (0.6-1.4) 2.95 (1.3-6.6)	1.79 (0.8-4.2) NR	NR NR	NR NR	Tanzania	1997-2000	Watson-Jones <i>et al.,</i> 2007 [36] [¥]

Table 1-1: Summary of studies associating curable STIs/RTIs and malaria with adverse pregnancy outcomes continued

Table 1-1: Summary of studies associating curable STIs/RTIs and malaria with adverse pregnancy outcomes continued

IJH: Investigators of John Hopkins; IUGR: Intrauterine growth retardation LBW: Low birth weight NR: Not reported PROM: Premature rupture of membranes PD: Preterm delivery ^a Adjusted risk ratio, the other figures are adjusted odds ratios. Bracketed figures are 95% confidence Intervals for odds ratios. ^b Preterm delivery before 32 weeks ^c Preterm delivery before 35 week ^d preterm delivery before 37 weeks ^eBacterial vaginosis diagnosed at 16 to 20 weeks ^fBacterial vaginosis diagnosed at 28 to 32 weeks ^gAmong treated women *preterm delivery of low birthweight infant ^{*}first row represents untreated women and second row represents treated women [†]22.2% versus 10.4% prevalence of Chlamydia among women with preterm delivery and full term delivery ^{*}Mean birth weight of preterm babies versus full-term, 2252g and 2970g, *P* = 0.005 Bold type: statistically significant association

1.2.4 Effect of treatment of curable STIs/RTIs in pregnancy

The benefits of STIs/RTIs treatment on pregnancy outcome have been demonstrated in randomised trials. Three studies on the effects of treatment of STIs/RTIs were identified in the review by Mullick and colleagues [67-69]. A randomised trial in the US showed that treatment with erythromycin between 26 and 30 weeks gestation resulted in reduced frequency of PROM (6%) compared to placebo recipients (16%), P < 0.01 [68]. In a double-blind placebo controlled clinical trial, a single dose of ceftriaxone given to pregnant women in Kenya between 28 and 32 weeks led to a significant increase in mean birthweight [3.21kg versus 3.06kg, (P < 0.01)] [69]. A randomised trial of presumptive sexually transmitted therapy in pregnancy in Uganda, showed that treatment with a single dose of metronidazole combined with azithromycin and cefixime reduced rates of neonatal death (rate ratio 0.83; 95% Cl, 0.71-0.97), LBW (rate ratio 0.68; 95% Cl, 0.53-0.86), and PD (rate ratio 0.77; 95% Cl, 0.56-1.05) [67].

Studies focusing on specific infections have shown benefits of treatment of STIs/RTIs in pregnancy. A cohort study in the US found an increased risk in the incidence of PROM in untreated women with positive Chlamydia cultures compared to those with positive cultures who received treatment or those with negative cultures [70]. Another study in the US showed a decreased risk of PROM in successfully treated Chlamydia positive patients compared to patients who were treated but had either recurrent or persistent Chlamydia infection at the end of pregnancy (aOR 0.31; 95% CI) [71]. In the third study in India the mean duration of gestation for PD was significantly higher among Chlamydial-infected women who received treatment in comparison to untreated women [35.5 versus 33.1 weeks (P < 0.05)] [72].

Treatment of syphilis has also been shown to be beneficial, particularly if done in the first trimester compared to the third [61]. A study in Tanzania demonstrated that there was no increased risk of adverse pregnancy outcomes for women treated for high titre active syphilis (OR 0.76; 95% Cl, 0.4 -1.4) or low titre active syphilis (OR 0.95; 95% Cl, 0.6 -1.5) compared with sero-negative women. Women were recruited, tested for syphilis and treated on the same day (mean recruitment age, 25.8 weeks) [61]. However, whilst treatment of women who were RPR (rapid plasma reagin) positive at antenatal clinics in Kenya significantly improved pregnancy outcome, it

did not eliminate the risk altogether [60]. Indeed, in this study, infected women who did not receive treatment had a marked higher risk of ABO (OR 4.1; 95% CI, 2.4-7.2) compared to uninfected women; but infected women who received treatment still experienced a 2.5-fold higher risk of ABO than uninfected women. Women who were treated but were still syphilis positive at delivery had significantly more ABO than those without syphilis (14.7 versus 6.2; P < 0.05) but less than untreated women (26.2% versus 14.7%; P < 0.05).

Treatment of BV with clindamycin in the treatment phase of a controlled drug trial in the US was associated with reduced PD (RR 0.5; 95% CI, 0.3-0.8) and PROM (RR 0.5; 95% CI, 0.2-1.4) [23]. Findings in another study in the US showed that women with clinically diagnosed BV, and were treated using metronidazole and erythromycin, had a lower incidence of PD (31% with treatment versus 49% with placebo, P = 0.006) [73].

Another study in the US designed to evaluate the effect of early BV screening (at < 22 weeks gestation) and treatment followed by re-screening and re-treatment if necessary, found that treatment of BV (with oral metronidazole or intravaginal metronidazole or clindamycin) was significantly associated with a reduction in the risk of PD (OR 0.5; 95% CI, 0.3-0.8) [74]. Similar results were found in another study in the United Kingdom where screening and treatment with intravaginal clindamycin were done early (13-20 weeks gestation)[75]. Treated women experienced a significant reduction in PD compared with the placebo group [4% versus 10%, (P < 0.03)] [76].

In a review of studies on screening, treatment, or ABO data in women with asymptomatic BV, no benefit was found in the treatment of asymptomatic BV [77]. Treatment of *T. vaginalis* infection in pregnancy has been shown to have no effect on gestational age and birthweight. A study in South Africa found that gestational age and birth weight of infants born to women who were treated with benzoyl-metronidazole and in those who did not receive treatment were similar [78]. Treatment was effective in clearing infection. A review aimed at determining whether antibiotic treatment for BV or *T. vaginalis* during pregnancy decreases the risk of PD and associated ABO concluded that there is no evidence to support the use of antibiotic treatment for these infections for the purpose of reducing the risk of PD
and associated ABO [75]. These conclusions were based on two studies. One US study evaluating treatment of asymptomatic trichomoniasis (selectively enrolled asymptomatic women) with metronidazole suggested that the drug failed to prevent PD [79]. In this study women treated with metronidazole were significantly more likely to deliver a preterm infant than untreated women (RR 1.8; 95% Cl 1.2-2.7). The second study was a sub-analysis of a randomised trial in Uganda which showed that women who were treated for trichomoniasis with metronidazole were 2.5 times more likely to deliver a LBW infant (RR 2.49; 95% CI 1.12-5.50) [80]. Although the original study was not designed to assess the effect of T. vaginalis treatment, the authors inferred that this might have been due to metronidazole exposure. However, what was observed in this study may be attributable to some other factor rather than what was inferred by the authors since the risk of PD was not increased in other clinical trials of metronidazole treatment of bacterial vaginosis during pregnancy [73, 81-83]. Furthermore a sub-analysis of a study of pregnant women in four sub-Saharan African sites found that treatment of trichomoniasis with metronidazole did not influence the risk of PD or LBW, women randomised to the antibiotic arm were not more likely to deliver a preterm infant (20.9% versus 19.8%, P = 0.84) or to deliver an infant with a lower mean birth weight (2992 versus 2930 P = 0.27) [84]. The treatment effectively resolved *T. vaginalis* infection.

1.3 Prevalence of malaria and STIs/RTIs

1.3.1 General overview of malaria prevalence in pregnancy

Several reviews have reported the prevalence of malaria in pregnancy. Brabin reviewed 14 studies conducted before 1980 and found peripheral and placental parasitaemia ranging from 2-76% and 2-74%, respectively [42]. This review included two studies from Panama and 12 conducted in Africa. The wide range of prevalence of malaria in pregnancy is partly due to the difference in the level of malaria transmission in the study sites. Generally, the prevalence of malaria in pregnancy was higher in rural than urban areas. The median prevalence was at 27%. Another review of studies done between 1985 and 2000 in eight countries found maternal malaria infection, defined as placental or peripheral, in all gravidae ranged between

6 and 65% [54]. The median prevalence of maternal parasitaemia was 27%. A similar estimate of 26% was obtained for placental parasitaemia (range 5-52%) from another review of studies conducted between 1980 and 2001 [85]. Desai *et al.,* reviewed 13 studies conducted between 1986 and 2004 of *P. falciparum* infection in pregnancy in areas of low-transmission settings in Africa and found a median prevalence of peripheral and placental parasitaemia at 13.7% and 6.7%, respectively [86]. The point prevalence estimates in the review by Desai *et al.,* were generally lower since the review concentrated on low transmission areas.

The meta-analysis by Chico *et al.* reviewed studies conducted among antenatal attendees in sub-Sahara Africa [6] and reported pooled prevalence estimates for malaria and STIs/RTIs. The estimates were reported separately for West and Central Africa and for East and Southern Africa. This review included studies of women receiving antenatal care conducted between 1990 and 2011 and excluded studies that had selective enrolment, i.e. high risk groups such as HIV-seropositive women. These estimates were calculated using a standard method for correcting errors of magnitude based on the known specificity and sensitivity of individual diagnostic methods [87].

Pooled prevalence estimates for Eastern and Southern Africa from this review were as follows: placental malaria 25.8% (95% CI, 19.7-31.9) and peripheral malaria 32.0% (95% CI, 25.9-38.0). Prevalence estimates that were obtained for West and Central Africa were higher than those obtained for East and Southern; placental and peripheral parasitaemia was at 39.9% (95% CI, 34.2,-45.7) and 38.2% (95% CI, 32.3-44.1) respectively [6].

At the patient level, placental parasitaemia is almost always higher than measures of peripheral parasitaemia [88-90]. When this has not been the case in systematic reviews or meta-analyses, this is because not all studies conduct placental histology; therefore, fewer data points are often pooled for estimates of placental infection. If these happen to be studies where the prevalence of parasitaemia is lower, then the overall frequency will be lower than pooled measures of peripheral infection.

1.3.2 General overview of STI/RTI prevalence

The review (mentioned in earlier text) by Mullick *et al.* 2005 of studies published between 1987 and 2004 also examined the prevalence and the impact of STIs/RTIs on pregnancy outcomes in developing countries. Results from this review by Mullick *et al.* showed that up to 40% of pregnant women in Africa had trichomoniasis and bacterial vaginosis, 2.5–17% had serological evidence of syphilis, 3–6% while the prevalence of gonorrhoea and Chlamydia ranged from 2–7% and 3–29%, respectively. Studies conducted between 1990 and 2011 reporting point prevalence estimates for malaria and STI/RTIs among pregnant women were identified in a more recent review and meta-analysis by Chico *et al.*, 2012. Pooled prevalence estimates for West and Central Africa from this meta-analysis are as follows: syphilis, 3.5% (95% CI, 1.8%-5.2%), gonorrhoea*e*, 2.7% (95% CI, 1.7%-3.7%), Chlamydia 6.1% (95% CI, 4.0%-8.3%), trichomoniasis 17.8% (95% CI, 12.4%-3.1%), BV, 37.6% (95% CI, 18.0%-57.2%). For East and Southern Africa, the estimates were as follows: syphilis 4.5% (95% CI, 3.9-5.1), gonorrhoea 3.7% (95% CI, 2.8- 4.6), Chlamydia 6.9% (95% CI, 5.1- 8.6), trichomoniasis 29.1 (95% CI, 20.9-37.2) and BV 50.8% (95% CI, 43.3- 58.4).

1.4 Prevalence of malaria and STIs/RTIs in pregnancy in Zambia and neighbouring countries

The review by Chico et al. covered a broader region in sub-Saharan Africa. In this study a review was conducted for a smaller region of sub-Saharan Africa, Zambia and neighbouring countries. This was done to focus on areas surrounding the study site in which a substantial number of prevalence studies have been conducted. To explore studies reporting the prevalence of malaria and STIs/RTIs in pregnancy conducted between 1990 and 2015 in Zambia and neighbouring countries, a literature search was conducted. Four data bases, Pubmed, Medline, Embase and Global Health were searched for studies reporting P. falciparum, T. pallidum, N. gonorrhoea, Chlamydia trachomatis, T. vaginalis and BV using medical subject headings and free texts. The Medline, Embase and Global Health data bases were searched simultaneously via Ovid® using combinations of the following key words; "antenatal/prenatal", "Pregnancy", "sexually transmitted disease",

"syphilis/*Treponema pallidum*", "Chlamydia/*Chlamydia trachomatis*", "Gonorrhoea/*Neisseria gonorrhoeae*/gonorrhoea/gonorrhea", "trichomoniasis/*Trichomonas vaginalis*", "bacterial vaginosis", "malaria", "*Plasmodium falciparum*" "test", "screen" and "Africa". Search terms that were used are shown in Appendix 1.1b. The last search was conducted in April 2016.

If an article reported two or more estimates e.g., if more than one diagnostic method was used, the highest/higher estimate was picked. Studies of women seeking treatment at STI facilities, non-pregnant women and those reporting estimates in high-risk groups such as commercial sex workers and in HIV-positive pregnant women were excluded. Multi-year studies reporting prevalence estimates that included data collected before 1990 were also excluded. Studies conducted in countries that are not Zambia's immediate geographic neighbours were also excluded, hence only studies conducted in Angola, Botswana, Democratic Republic of Congo (Congo DR), Malawi, Mozambique, Namibia, Tanzania and Zimbabwe were included. All the articles obtained from all the search terms were originally retained and duplicates were then removed and other studies were excluded based on the criteria stated above. Appendix 1.2a to 1.2g show point prevalence estimates for malaria and STIs/RTIs from studies conducted in Zambia and neighbouring countries and diagnostic methods used in each study. Figure 1.2 shows the identification and screening of studies in the review.

Figure 1. 2: Flow chart



1.4.1 Prevalence of malaria in pregnancy in Zambia and neighbouring countries

Studies in Zambia and neighbouring countries conducted between 1990 and 2015 have reported the prevalence of placental and peripheral malaria in all gravidae ranging from 9.8-52.0% and 4.7-51.9%, respectively (Appendix 1.2a and 1.2b). The wide range of prevalence of malaria in pregnancy is partly due to the difference in the intensity of malaria transmission in the various study sites. The median prevalence of placental malaria (from 12 studies) and peripheral parasitaemia (from 22 studies) was 20.3% and 20.5%, respectively.

1.4.1.1 Diagnostic methods for malaria and published sensitivity and specificity

Diagnostic methods for malaria and published sensitivity and specificity are presented within Appendix 1.3. The gold standard for malaria diagnosis, Giemsa stain microscopy, has sensitivity measures ranging from 50-90% and specificity of approximately 95% [91-93]. The sensitivity and specificity of Gram stain microscopy are greatly influenced by the skills and workload of microscopists [94] and therefore the sensitivity and specificity of the method are expected to be on the higher end when the method is carried out by trained individuals with reasonable workload. One advantage of standard microscopy is that the method requires a relatively short time for diagnosis when used in areas of high transmission and parasites are present in high concentrations (approximately 1,000 parasites per microlitre $[\mu L]$ of blood). However, if parasite densities are very low, examination of each slide is labourintensive [94]. The use of PCR for the diagnosis of malaria is highly sensitive in the detection of parasites. The method has been reported to have 91% sensitivity and specificity [95] and has the ability to measure infections where parasite counts are as low as five per µL of blood [96, 97], although the sensitivity of PCR has been shown to decrease markedly at parasite densities $< 500/\mu L$ [98]. In a study in Taiwan a high proportion (58.2%) of microscope-positive samples were negative by PCR and 98.0% of these infections had fewer than 250 parasites/ μ l of blood [98]. In another study PCR failed to detect parasite DNA in two microscopy positive samples [99]. Other studies have reported that PCR may occasionally yield false negative results [98, 100, 101] sometime as a result of inappropriate DNA isolation [102, 103]. Among the 22 studies reporting prevalence of peripheral malaria among pregnant women in

Zambia and neighbouring countries Giemsa stain microscopy was used in 14, PCR in seven and RDT in one study (Appendix 1.2f)

Since Giemsa stain microscopy was used in the majority of these studies, it would be expected that there was some level of underestimation of the actual malaria prevalence in the region. However, the estimation can be considered reliable as they are expected to lie within acceptable ranges of the actual prevalence provided diagnoses were conducted under appropriate conditions.

The use of placental blood microscopy in placental malaria diagnosis has 63% sensitivity and 98-99% specificity [104]. Placental histology is reported to have 91% sensitivity and 98-99% specificity [104]. Placental blood microscopy and tissue histology (score based on more precise method, i.e. histology), placental histology, impression, smears with microscopy (score based on most precise method, i.e. histology) and placental impression and blood smears with microscopy (score based on more precise method, i.e. histology (score based on more precise method, i.e. histology) and placental impression and blood smears with microscopy (score based on more precise method, i.e. histology) and placental impression and blood smears with microscopy (score based on more precise method, i.e. impression) have all been reported to have 91% sensitivity and approximately 100% specificity [6, 105].

In the case of studies that used more than one method for the detection of placental malaria the estimate based on the method with higher sensitivity and specificity was recorded. The majority of studies (eight out of the 14) that contributed to the range of the prevalence estimates were based on histology (five) and impression (three) both of which are reported to have the same sensitivity and specificity measures. Placental blood microscopy was used in two of the studies and PCR was use in the remaining four studies. Based on the published sensitivity and specificities of the methods one would expect the estimates to be slightly lower than the situation obtaining on the ground. However, these estimates can be considered reliable and the wide range between the highest and lowest recorded prevalence estimates are mainly due to differences in malaria prevalence in the different areas in which these studies were conducted.

1.4.1.2 The malaria situation in Zambia

In Zambia, malaria is endemic in all 10 provinces and *P. falciparum* is responsible for approximately 95% of all cases [106]. There are an estimated 4.3 million clinically

diagnosed cases of malaria and 6,150 deaths annually [106]. Transmission of malaria is year-round but peaks in the rainy season (November-April). An estimated 716,192 pregnancies in Zambia were at risk of malaria in 2010 based on projections from the 2000 census [107].

Since 2006, the National Malaria Control Programme has conducted a Malaria Indicator Survey (MIS) every two years to measure the prevalence of malaria by microscopy in children under five years of age in selected sites. The prevalence of malaria parasitaemia decreased in Zambia from 2006 to 2010 in some regions, while little change was observed in others [108]. Of concern is that parasitaemia declined between 2006 and 2008 in Eastern, Northern, and Luapula Provinces, but was higher in the 2010 survey with parasite rates in young children in Luapula Province exceeding 50% in 2010. Although a substantial decline of 66% in patient cases and deaths occurred in 2000-2008 following the introduction of multiple interventions, a malaria upsurge occurred in 2009-2010 when vector control interventions were disrupted following delays in the disbursement of funds [109]. There has been a 48.3% decline in the national overall malaria prevalence in children under five years of age between 2006 and 2012 (22.1% versus 14.9%) [109].

1.4.1.3 Malaria treatment policy in Zambia

Zambia revised its national malaria drug treatment policy between 2000 and 2005 to adopt artemisinin combination therapy (ACT) as the first-line treatment for uncomplicated malaria [110]. Quinine is the first-line treatment for uncomplicated malaria in pregnancy in the first trimester, whereas artemether-lumefantrine (AL) is for use in second and third trimesters. For complicated malaria, parenteral quinine is recommended in all trimesters; SP is used by policy for IPTp and also serves as a drug of choice for people who cannot tolerate AL and during periods of AL stock-outs in health facilities [111]. The Zambian malaria policy states that IPTp-SP doses are to be administered during pregnancy at scheduled ANC visits, spaced one month apart after 16 weeks of gestation.

1.4.1.4 Prevalence of P. falciparum resistance markers in Zambia

Monitoring SP resistance markers is important in areas where IPTp-SP is policy as these markers have been associated with compromised efficacy of SP for IPTp [1-3].

A few studies conducted in Zambia have estimated the prevalence of SP resistance markers in the general population and in pregnant women. In 2006, the frequency of dihydropteroate synthase (DHPS) and dihydrofolate reductase (DHFR), mutations associated with parasite resistance to SP were found to range from 71-92% for the triple mutant DHFR and 39-71% for the double mutant DHPS in the general population in six districts in Zambia [112]. In another study conducted in 2006 in the southern part of Zambia, the prevalence of the DHFR triple mutant was found at 41.3% and the DHPS double mutant was at 16% in a random sample of 95 selected from 188 malaria positive samples [113]. The frequency of the quintuple mutant was at 6.5%. Higher prevalence estimates of the quintuple mutant have been found in pregnant women in Northern Zambia. In a study conducted between January 2010 and May 2011 in Mansa the prevalence of the quintuple mutant among 84 malaria positive samples was found in 63.1% (n = 53) and 2.3% (n = 2) of these had an additional 581-G mutation [114]. This was the first published record of the sextuple mutant in Zambia. Another study also conducted in February to April 2013 in the Northern part of Zambia, in Nchelenge District the quintuple mutant in a sample of 72 malaria positive pregnant women was found in 17% (n = 12) of the samples. The sextuple mutant was found in 3% (n = 2) of the samples [115].

1.4.2 Prevalence of curable STIs/RTIs in pregnancy in Zambia and neighbouring countries

Studies of pregnant women in Zambia and neighbouring countries conducted between 1990 and 2015 and identified in the literature search have reported prevalence of individual STIs/RTIs with wide ranges between the lowest and highest reported figures. From these studies it has been reported that 0-14.6% of pregnant women have syphilis; 0.5-18.6% have gonorrhoea and 1.7-17.5% have Chlamydia; while the prevalence of trichomoniasis and BV range between 2.4-32.2% and 4.3-37.6% respectively. The median prevalence of individual STIs/RTIs are as follows: syphilis 5.6%, gonorrhoea 2.8%, Chlamydia 6.6%, trichomoniasis 17.4% and BV 24.9%. Appendices 1.2c to 1.2g show point prevalence estimates of STIs/RTIs reported in studies conducted in Zambia and neighbouring countries and includes diagnostic methods used to estimate prevalence.

1.4.2.1 Diagnostic methods for STI/RTIs and published sensitivity and specificity

Diagnostic methods for STIs/RTIs employed in the various studies in Zambia and neighbouring countries and published sensitivity and specificity measures are shown within Appendix 1.3. Methods used for detection of syphilis are rapid plasma reagin (RPR) and Venereal Disease Research Laboratory (VDRL) test and confirmation is conducted using *Treponema pallidum* haemagglutination assay (TPHA), *Treponema pallidum* particle agglutination (TPPA), determine Syphilis *-Treponema pallidum*, microhaemagglutination assay-*Treponema pallidum* (MHA-TP) and immunoglobulin G (ELISA) tests. When RPR is used for syphilis diagnosis and a confirmatory test is conducted both sensitivity and specificity are approximately 100% [6]. When used alone the sensitivity measure is 86-100% for RPR and 78-100% for VDRL and the specificity is 93-98% and 98% for RPR and VDRL respectively [116].

Thirty four prevalence estimates (Appendix 1.2c) of syphilis prevalence were reported from 33 studies, one study reported two separate point prevalence estimates, one for first ANC attendees and another for delivering women. Of these 34 estimates four were based on RPR only and one on VDRL test only. The rest of the estimates were based on either RPR or VDRL and a confirmatory test (TPHA or ELISA or Determine Syphilis TP or MHA-TP) in each case. Estimates obtained from RPR or VDRL only would be expected to be higher than those where a confirmatory test is used. Since the majority of studies used the combination of diagnostic tools known to have the highest sensitivity and specificity, one would expect that the reported estimates of syphilis prevalence among ANC attendees in Zambia and surrounding countries are reliable. Therefore, it can be concluded that the reported prevalence, whether low or high, is mainly due to the actual prevalence in the study setting rather than in differences in diagnostic methods.

The prevalence of Chlamydia among pregnant women in Zambia and surrounding countries has been reported to range between 1.7 and 17.5% (Appendix 1.2d). Cervical swab culture for diagnosis of Chlamydia has been reported to have sensitivity ranging between 74 and 94% while specificity is between 98 and 99%

[117]. Enzyme immunosorbent assay has been widely used for detection of Chlamydia and is reported to have sensitivity ranging from 71-87% and specificity ranging from 97-99% [117]. Urine based PCR has 78-89% sensitivity and 99-100% specificity and cervical swab based PCR has 99% sensitivity and 99-100% specificity [117]. The use of Deoxyribonucleic acid identification assay is reported to have 100% sensitivity and specificity [6]. In the 11 studies reporting the prevalence of Chlamydia among pregnant women in this region, a variety of methods were used to obtain these estimates in these studies as follows; culture was used in one, EIA in four, PCR in two, LCR in two, DNA-ID in one and ELISA was used in one. Based on the range of sensitivity measures for the methods that were used to measure prevalence, it is expected that there was some level of underestimation of the prevalence. However, some of the studies used DNA-ID, PCR and LCR which have higher sensitivity and specificity.

It is reported that 0.5-18.6% of pregnant women in Zambia and surrounding countries have gonorrhoea (Appendix 1.2e). Detection of gonorrhoea using gram stain is reported to have 50-70% sensitivity and 95-100% specificity [118]. Sensitivity and specificity of culture and gram stain (based on mean of culture) have been reported to be approximately 80% and 100%, respectively [6]. Ligase chain reaction with cervical swab has been reported to have the highest sensitivity and specificity, 95-100% and 98-100% respectively, while for PCR with cervical swab sensitivity and specificity have been estimated to be at 89-97% and 94-100%, respectively [118].

Of the 13 studies conducted in Zambia and surrounding countries, 14 point prevalence figures were reported for gonorrhoea among ANC attendees. In these studies culture was used in eight (with one study reporting two culture based estimates), culture and gram stain were used in two, PCR in two and LCR in one. Since the use of culture alone and culture combined with gram stain have lower sensitivity and specificity, and were used in the majority of studies, the individual prevalence estimates would be expected to be slightly lower than the actual prevalence but within acceptable ranges.

The prevalence of trichomoniasis is reported to range between 2.4 and 32.2%. Diagnosis of trichomoniasis has widely been done using wet mount microscopy which has sensitivity ranging from low to moderately high (38-82%) and high specificity (100%) [117]. All of the 16 estimates of trichomoniasis prevalence from 15 studies conducted in Zambia and neighboring countries were obtained using wet mount microscopy, except one estimate from a study conducted in Zambia in which PCR was used. Based on published sensitivity and specificity measures of the wet mount microscopy, it is expected that the actual prevalence of trichomoniasis in these settings were generally underestimated.

Bacterial vaginosis is the most prevalent of the STIs/RTIs with reports of prevalence ranging between and 4.3 and 37.6% among pregnant women of Zambia and surrounding countries. Over the years diagnosis of BV has been done using clue cell count, Amsel criteria [119] and the Nugent score [120]. The sensitivity and specificity of the Amsel criteria are reported to be 51% and 98% [121]. The Nugent method has higher sensitivity (86-89%) than the Amsel criteria but its specificity is reported to be between 94-96% [122-124].

Of the 12 point prevalence estimates from 11 studies reporting BV prevalence in Zambia and neighbouring countries, six were based on the Amsel criteria; four were based on the Nugent criteria and two were based on the clue cell count method. Based on published sensitivity and specificity, the reported prevalence estimates would generally be expected to be lower than the actual prevalence in the study population.

1.4.3 General assessment of prevalence literature

Sensitivity and specificity measures vary among different diagnostic tests. However, a diagnostic test conducted at different times under standard laboratory conditions would be expected to have similar sensitivity and specificity measures. It was noted that similar diagnosed methods have been used over time in the study period (1990-2015) within which prevalence studies reviewed in thesis were conducted. No particular method predominated a certain period in time (e.g. the first 10 or 15 years) among the various diagnostic methods used in the detection of the different infections. In this context time has no influence on the prevalence estimates in the studies that were reviewed. It was also noted that the prevalence figures from studies conducted in the period between 1990 and 2015 do not show any downward or upward trend with time across the various infections from different sites in Zambia and surrounding countries.

In general, methods with low sensitivity would underestimate and those with low specificity would overestimate prevalence. Overall the methods used in the diagnosis of malaria and STIs/RTIs have been widely applied and, therefore, variation in estimates are likely due to true differences in the prevalence of the infections in the different study sites.

The median prevalence estimates obtained from studies in Zambia and neighbouring countries were lower across all the infections, in comparison to pooled prevalence estimates obtained from the meta-analysis by Chico et al. [6] This may be explained by the fact that Chico et al. corrected point estimates to increase the precision of each measure before generating pooled prevalence estimates. The median prevalence for each infection obtained from studies in Zambia and neighbouring countries either fell within (Chlamydia, gonorrhoea and placental malaria) or outside (syphilis, trichomoniasis, BV and peripheral malaria) the 95% CI of the pooled prevalence estimates obtained from the meta-analysis by Chico et al., 2012. This could further be explained by notable differences in number of studies included in the meta-analysis by Chico et al., and in the review conducted in this thesis i.e., syphilis (seven), trichomoniasis (four) and BV (six) and peripheral malaria (12). In the case of placental malaria the difference in the number of studies between the two was minimal (two) and the number of studies was the same in the case of gonorrhoea and Chlamydia. Another possible explanation is that the prevalence of some of the infections is simply higher in some regions/study sites than others.

1.5 Risk factors for STI/RTI and malaria

1.5.1 Risk factors for STIs/RTIs

Studies conducted in Zambia and neighbouring countries reporting prevalence of malaria and STIs/RTIs among pregnant women identified in this thesis were further reviewed for predictors of infections.

It was noted that not all the studies reported risk factors for infection. Some of the studies identified risk factors for individual infections while in others predictors of infection were identified for different combinations of infections e.g. Chlamydia and gonorrhoea [125-127]; HIV, HSV and syphilis [128]; BV, trichomoniasis and candidiasis [128]); syphilis, trichomoniasis and gonorrhoea [129]; HSV-type 2, trichomoniasis, Chlamydia, gonorrhoea, syphilis and HIV [130]. Risk factors for STIs/RTIs and malaria identified in studies conducted in Zambia and neighbouring countries are summarised and presented in Appendix 1.4.

Among these studies syphilis has been associated with the following factors: HIV infection [131], history of stillbirth [132, 133], self-reported previous spontaneous abortion [133], history of past STI [131], genital ulcers [131, 134], vaginal discharge [131], genital warts [131], married status or living with a partner [131, 135], infection with trichomoniasis [134], higher parity and gravidity [132], older age [136], lower education [136, 137], having more than one lifetime partner [128, 133], previous pregnancies [138] and having a partner who takes alcohol [128].

Infection with gonorrhoea was associated with having more than one lifetime partner [139]. Infection with gonorrhoea and/or Chlamydia was associated with following factors in one study; vaginal discharge, having had a new sexual partner in the last three months, being separated from one's partner for more than 3 months, presence of a purulent discharge, lower education, current use of a condom [125]. Other predictors of Chlamydial/gonococcal infection are older age group (above 25 versus \leq 25 [126] and \geq 30 compared to 20-29 group [127]), single status, being in a polygamous marriage, having more than one sexual partner over the previous year and having delivered more than five years earlier [126]. Trichomoniasis has been associated with polygamy [129], lower education level of sexual partner, abnormal discharge, genital ulcer and genital warts [140]. Infection with trichomoniasis and/or BV and/or candidiasis has been associated with early sexual debut (< 20 years) [128].

In a study in which risk factors for composite STI (HSV-type 2, trichomoniasis, Chlamydia, gonorrhoea, syphilis and HIV) were identified, the presence of any STI was associated with being in a long-term relationship as opposed to a short term one, increasing age difference between the girl and her partner and history of prior pregnancy [130].

Geographical location has been shown to be an important risk factor for STIs/RTIs when one area is compared to another, e.g. semi-urban and rural versus urban [138] or one study site versus another [136, 140].

1.5.2 Risk factors for Malaria infection

In the studies conducted in Zambia and neighbouring countries identified in this thesis, the following factors were associated with an increased risk of peripheral malaria; younger age [89, 115, 141], wet season [141] primigravidity/primiparity [142, 143], wealth quintile (all categories of wealth below the highest) [144] and malaria infection earlier in pregnancy [89, 145]. Bed net use was found to be protective against peripheral malaria infection [1].

A higher risk of placental parasitaemia has been associated with younger age [89], primiparity [141, 143, 146] and malaria infection earlier in pregnancy [89]. Iron deficiency [146], bed net use [1] and receiving IPTp-SP during pregnancy [1, 147] have been associated with a reduced risk of placental malaria infection.

1.5.3 Conceptual frameworks for infection

In order to identify both distal and proximal risk and protective factors associated with malaria and STI/RTI, two conceptual frameworks were constructed based on evidence from literature. Under these frameworks (Figures 1.3 and 1.4) it is assumed that demographic and socio-economic factors directly or indirectly (via maternal/partner characteristics) influence the risk of infection.

Figure 1. 3: Conceptual framework for STIs/RTIs



Figure 1. 4: Conceptual framework for malaria infection



As earlier mentioned the prevalence of malaria and STI/RTI has not been reported in any epidemiological setting therefore risk factors for coinfection are not known. It is plausible to assume that distal and proximal risk factors for malaria infection and STIs/RTIs play a role in their coinfection. Figure 1.5 shows a conceptual framework for malaria and STI/RTI coinfection based on risk factors for malaria and STI/RTI identified in literature. Unprotected sex may result in pregnancy and STIs. Pregnancy brings about reduced immunity [43] which makes pregnant women in a malaria endemic area susceptible to malaria infection.



Figure 1. 5: Conceptual frame work for malaria and STI/RTI coinfection

1.6 Prevalence of malaria and STIs/RTIs coinfection

It is clear from literature that there is paucity of information on the prevalence malaria and STIs/RTIs coinfection in pregnancy. In their review, Chico *et al.*, noted that only one study from Tanzania reported the prevalence of malaria among women who tested positive for syphilis [6]. In this retrospective cohort study, it was found that among 73 women with high-titre (\geq 1:8) active syphilis, 34% concomitantly had maternal malaria, 18% had past chronic placental malaria and 48% had histological evidence of current malaria and chronic changes indicative of malaria infection earlier in pregnancy [10]. However, the occurrence of other curable STIs/RTIs was not reported in this paper.

1.7 Current antenatal Care package

1.7.1 Antenatal care package for malaria in pregnancy

Intermittent preventive treatment in pregnancy with SP is recommended by the World Health Organization for reducing the risks associated with malaria in pregnancy [148]. Randomised controlled trials have shown IPTp-SP decreases the risk of maternal anaemia, placental parasitaemia, and LBW [149-152]. In a review of nine trials conducted in Africa IPTp-SP reduced the risk of placental malaria (RR 0.48; 95% CI, 0.35-0.68), LBW (RR 0.71; 95% CI, 0.55-0.92), and anaemia (RR 0.90; 95% CI, 0.81-0.99) [153]. At least one dose of IPTp-SP has been shown to protect against maternal parasitaemia (OR 0.20, 95% CI, 0.12-0.34, P < 0.001) in an observational study in Ghana [154] and to protect against LBW in Benin [155]. Administration of three doses instead of two doses of SP has been shown to reduce the prevalence of placental malaria, LBW, and preterm delivery by 50% in a high malaria transmission area of Mali [156]. In comparison to a 2-dose regimen, three doses of IPTp-SP are superior in reducing the effect of malaria in pregnancy and appear to be effective even in areas with high levels of SP resistance [3]. However, the protective effect of IPTp-SP has been compromised [157] and a decline in IPTp-SP efficacy has been reported in Africa [158]. In an area in Tanzania where 14 days parasitological SP treatment failure rate was as high as 38.8% (95% CI, 26.8-50.8) [159], IPTp-SP was significantly associated with placental parasitaemia (P = 0.03), did not have an effect on LBW [OR 0.71 (95% CI 0.33-1.54) P = 0.39] and was significantly associated with foetal anaemia [1.91 (95% CI 1.14-3.19) P = 0.01] [2]. However, in Malawi the use of IPTp-SP was not associated with higher parasite densities, greater placental inflammation, or worsened delivery outcomes despite the increasing prevalence of SP-resistant P. *falciparum* haplotypes [160]. A recent systematic review and meta-analysis reports no evidence of paucigravidae being protected against delivering a LBW baby in areas where the parasite population of Arg-581 is greater than 10.1%. [161]. Therefore, an evidence-based policy revision is urgently needed.

1.7.2 Antenatal care package for curable STIs/RTIs in pregnancy

Screening and treatment of syphilis is recommended as a routine part of antenatal care [162, 163]. Detection of STIs/RTIs is most accurately done using technologically advanced and expensive diagnostic techniques. These techniques require trained staff, laboratory equipment, specific storage and transportation conditions [5]. In order to assist countries with limited resources, the WHO has developed syndrome-based algorithms for the diagnosis and management of *gonorrhoea*, *Chlamydia*, trichomoniasis and BV in symptomatic individuals. Unfortunately in women high proportions of gonococcal (80%) and Chlamydial (70-75%) infections are asymptomatic [164]. Approximately 50% of women infected with *T. vaginalis* are asymptomatic but even higher proportions (60-70%) of asymptomatic *T. vaginalis* infection and BV have been reported [165, 166]. The implication is that the use of syndromic management would lead to many asymptomatic cases going unnoticed, and thus, a large proportion of STIs/RTIs in pregnancy will remain undetected and untreated.

1.7.3 Antenatal care in Zambia

In Zambia the ANC package includes the following; weight measurement, height measurement, blood pressure measurement, urine sample collection for analysis, blood sample collection for analysis, voluntary counselling and testing, provision of iron supplementation , IPTp-SP, discussion of birth preparedness plan, provision of treatment for intestinal parasites and tetanus toxoid vaccination [167]. Blood testing for screening of maternal syphilis, HIV and anaemia are conducted as part of ANC [168]. Although syndromic management is not documented in the National HIV/AIDS/STI/TB Policy [169]. Therefore in practise the syndromic approach is used for diagnosis of other STIs whenever women attending ANC come with complaints of signs and symptoms that are indicative of STIs.

In a study where two national datasets with detailed antenatal care provider and user information i.e. the 2005 Zambia Health Facility Census and the 2007 Zambia Demographic and Health Survey (ZDHS), were analyzed to describe the level of ANC service provision at 1,299 antenatal facilities in 2005 and the quality of ANC received by 4,148 mothers between 2002 and 2007 [170]. In the study mentioned above, criteria were developed to describe quality of ANC and different levels of provision of ANC. This included information on ANC use, type of ANC provider, place of ANC provision, number of ANC visits, timing of first ANC visit and on the interventions received. Having received "good quality ANC" was defined as having attended at least the recommended four ANC visits with a skilled provider and received at least eight antenatal interventions, while the definition of "moderate quality ANC" required four visits with a skilled provider and five to seven antenatal interventions [170]. Only 45 antenatal facilities (3%) fulfilled their developed criteria for optimum ANC service, while 47% of facilities provided adequate service, and the remaining 50% offered inadequate service. Although 94% of mothers reported at least one ANC visit with a skilled health worker and 60% attended at least four visits, only 29% of mothers received good quality ANC, and only 8% of mothers received good quality ANC and attended in the first trimester [170]. Results from this study showed that "effective ANC delivery", as estimated by their indicators, was way below ANC coverage.

Percent distribution of women aged 15-49 who had a live birth in the five years preceding the 2013-14 ZDHS by number of antenatal care (ANC) visits for the most recent live birth were slightly different between women in the urban and rural areas of Zambia. In the urban areas 0.9%, 1.6%, 40.7%, 55.9% and 1% of women aged 15-49 attended none, one, two to three and four or more ANC visits respectively. In the rural areas distribution was as follows: 1.7%, 2.0%, 40.2%, 55.2% attended none, one, two to three and four or more ANC visits respectively. The median gestational age at first ANC was 4.8 months and 4.7 months among urban and rural women [167] respectively.

The proportions of ANC visits among Zambian women seem to suggest that a good number of women do not receive adequate ANC interventions indicating lost opportunity for delivering interventions.

1.8 Birthing practices and prevalence of adverse pregnancy outcome in Zambia

1.8.1 Practices and beliefs about pregnancy and childbirth in Zambia

In Zambia western religious practices are widespread and Christianity co-exist with deep spirituality of local tribes [171]. In general there is belief in the Christian God and the spirit realm of ancestors, which has allowed for the acceptance of the existence of witchcraft [171]. Beliefs in witchcraft are often applied to pregnancy, labour and birthing practices.

Zambian women often deliver in seclusion or in the company of a few selected women at home [172]. Several factors contribute to the preference of a secretive home delivery. In general pregnancy due dates and the start of labour are kept secret for fear of witchcraft harming the child and women deliver with a few people present to avoid supernatural harm to the child [173]. Some of the beliefs contribute to the proportions of women who deliver at home as opposed to making use of health professionals and facilities. A populationbased cross sectional study survey by Mwewa and Michelo conducted among women (n= 499) of Nchelenge District in 2008 to identify factors associated with home deliveries showed that the prevalence of home deliveries was 43%, 95% CI (38.62, 47.48) [174]. Among the factors cited for home deliveries were long distances to the health facilities coupled with lack of transport, abrupt unexpected labour, circumstances beyond control (such as embarrassment due to not having the required items that delivering mothers are expected to present with at health facilities) and myths and traditional beliefs [174].

The study also identified some practices that are a result of myths and traditional beliefs that prevent women from delivering at health facilities. Among the common traditional beliefs pointed out in the study by Mwewa and Michelo is what is locally called "incila". According to information from a focus group discussion most women decided to deliver at home in order to be attended by their grandmothers and be treated for "incila", which according to the participants, was a situation where the partner or husband of the pregnant woman or the pregnant woman herself engaged in sexual affairs with other people during the woman's pregnancy, as a result of which the woman could have difficulties at delivery [174].

It was stated by a focus group discussion participant aged 56 from Kabuta village in the study by Mwewa and Michelo that 'The process of preventing death by "incila" requires the woman to divulge confidential information to the women assisting her delivery so that she delivers well in addition to taking the medication and so some women even go further away from their own villages to other villages for confidential reasons because some traditional birth attendants (TBAs) do not keep secrets. If there were too many men that the woman may have had extramarital affairs with while pregnant and they could not all be counted, then she is required to put maize meal into a bowl as a gift, for everything to end there and this cannot be done at the clinics'.

A 59 years old male focus group discussion participants from Kambwali village stated that 'The other reason why women prefer to deliver at home is to have their babies protected from "icifutato", which is a situation where the baby may die if the father of the child recommences sex with a different woman other than his spouse who has not even recovered and healed after delivery of the child. So the women will prefer to deliver at home so that the baby after being born is bathed in water that is medicated to prevent death by "icifutato". There are other traditional beliefs identified in this population from the study mentioned above (including the use of sorcery out of envy and malice to bring doom upon the couple) that prevent women from delivering at health centres as a way of ensuring the "wellbeing" of the mother and the child by carrying out certain traditional practices which cannot be carried out at health facilities.

In another study conducted in Kalomo, Southern Zambia by Sialubanje *et al.*, a negative attitude towards nurses; low risk perception regarding their personal susceptibility to pregnancy and labour complications; dependence on husbands for financial support and decision making; social cultural norms (unwillingness to be delivered by a young nurse or a male staff of the health facility) were cited among reasons why women delivered at home [175]. Furthermore trust in the TBAs and being familiar with TBAs were given as reasons that encouraged women to deliver at home [175].

In contrast to low risk perception regarding pregnancy and labour complications, over 80% of young focus group discussion respondents, a few older women (mainly those who had given birth at a health facility), and all the midwives explained that most young women delivered at the health facility because they had no experience giving birth and that they were afraid of labour complications if they gave birth at home [175]. Although the study by Mwewa and Michelo did not explore reasons that influenced women to deliver at a Health facility as in the case of the study by Sialubanje *et al.*, it is likely that women who deliver at

health facilities in the northern part of Zambia are encouraged to do so for similar reasons.

In 2010 the Zambian Ministry of Health announced that it had no intention of training TBAs because certain situations which expectant mothers faced required the attention of qualified medical personnel [176]. The ministry charged that TBAs contributed to high levels of maternal mortality rate in the country hence the need for expectant mothers to deliver from health centres. The Ministry of Health carried out massive sensitisation campaigns on the dangers of delivering from home through safe motherhood action groups [176]. In the study by Sialubanje *et al.*, all Focus group discussions and in-depth interviews were aware of the new policy but some TBAs in the focus discussion groups argued that women still delivered at home and some TBAs still conducted deliveries [175].

1.8.2 Prevalence of adverse birth outcomes in Zambia

The 2013-14 ZDHS reported the prevalence of LBW among deliveries at 10% in Zambia based on figures from Lusaka, Copperbelt and Southern Provinces [167]. A similar estimate was found in the Zambia Electronic Perinatal Record System (ZEPRS) study conducted across 25 facilities in the Lusaka public health sector from June 2007 to January, 2010. Records of 115,552 pregnant women showed that the proportion of LBW and PD was 10.8% and 34.0% respectively, and the incidence of stillbirth was at 24 per 1000 deliveries [177].

Figure 1. 6: Conceptual framework for adverse birth outcomes



Adopted from Olusanya and Ofovwe, 2010 [178]

Adverse birth outcomes that are generally described in literature in relation to curable STIs/RTIs and malaria and are of interest in this study are IUGR, PD, stillbirth and LBW.

In order to identify both distal and proximal risk factors associated with adverse birth outcomes an approach similar to the conceptual framework described by Olusanya and Ofovwe, 2009 [178] (Figure 1.1) was adopted. Under this framework it is assumed that demographic and social economic status directly or indirectly (via maternal and infant characteristics) influence birth outcomes. Maternal obstetric characteristics have direct impact on infant status. The variables of interest have been guided by evidence from published literature. Maternal socio-demographics included age, marital status, education, economic status, number of sexual partners and age at sexual debut. Maternal factors included height, past history of stillbirth, miscarriage and PD, hypertension, anaemia, infection with curable STIs/RTIs, malaria and HIV. Infant factors included IUGR and sex. The sex of the baby is an independent determinant of ABO.

1.9 Rationale and objectives

1.9.1 Rationale

The individual prevalence of malaria and curable STIs/RTIs among antenatal care attendees in sub-Saharan Africa is considerable with the combined prevalence of curable STIs/RTIs being equal to, if not greater than malaria [6]. However, the prevalence of curable STIs/RTIs and malaria coinfection in pregnancy is entirely unknown. Despite the interest in the associations between malaria and birth outcomes, and also between curable STIs/RTIs and birth outcomes, no study has assessed the occurrence of malaria and curable STIs/RTIs coinfection and identified its risk factors.

Currently, the recommended antenatal care package includes IPTp with SP for the control of malaria, screening and treatment of syphilis, and syndromic management for other curable STIs/RTIs. However, in the context of emerging malaria parasite resistance to SP and the limited accuracy of the syndromic management, new strategies to control these infections in pregnancy are needed. One of the strategies proposed is preventive therapy that is safe and effective against malaria and curable STIs/RTIs. Azithromycin–based combination therapy offers these properties and is a possible candidate replacement for SP [179]. Thus characterising the prevalence of malaria, curable STIs/RTIs and their coinfection in pregnant women will not only fill this gap, but will also provide evidence in support of an integrated solution for prevention and control of these infections in pregnancy.

Despite the individual public health importance of malaria and curable STIs/RTIs, their impact on pregnancy outcomes when they occur together has not been assessed.

Measuring molecular markers of SP resistance is proposed to monitor the efficacy of IPTp-SP. Although the frequency of SP resistance markers at which IPTp would cease to be effective is unknown, monitoring trends of SP resistance markers is important, particularly in pregnant women where this information is limited [180]. Furthermore this information may support the need for finding alternative drugs for IPTp that may be effective against malaria and common STIs/RTIs.

Therefore this study has the potential to contribute to policy revision in the interest of reducing the public health impact of malaria and STIs/RTIs in pregnancy.

1.9.2 Objectives

The primary objective of this thesis was to estimate the prevalence of malaria, curable STIs/RTIs, and their coinfection among pregnant Zambian women.

Secondary objectives were as follows:

- 1. To identify risk factors for malaria, curable STIs/RTIs and their coinfection among pregnant Zambia women.
- To determine the prevalence of DHFR and DHPS mutations associated with SP resistance among pregnant women.
- 3. To measure the *in vivo* efficacy and therapeutic failure of SP in pregnant women over a period of 28 days post-IPT.
- 4. To estimate the prevalence of ABO and identify their risk factors among pregnant Zambian women.

2 Methods

2.1 Chapter introduction

This chapter presents all the methods that were used in the study including design, sample size calculation, inclusion and exclusion criteria, enrolment procedure, data and sample collection, laboratory procedures and analyses.

2.1.1 Study design and sample size

2.1.2 Study design

This was a prospective cohort study of antenatal attendees. Women were enrolled and observed alongside routine ANC activities. The study involved the following: (i) Screening of women for malaria and common curable STIs/RTIs at first antenatal visit; (ii) screening of women for malaria at day 28 post IPTp-SP, and (iii) following them up to delivery to capture delivery characteristics. Figure 2.1 shows the flow of research activities that were conducted.

Figure 2. 1 Flow-chart of research activities



Consenting eligible pregnant women were recruited at their first ANC visit. Samples were collected for screening of curable STIs, bacterial vaginosis and malaria. Participants (at >16 weeks gestation) were given their first dose of SP for intermittent preventive treatment and demographic and socio-economic data was collected. Women were followed up after 28 days and screened for malaria and given a second dose of SP. At delivery placental tissue was collected for the diagnosis of placental malaria and birth characteristics, maternal factors and medical and drug histories were recorded.

2.1.3 Sample size calculation

The systematic review and meta-analysis by Chico *et al.* reported the following pooled prevalence estimates of curable STIs/RTIs and malaria from East and Southern Africa : syphilis 4.5%; gonorrhoea 3.7%; Chlamydia 6.9%; trichomoniasis 29.1%; BV 50.8%; placental malaria 25.8%; and peripheral malaria 32.0% [6]. As there were no data on the prevalence of malaria and curable STIs/RTIs coinfection from previous studies, it was assumed that the prevalence of malaria and any one curable STI/RTI is 10%. To measure a 10% prevalence with a 95% confidence limit around a +/-3% precision, a sample size of 843 women is required to estimate the proportion of pregnant women coinfected with malaria and at least one of the curable STIs/RTIs.

The 2013-14 Zambia Demographic Health Survey reported that the prevalence of LBW among deliveries in Zambia was 9% [181]. A similar estimate was found in the Zambia Electronic Perinatal Record System study conducted across 25 facilities in the capital city (Lusaka) public health sector from June 2007 to January, 2010. Records of 115,552 pregnant women showed that the proportion of LBW was 10.8%, PD was 34%, and the incidence of stillbirth was at 24 per 1000 deliveries [177]. Thus, to explore the determinants of ABO, it was assumed that 10% of the women would experience at least one ABO (PD, LBW, IUGR and stillbirth).

The incidence of ABO among women detected to have curable STIs/RTIs or malaria during antenatal period was expected to be higher than that among uninfected women. Thus the incidence of ABO in the unexposed women was assumed to be 7.5%. A study with 80% power and 95% confidence to detect risk factors that have an OR > 2 requires 984 women. To account for losses to follow up and withdrawals from the study, the sample size was increased by 10% [182]. Therefore, 1,086 women were recruited. Table 2-1 shows different assumptions and sample size calculations for the determination of risk factors for ABO.

Power	Proportion of unexposed with ABO	Odds ratios								
		1.5			2.0			2.5		
		Unexp.	exp	total	Unexp.	exp	total	Unexp.	exp	Total
70%	5	2688	896	3584	825	275	1100	438	148	584
	7.5	1866	622	2480	579	193	722	309	103	412
	10	1458	486	1944	459	153	612	249	83	322
80%	5	3423	1141	4564	1053	351	1404	558	186	744
	7.5	2373	791	3164	738	246	984	393	131	524
	10	1854	618	2472	582	194	776	315	105	420
90%	5	4599	1533	6132	1416	472	1888	750	250	1000
	7.5	3186	1062	4248	990	330	1320	528	176	704
	10	2487	829	3316	780	260	1,040	420	140	560

Table 2- 1: Sample size calculation for determination of risk factors for adverse birth outcomes

Unexp: Unexposed

Exp: Exposed

Sample size was calculated with 95% confidence and a 3:1 ratio of unexposed to expose. Exposure was infection with malaria, any STI/RTI or both.

To estimate the prevalence of SP resistance markers among pregnant women the sample size was estimated assuming that the prevalence of the quintuple mutation among pregnant women was 50% due to limited data. To detect a prevalence of 50% with 95% CIs of +/-10%, a sample size of 96 malaria positive women was needed.

The sample size needed to measure therapeutic outcomes in pregnant women receiving IPTp-SP was calculated based on the WHO treatment efficacy protocol which calculates sample size based on the expected proportion of treatment failures [183]. The proportion of treatment failure among pregnant Zambian women is unknown. It was assumed that the proportion of therapeutic failures was 10%. To detect a 10% failure with 95% CIs of +/-5%, 138 asymptomatic malaria positive women were required.

2.2 Study area and sampling

2.2.1 Study area

Nchelenge is a rural district located in the Northern part of Zambia in the Luapula Province on the shores of Lake Mweru (Figure 2.2A) and has a population of 173,680 [184]. The district has a tropical climate with a rainy season (November to April) and a dry season (May to October).

Nchelenge District has 10 health centres, three health posts and one first level hospital [185]. The district was selected on the basis that it has a high prevalence of malaria and the prevalence of the individual STIs/RTIs of interest was expected to be moderate to high based on indications from RPR seropositives and HIV-infected women [186]. Luapula province has the highest prevalence of malaria in children under the age of five years with a prevalence of 38% by microscopy and 56% by rapid diagnostic test [187]. In a study conducted in March to April, 2012 in Nchelenge District among 782 children less than 10 years of age parasite prevalence by microscopy was reported at 30.2% (n = 236) [188]. Between 1990 and 2015, only two published studies both conducted in Luapula Province, have estimated the prevalence of malaria among pregnant women. One is a study conducted in Mansa in 2009 to 2010 and placental tissue samples of 37% (162/435) of delivering women showed histological evidence for malaria infection during pregnancy [90]. In another study conducted in 2013 in Nchelenge District, 22% of ANC attendees were positive for peripheral malaria by PCR [115].The prevalence of RPR-positive women at first ANC attendance was estimated at 10.0% [186]. The prevalence of HIV among ANC attendees at Kashikishi and Nchelenge health centres, in 2013 was 13.0% and 13.3%, respectively [186]. These estimates combined new cases and already known HIV positive cases.

Although other economic activities exist in Nchelenge District, the inhabitants are mainly fishermen and/or farmers. The population focuses on farming during periods of the fishing ban which is in effect from 1st December to 1st March each year.

The study was conducted in a catchment area of two health centres, Kashikishi and Nchelenge Health centres, with a total catchment population of 44,115 [185]. The community in the catchment area of two health centres, which served as recruitment sites, was selected due to the close location of the two health centres to the only hospital in the district. Figure 2.2B shows the location and the catchment area of the two health centres.

According to the 2013-14 Zambia Demographic and Health Survey, 94.6% of women aged between 15 and 49 in Luapula Province reported having received ANC from a skilled health care provider during pregnancy for their most recent birth in the preceding five-year period [167]. The coverage of IPTp-SP reported in the MIS at Provincial level was as follows; 89.7%, 76.6% and 57.6 % of women took at least one dose, two or more doses, and 3 or more doses of IPTp-SP, respectively [189]. At the time of the study, Kashikishi health centre offered laboratory, ANC and maternity services while Nchelenge Health Centre offered ANC services but not maternity services and depended on Kashikishi Health Centre for laboratory services.

The only hospital in the area is Saint Paul's Missionary Hospital. In 2013, 2620 deliveries were recorded at the hospital, 306 (11.7%) by caesarean section and 2314 were normal births. Of the recorded births, 2556 were live births and 64 (2.4%) were stillbirths, 29 (1.1%) macerated and 35 (1.3%) fresh stillbirths. Of the 2556 live births 564 (21.5%) were born with a LBW (\leq 2500). The system does not capture the number of preterm deliveries.



Figure 2. 2: Map showing the position of Nchelenge district in Zambia and the study area

Key: A-Blue area indicates the area within the borders of Zambia covered by Lake Mweru. Purple area indicates the location of Nchelenge District. B-Blue triangles indicates homes of women recruited at Kashikishi Health Centre, red dots indicate homes of women recruited at Nchelenge Health Centre, the two purple squares indicate the location of Kashikishi and Nchelenge health centres. Locations of participant homes and health centres were plotted using global positioning system coordinates. Maps: Courtesy of Ministry of Lands, Lusaka, Zambia
2.2.2 Sampling

Women visiting the two selected Health Centres for their first antenatal care visit between November 2013 and April 2014 were invited to join the study and enrolled consecutively until the desired sample size was reached. The participants who took IPTp-SP at enrolment were followed up at day 28 and screened for peripheral malaria. All women enrolled in the cohort were then followed to delivery.

2.3 Inclusion and exclusion criteria

2.3.1 Inclusion criteria

- Women of all ages attending the ANC for their first visit
- Gestational age ≤ 32 weeks
- Willing to take part in the study and give consent
- Agree to have their HIV test result from routine counselling and testing recorded by a member of the research team

2.3.2 Exclusion criteria

• Women reporting to have taken antimalarial and/or antibiotic treatment within the previous 4 weeks

• Women aged below 18 years without a guardian present to give written consent

2.4 Enrolment and antenatal care package

Enrolment and biological sample collections were conducted alongside the provision of routine ANC. Booking and ANC revisits were done once each on different days of the week at each health centre. Each week the ANC clinic for booking started with a health talk which covered various aspects of ANC including nutrition, curable STIs, HIV and malaria. At the end of the health talk the study was introduced to the whole group of first ANC attendees and their partners. It is worth noting that in the community where this study was conducted, attending ANC booking with spouses/partners has been encouraged over the years as a way of improving maternal health especially when dealing with issues relating to HIV and this was generally accepted by community members. Therefore, women and their partners commonly undergo counselling and testing together.

Height, weight and blood pressure of each participant were measured and recorded on the ANC cards. HIV testing was routinely conducted using finger-prick blood with two rapid test kits (Determine[®] HIV-1/2 [Abbott Diagnostic Division, Hoofddorp, The Netherlands] and Uni-Gold[™] Recombigen[®] HIV-1/2 [Trinity Biotech USA Inc. New York, USA]). Women were generally tested together with their spouses/partners. Individuals found to be seropositive using Determine[®] HIV-1/2 were then tested with Uni-Gold[™]; HIV status was concluded based on the result of the second test. Before determination of gestational age, ANC attendees were individually given copies of the information sheet which had been translated into Bemba (Appendix 2.1). Upon determination of a gestational age \leq 32 weeks by last menstrual period and fundal height assessment, eligibility screening was completed by filling out the eligibility screening form (Appendix 2.2). Eligible women were then requested to give written consent, which they did after their assent, by signing two copies of the consent form (Appendix 2.3) or appending a thumbprint. In the case of women aged below 18 years, written consent was sort from their guardian if present after their assent. One copy was kept by the research team and a second copy was given to the participant together with a copy of the information sheet. Participants were at this point assigned an identification number which was used to anonymize all the paperwork and samples collected from a particular participant. A record of the participants' names and Identification number was kept separately by the study team. All study materials were kept in a private office and were only accessible to authorised research team members.

As part of the antenatal care package, all ANC attendees were given iron and folic acid supplements to last them up to their next ANC visit. First ANC attendees who had complaints indicative of STI/RTI were further assessed by a clinician for decisions on treatment. Women and/or their partners who tested positive for HIV for the first time were also assessed for eligibility to receive antiretroviral therapy by their regular clinician. All women with \geq 16 weeks gestation were given IPTp-SP except those on cotrimoxazole prophylaxis due to toxicity concerns.

2.5 Data and sample collection

2.5.1 Data and sample collection at recruitment

One vaginal swab and one cervico-vaginal swab were collected from each participant by a trained midwife. Non-lubricated vaginal specula were used to collect the samples. The vaginal swab was collected using a dry cotton swab. The swab was placed in the vaginal cavity for 10 seconds and immediately rolled onto a slide for diagnosis of BV. A Dacron[®] (Medical Wire & Equipment, Wiltshire, England, United Kingdom) was used to collect a cervico-vaginal sample for the detection of *C. trachomatis*, *N. gonorrhoea* and *T. vaginalis* by PCR. The cervico-vaginal swabs for PCR were placed in cryo vials (Narang Medical Limited, New Delhi, India), labelled with each participant's ID and date of collection. The swabs from each participant were transported and stored at -20^oC for a few hours pending extraction which was conducted within 24 hours.

A questionnaire (Appendix 2.4) was administered to each of the enrolled women by a trained field worker in a private room. Gestational age was then measured by ultrasound using crown-rump length in first trimester pregnancies and a combination of biometric parameters (biparietal diameter, head circumference, abdominal circumference, and femur length) in the second and third trimester pregnancies. Women with a gestational age > 32 weeks were excluded at this point and counted among those who were ineligible. Peripheral and venous blood (approximately 4ml) was collected from each participant for malaria and syphilis screening, respectively. Peripheral blood was collected via finger pricking and venous blood was collected from the arm. Approximately 160µl of peripheral blood was placed on four circles of Whatman[®] filter paper and labelled with the participant ID number. The thick smear and blood spots were later used for the diagnosis of malaria by microscopy and polymerase chain reaction (PCR), respectively.

2.5.2 Collection of global position system coordinates

Landmarks near the homes of participating women and home addresses of all the participants were collected at enrolment. Members of the research team visited the homes of participants guided by the information collected at recruitment. Permission to visit a participant's home was obtained at recruitment. The reason for

the visit was to collect global positioning system coordinates of participants' homes for the purpose of locating women at the time of delivery if they required transportation to the hospital.

2.5.3 Follow-up at day 28

This part of the study required 138 asymptomatic malaria-positive women in order to determine prophylactic and therapeutic failure of IPTp-SP over a period of 28 days after SP administration under direct observation. Only women who were eligible to receive SP i.e., those who presented with a gestational age \geq 16 weeks and were not currently receiving cotrimoxazole were included in this part of the study. At the follow up visit at day 28, peripheral blood for a thick smear and for PCR diagnosis of malaria was collected from participants who had taken SP at day 0. Blood sample collection was only done after verbal confirmation that a participant had not received anti-malarial treatment besides the IPTp-SP given at the time of enrolment. The second dose of SP was administered and efforts to collect samples from participants who did not show up 28 days post-IPT were made through home visits on the same day. Participants who developed malaria symptoms before 28 days were tested and treated by the health centre staff following national guidelines as routinely done. Such cases were not followed up at day 28 post-IPTp-SP.

2.5.4 Data and sample collection at delivery

Participants were followed to delivery and placental blood and tissue were collected by midwives. The midwives at the Kashikishi Health centre and Saint Paul's Missionary Hospital were informed of the study from the start and were trained on how to identify participants from their ANC cards/records and collect placental samples and delivery characteristics. Instructions on sample collection were also made available to the maternity ward staff. Placental blood for the measurement of haemoglobin was collected by making an incision in placental tissue using a blade and blood was drawn using a syringe and immediately placed in Ethylenediaminetetraacetic acid tubes for determination of haemoglobin level. Haemoglobin was read using an ABX Micros ES60 (HORIBA ABX SAS). Placental tissue samples were collected as follows: A 2cm x 2cm x 1cm specimen was taken from the

maternal side, and placed in 10% formalin filled bottles labelled with the patient's ID number and date of collection. Delivery characteristics and drug history were collected at delivery using the second part of the questionnaire (Appendix 2.4).

New born babies were weighed and birth characteristics recorded. Stillbirth was defined as foetal death at \geq 28 weeks gestation [190], PD as delivery at < 37 weeks, LBW as birthweight < 2500 grams and IUGR was defined as an infant with LBW born at \geq 37 weeks gestation [36, 61]. Gestational age after birth was recorded based on earlier assessment at enrolment by ultrasound.

To ensure that major outcomes were captured in most of the deliveries, two mobile phone numbers belonging to team members were given to participants to call at the time they went into labour and required transport to the hospital/health centre. Those who called for transport were located with the help of global positioning system coordinates collected earlier in the study. Participants were also urged to call if they experienced a delivery, late miscarriage or stillbirth at home. A driver and a fieldworker followed-up such cases to collect samples and data at delivery and/or provide transport to the hospital where the mothers were attended to and samples and data were collected. A study completion form was filled out for each participants and their completion of the study or loss to follow-up was documented (Appendix 2.5)

2.6 Laboratory testing and quality control

2.6.1 Syphilis testing and treatment of seropositives

Diagnosis of syphilis by rapid plasma reagin (RPR) for samples collected at both Nchelenge and Kashikishi Health centres was conducted at Kashikishi Health Centre laboratory. Syphilis testing was done using rapid plasma reagin (RPR) (Omega Diagnostics Limited, Alva, Scotland, United Kingdom). Since it was not possible to have RPR test results on the same day, results for syphilis test were ready the day after sample collection. Seropositive women were followed up (incorporated within the global positioning coordinate collection exercise explained above) within three days and presented with a note to return to the health centre with their partner for treatment with benzathine penicillin G (2.4 million units IM weekly x 3 doses) according to the national STI policy. Those who did not show up were started on treatment at the day 28 follow-up if they returned to the health centre. A second note was delivered to those or the homes of those who did not return at day 28 for them to return to the health centre with their partner for treatment.

For quality control 55/1077 (5%) serum samples were taken to the national STI reference laboratory at the University Teaching Hospital (UTH). Testing of all RPR seropositive samples with *Treponema pallidum* haemagglutination assay (TPHA) for confirmation of syphilis and RPR reactivity at 1:8 titre were also conducted at the national STI laboratory.

2.6.2 Malaria diagnosis by microscopy

Thick blood films were stained using 10% Giemsa at Kashikishi Health Centre Laboratory. Slides were there after transported to the National Malaria Control Centre for reading by two independent microscopists. Parasite density was determined by assuming 8000 white blood cells (WBCs) per μ l and counting the number of parasites per 200 WBCs. In cases where fewer than nine parasites were counted against 200WBCs, parasites were counted against 500WBCs. Two-hundred high power fields were read before declaring a slide negative. In the event that discrepant results were observed between the two microscopists, a third observer read the slide to reach an agreement on diagnosis and parasite count.

2.6.3 Diagnosis of placental malaria by histology

Formalin fixed placental tissue specimens were sent to the UTH histopathology laboratory for processing. Fixed placental biopsies were wax embedded, and two sections were stained with Gurr's modified Giemsa and another two with haematoxylin and eosin. Slides were examined by a qualified pathologist for the presence of parasites, pigment or both. Placental histology was classified as described by Rogerson *et al*, 2003 [191]. The classes were as follows: (1) Parasites, no pigment in monocytes or fibrin; (2) Parasites, pigment in monocytes +/- fibrin Parasites; (3) Parasites, pigment in fibrin, (4) No parasites, pigment only (past infection) and (5) No parasites or pigment (no infection). For quality control, the reading of 6.5% (46/702) randomly selected slides was repeated by a second observer. Any disagreements were resolved by a third observer.

2.6.4 Diagnosis of bacterial vaginosis by microscopy

The vaginal smear samples for the diagnosis of BV were air dried and Gram stained using safranin as a counter stain. Gram staining of slides was conducted at the St. Paul's Mission Hospital laboratory and thereafter slides were transported to the University of Zambia for reading. Results were classified based on the Nugent criteria [120] and recorded in the appropriate result record form (Appendix 2.6). For quality control (5%) 55/1085 randomly selected slides were transported to the UTH Microbiology laboratory for repeat reading.

2.6.5 Nucleic acid amplification

In this study, in-house standard end-point PCR assays were employed to detect the presence of the *C. trachomatis, N. gonorrhoea* and *T. vaginalis* in cervico-vaginal swabs; and *P. falciparum* in dried peripheral blood spots.

DNA extraction from cervico-vaginal swabs was conducted within 24 hours of collection and the extracts from each sample were stored at -20^oC pending molecular detection by PCR. Samples were then transported on ice to the Tropical Gastroenterology and Nutrition Group laboratory at UTH where molecular detection of *C. trachomatis, N. gonorrhoea* and *T. vaginalis* was conducted. In the case of *P. falciparum,* dried blood spots were transported to the Tropical Gastroenterology and Nutrition Group laboratory detection by PCR.

2.6.5.1 PCR primers

Plasmodium falciparum was detected using the Snounou primers [94]. For the detection of *N. gonorrhoea*, primers targeting the *orf1* gene (*Ngu1* and *Ngu2*) were employed to amplify a 260 base pair (bp) product specific for *N. gonorrhoea* [192]. Primers KL1 and KL2 were used to amplify a Chlamydial plasmid 241 bp fragment for the detection of *C. trachomatis* [193]. Plasmid based PCR reactions for the detection of *C. trachomatis* have been shown to be 10 to 1000 more sensitive than procedures

which use the major outer membrane protein and ribosomal DNA as templates for PCR amplification [194, 195] and have been used for the detection of *C. trachomatis* in clinical samples [196]

Primers based on *T. vaginalis* repeated DNA target for PCR identification have been used for the molecular detection of *T. vaginalis* [197, 198]. The primer pair, TV3 and TV7, was used to amplify a 300bp piece of the genome. This primer set has been demonstrated to have higher sensitivity than other primer sets [199]. The primer names and sequences for these four organisms are as shown in Appendix 2.7.

2.6.5.2 DNA extraction and PCR for malaria parasite detection

Filter paper containing blood spots were air dried and individually stored in envelops at room temperature pending DNA extraction for the diagnosis of malaria by PCR. Parasite DNA extraction was carried out using the chelex method as described previously [200]. The extract was transferred in approximately 100 μ l aliquots to storage vials and were stored -20 ⁰C pending PCR.

The detection of *P. falciparum* was carried out using a nested PCR method as described by Snounou *et al.*, [201] with modifications to the PCR parameters [202]. Briefly, all PCR reactions were carried out in total volumes of 25µl using Thermo Scientific[®] Dream Taq PCR Master Mix (2X) and 0.5µM of each primer and 2µl of template. A 2µl of the primary amplicon was used as a template in the secondary reaction. The PCR parameters used for both the primary and secondary reactions were as shown in Appendix 2.8.

2.6.5.3 Detection of SP resistance markers

To estimate the prevalence of SP resistance among first ANC attendees, a random sample of 96 was selected from among the *P. falciparum* malaria positive samples detected by PCR. Ninety-six was the required sample size for the characterisation of SP resistance markers in the study area. Therefore, only 96 samples were processed for this part of the study.

Parasite DNA template extracted earlier for malaria detection was used for this part of the study. Nested PCR and enzyme restriction digestion as previously described [203] were employed to detect antifolate polymorphisms in the DHFR and DHPS genes. Names and sequences of the primers that were used are shown in Appendix 2.9. The names of the enzymes that were employed and their expected activity depending on the whether the amplicon carried the wild type or mutant strain were as shown in the Appendix 2.10.

Negative and positive controls were included in every batch and put through the entire process from amplification electrophoresis. Secondary PCR amplicons were analysed by gel electrophoresis to confirm amplification and band intensity before enzyme digestion. Amplicon and restriction fragments were analysed on ethidium bromide agarose gels and visualised under UV light on a Biosens (Genescope V1.76) digital imaging system.

2.6.5.4 Determination of *in vivo* efficacy and prophylactic effect of sulphadoxinepyrimethamine

The pair of blood samples collected at recruitment and 28 days post-IPT visit were used to determine the in vivo efficacy to clear parasitaemia and the post-IPT prophylactic effect of SP. The sample size required for this part of the study was 138 asymptomatic malaria positive pregnant women. A woman with a malaria positive blood smear at first ANC visit and found to have a malaria positive blood slide at day 28 post-IPT visit, was classified as a case of therapeutic failure of SP. It was assumed that after a participant with parasitaemia initially receives SP, subsequent detection of parasitaemia 28 days post-IPT was caused by either parasite strains present before SP administration (recrudescence) or parasite strains acquired after treatment (reinfection). Therefore, therapeutic failure cases were further classified as re-infection or recrudescence by PCR analysis. To differentiate between recrudescence and re-infection in women who tested positive at both the initial (day 0) and second screening (day 28), genotyping was done to examine genes coding for the merozoite surface protein 2 (MSP2) [204, 205]. Women who had a negative blood slide at visit one and became slide positive at day 28 post-IPT were classified as cases of prophylactic failure of IPTp-SP.

Polymorphic regions of MSP2 were amplified by nested PCR. The primary PCR primers corresponding to the conserved sequence flanking this region and the secondary PCR primers were used to amplify the IC3D7 and FC27 allelic families of

msp-2. For controls DNA from HB3 and 3D7 laboratory strains was used. Appendix 2.11 presents the sequences of the primers that were used for this part of the study. The secondary amplicon from each sample was analysed using electrophoresis on 2% ethidium bromide stained agarose gels. Samples from an individual participant were loaded in adjacent lanes. In cases where there was no amplification, PCR was repeated using 3 times the quantity of template DNA. In cases where no amplicon was detected after the second reaction, amplification was classified unsuccessful. Alleles were considered the same for day 0 and day 28 within 10 base pairs. Reinfection was defined as having completely different alleles between parasites from day 0 and day 28. If a similar allele was found between parasites at 0 and at day 28 this was classified as recrudescence. Comparisons of band sizes were done using a manual method by two different qualified molecular biologists and any discrepancies were settled by a third person.

2.6.5.5 DNA Extraction and detection of curable STIs/RTIs by PCR

Extraction of DNA from cervico-vaginal swabs was done within 24 hours of sample collection using the Quick-gDNA[™] (Zymo Research Corp, California, USA) Miniprep kit following the manufacturer's instructions. The extract was stored at -20⁰C awaiting nucleic acid amplification.

PCR reactions for the detection of the three organisms, *C* trachomatis, *N*. gonorrhoea and *T*. vaginalis were set up separately for each sample. All PCR reactions were carried out in total volumes of 25µl using Thermo Scientific[®] Dream Taq PCR Master Mix (2X) and 1µM of each primer and 4µl of template. Amplicon was resolved on 2% ethidium bromide agarose gels and visualised under UV light.

All the PCR parameters that were employed in this study are as indicated in Appendix 2.8.

2.6.6 Quality control of PCR methods

All the PCR assays were tried on known positive samples before they were used on research samples. Positive and negative controls were included at the extraction, amplification and electrophoresis stages. A negative and a positive control were

included in every batch of 46 samples. Five percent (55 samples) of the samples were randomly selected and processed using the Seeplex[®] STI Master Panel 1 (V2.0) [Seegene Technologies Inc. California, USA]. The Seeplex[®] STI Master Panel 1 by Seegene is a multiplex conventional PCR system for the detection of seven organisms including *C. trachomatis*, *N. gonorrhoea* and *T. vaginalis* from urine, vaginal swabs and liquid based cytology specimens.

2.6.7 Quality assurance of data and sample collection

The questionnaire was translated into Bemba, the language that is commonly spoken in the study district. The questionnaire was pretested among 30 eligible participants at an ANC booking to determine pattern of response and clarity of the questions. Questions were rephrased, removed or added where necessary.

To ensure that good quality data was collected, research assistants were trained in data collection. Assistants were urged to ensure that questionnaires were checked for missing data before releasing the participant. The collection of cervico-vaginal samples was restricted to midwives after training on how samples were to be collected for diagnosis of BV and other STIs/RTIs. Other biological samples were collected by trained laboratory technicians who routinely carry out procedures such as venous and peripheral blood collection and slide preparation.

2.7 Ethical consideration

The study protocol was reviewed and approved by the University of Zambia Biomedical Research Ethics Committee, reference number 004-02-13 (Appendix 2.12) and the London School of Hygiene and Tropical Medicine Observational/Interventions Research Ethics Committee (reference number 6292) before the research was carried out. In addition, the Ministry of Health (MoH) in Zambia also reviewed the research proposal after the ethical approval had been granted. The research project was then discussed with relevant local authorities in the study district such as the district health management team, community leaders and the in-charge personnel at the two health centres.

At the beginning of each ANC booking session an explanation of the project was given to the group of first ANC attendees and their spouses/guardians/partners in

addition to the routine weekly health talk. The information included the aims of the study, the methods of data and sample collection, number of participants required, potential advantages and disadvantages of participation and the expected benefits of carrying out the research. The study population was also assured of confidentiality. After the details of the project had been explained potential participants were informed that participation was voluntary and participants could withdraw at any time. There was no age cut off for exclusion because inclusion of younger pregnant women was considered to be important in order to have better representation of pregnant women. Written consent was sort from guardians of women aged below 18 years if present after their assent. To ensure confidentiality interviews were conducted privately and each participant was assigned a unique identity number that was used to anonymise biological samples, questionnaires, and result record forms.

Participating women received everything that the ANC package routinely offered at the time of the study including iron and folic acid tablets. Women who were found RPR reactive were given a note to return to the health centre with their sexual partner for treatment. Pregnant women who had complaints of symptoms associated with STI/RTI were assessed by the regular clinician and managed based on their symptoms as part of the recommended ANC package. Women and/or their partners who tested HIV positive for the first time underwent assessment for eligibility to receive antiretroviral therapy and cotrimoxazole prophylaxis as per routine procedures. All women, except those who were receiving cotrimoxazole prophylaxis at the time, were provided IPTp-SP according to national guidelines. In the rare event that a woman showed up for ANC booking with symptomatic malaria, routine laboratory testing was done using a rapid diagnostic test kit (SD Bioline Malaria Ag P.f/Pan brand; Standard Diagnostics, Inc.) and treatment was administered based on national policy guidelines. Molecular diagnosis of Chlamydia, trichomoniasis and gonorrhoea was done in the latter part of the project and results were only available after the participants had delivered. Results were sent back to the two health centres and participants were who were traced were given a note to return to the health centres with their partners for treatment.

2.8 Data Processing and Analysis

Data were double-entered in EpiData version 3.1 software [206], cleaned, processed and analysed using Stata Software version 13 [207]. Cleaning involved visual checks of entered data for consistence and valid values. Frequencies of all variables were generated to check for missing data. Variables were recoded and composite variables were generated. Data on sources of income, level of education and fixed and durable assets were used to create an index of household wealth using principal components analysis [208]. The malaria and STI/RTI coinfection variable was generated from a combination of six outcomes, five curable STIs/RTIs and malaria. Coinfection was defined as infection with malaria and at least one of the five STIs/RTIs. High blood pressure (BP) was defined as systolic blood pressure \geq 140mm Hg and/or diastolic BP \geq 90 mm Hg. Hypertension was defined as two high BP readings [209]. A variable was created from BP readings collected at recruitment and before delivery but during labour. Adverse birth outcomes that were captured in this study were LBW, IUGR, stillbirth and PD. A new variable for composite ABO was generated from the four outcome variables and was defined as experiencing at least one of the four birth outcomes mentioned above. New variables were generated for the DHFR triple (Ile-51 + Arg-59 + Asn-108), DHPS double mutant (Gly-437 + Glu-540), quintuple (DHFR triple + DHPS double) and sextuple (quintuple + Gly-581) mutants.

Therapeutic efficacy was defined as aparasitaemia in a previously parasitaemic woman 28 days post-IPTp-SP without recrudescence. Women who were parasitaemic upon enrolment and were screened for malaria at day 28 were included in this analysis. Women who tested positive at day 0 and at day 28 were further separated into cases of recrudescence and reinfection. Therapeutic failure was defined as testing positive at day 0 and at day 28 due to recrudescence.

Prophylactic failure was defined as parasitaemia in a previously aparasitaemic woman 28 days after IPTp-SP administration. Women found aparasitaemic at recruitment and were tested for malaria at day 28 and those who tested positive at day 0 and were re-infected by day 28 were included in this analysis.

Quantitative variables (age and parasite count) were summarised using means and median. Characteristics of study participants were described using percentages and the Chi-squared test was used to assess differences in proportions between women

recruited at the two sites and between women followed up to delivery and those who were lost to follow-up. Standard statistical tests (i.e. Mann-Whitney *U* test, kruskal-wallis test, t-tests and analysis of variance were used to assess significance (*P* < 0.05) of differences in the distribution of continuous variables and comparison of means. Non-normally distributed variables were log-transformed before conducting t-tests or analyses of variance. The prevalence of malaria infection, determined by microscopy or by PCR, and their 95% CIs were estimated. The prevalence of the main outcomes (malaria, curable STIs/RTIs, malaria and curable STI/RTI coinfection, and ABO) and their 95% CIs were estimated. The prevalence of these outcomes was also estimated according to HIV status. The prevalence of malaria and coinfection was also estimated according to parity.

For malaria, the pre-specified outcome was malaria prevalence as determined by PCR. Results obtained from PCR diagnosis were used to assess potential risk factors for malaria infection. Potential predictors identified from literature were used to determine risk factors for infection in this study. Univariate analyses of potential predictors of: (i) Malaria infection; (ii) Infection with at least one STI; (iii) infection with BV, and (iv) coinfection with malaria and at least one STI/RTI, were conducted using logistic regression. Crude odds ratios were estimated. Potential predictors that showed significance at 10% were entered into a multivariable model and assessed using a likelihood ratio test. Factors that were found to be independently associated with infection at P < 0.05 were entered in a final model and adjusted odds ratios were obtained. Independent variables in final models were checked for interaction using a likelihood ratio test. The criterion for interaction was P < 0.01.

Mean birth weight, mean Hb and median gestational age were compared between the uninfected class and each of the other three pathology classes. Proportions of different ABO were estimated among the different placental pathology Chi-squared test was used to check for association between pathology class and ABO.

The prevalence of individual ABO and composite ABO among delivering participants and their 95%CIs were estimated. Risk factors for composite ABO were explored using univariate and multivariate logistic regression as described above. The variables for recruitment site, malaria, STIs/RTIs and their coinfection were included in the final model to assess whether their individual effect on birth outcome was

modified by the site of recruitment. Risk factors for individual ABO (PD, LBW and IUGR) were also explored using univariate and multivariate logistic regression as described above.

The prevalence estimates of the DHFR triple, DHPS double; DHFR + DHPS quintuple and sextuple mutants and their 95% CIs were calculated. Proportions of women in whom parasitaemia at day 0 was cleared by day 28 and those who experienced therapeutic and prophylactic failure were quantified. 3 Results (1): Maternal characteristics and prevalence estimates of malaria, STIs/RTIs and their coinfection

3.1 Chapter introduction

This chapter presents the participation distribution and socio-demographic characteristics of study participants. Characteristics are presented according to recruitment sites for assessment of any differences in proportions between the two sites. The chapter also describes prevalence of malaria, STIs/RTIs and their coinfection and HIV infection. The prevalence of malaria only and malaria and STI/RTI coinfection are presented according HIV infection as well as gravidity. STIs/RTIs are presented according to HIV status. Results are presented in this manner for the assessment of whether these factors modulate the prevalence malaria, STIs/RTIs and their coinfection in this study population.

3.2 Participation distribution

A total of 1086 women were recruited and 67.1% (n = 729) were successfully followed up to delivery. Details of the distribution of participants at recruitment and follow-up to delivery are presented in Figure 3.1.

Figure 3. 1: Participation flow-chart



Note: The loss to follow-up on the actual sample size (984) was 24%.

3.3 Socio-demographic characteristics of participants

Table 3-1 presents characteristics of study participants and their respective frequencies. The youngest participant was 14 and the oldest was 47 years old. The median age of participants was 25 years (Interquartile range [IQR], 20-30) and the median gestational age was 22 weeks (IQR, 19-25.5). Almost a quarter (24.1%) of women who were enrolled were primigravidae, 15.2% were secundigravidae and more than half (60.7%) were multigravidae.

The majority of these women, 80.6% (n = 874) were married and 40.7% (n = 426) reported six or fewer years of schooling. Nearly half of the 49.2% (n = 534) of the women reported ownership of a bed net (treated or untreated). Of 531 bed net owners, 78.3% (n = 416) reported to have slept under a bed net the previous night. Overall only 38.5% (416/1082) of participants slept under a bed net on the previous night. Indoor residual spraying had been applied to the homes of 20.9% (n = 226) of women at least once in the previous 12 months.

There were no significant differences in the socio-demographic characteristics (age, marital status, years of schooling, bed net use and wealth index) between women recruited at Kashikishi and at Nchelenge health centre. The pregnancy related factors (gravidity, past experience of miscarriage, past history of stillbirth or PD) and clinical symptoms and treatments (symptoms of STI/RTI, HIV status, and use of antimalarial drugs prior to ANC visit) also did not differ between the two sites. However, Nchelenge site had a higher prevalence of peripheral malaria parasitaemia than Kashikishi (PCR positive malaria 67.4% versus 53.5%, P < 0.001; microscopy positive malaria 37.6% versus 29.1%, P = 0.006).

All (N = 1085) Kashikishi (n = 747) P-value variable Nchelenge (n = 338) n (%) or median n (%) or median n (%) or median (IQR) (IQR) (IQR) 0.133 Age (median) 25 (20-30) 24 (20-30) 25 (20-31) Marital status Single 203 (18.7) 134 (17.9) 69 (20.4) 0.591 Married 874 (80.6) 607 (81.3) 267 (80.0) Widowed/divorced 8 (0.7) 6 (0.8) 2 (0.6) Years of schooling 426 (39.3) 287 (38.4) 139 (41.1) 0.398 None to 6 7 and above 460 (61.6) 199 (58.9) 659 (60.7) Gravidity Primigravidae 0.657 261 (24.1) 183 (24.5) 78 (23.1) Secundigravidae 165 (15.2) 117 (15.6) 48 (14.2) Multigravidae 659 (60.7) 447 (59.8) 212 (62.7) Bed net ownership 0.105 No 551 (50.8) 367 (49.1) 184 (54.4) Yes 534 (49.2) 380 (50.9) 154 (45.6) Bed net usage No 666 (61.6) 451 (60.6) 215 (63.6) 0.349 Yes 416 (38.4) 293 (39.4) 123 (36.4) 3 3 Missing* 0 IRS in previous 12 months 808 (78.1) 563 (75.7) 245 (72.7) 0.851 No 226 (21.9) 156 (21.0) 70 (20.8) 51 Yes 28 26 Missing* Wealth guintile Lowest 217 (20.0) 135 (18.1) 82 (24.3) 0.089 Second 221 (20.4) 151 (20.3) 70 (20.7) Middle 214 (19.8) 155 (20.8) 59 (17.5) 215 (19.9) Fourth 158 (21.20 57 (16.9) Highest 216 (19.9) 146 (19.6) 70 (20.7) Missing* 2 2 0 Past experience of miscarriage No 708 (85.1) 482 (85.3) 226 (87.3) 0.455 Yes 116 (14.1) 83 (14.7) 33 (12.7) NA (Primigravidae)* 261 182 79 Past experience of PD 0.797 No 784 (95.3) 538 (95.4) 246 (95.0) 39 (4.7) 26 (4.6) 13 (5.0) Yes NA (Primigravidae)* 261 182 79 Missing* 1 1 0

 Table 3- 1: Baseline characteristics of first ANC attendees of Nchelenge District, Zambia,

 according to recruitment site

Variable All (N = 1085) Kashikishi (n = 747) Nchelenge (n =338) P-value n (%) n (%) n (%) Prior delivery of a still 755 (91.6) 521 (92.2) 234 (90.4) 0.370 born No 69 (8.4) 44 (7.8) 25 (9.7) 79 Yes 261 182 NA (Primigravidae)* Prior delivery of a still born 755 (91.6) 521 (92.2) 234 (90.4) No 69 (8.4) 44 (7.8) 25 (9.7) 0.370 79 Yes 261 182 NA (Primigravidae)* Prior delivery of a Preterm baby 784 (95.3) 538 (95.4) 0.797 No 246 (95.0) 26 (4.6) Yes 39 (4.7) 13 (5.0) NA (Primigravidae)* 261 182 79 Missing* 1 1 0 Antimalarial treatment [‡] 0.995 No 693 (93.2) 312 (93.1) 1005 (93.1) 74 (6.9) 23 (6.9) Yes 51 (6.9) Missing* 6 3 3 STI/RTI symptoms 0.345 complaints among all No 951 (87.7) 650 (87.0) 301 (89.1) Yes 134 (12.4) 97 (13.0) 37 (10.9) STI/RTI symptoms complaints among infected 602 (88.0) 415 (85.4) 187 (87.4) 0.487 No 98 (14.0) 57 (14.6) 27 (12.6) Yes 260 384 124 NA (Negative)* HIV status 0.374 Negative 941 (86.8) 643 (86.2) 298 (88.2) Positive 143 (13.2) 103 (13.8) 40 (11.8) 0 Missing* 1 1 **ART among HIV Positives** No 82 (57.7) 58 (56.9) 24 (60.0) 0.734 60 (42.7) 44 (43.1) 16 (40.0) Yes 941 643 298 NA (HIV-negative)*

 Table 3-1: Baseline characteristics of first ANC attendees of Nchelenge District, Zambia,

 according to recruitment site continued

Variable	All (N = 1085)	Kashikishi (n = 747)	Nchelenge (n =338)	P-value
	n (%)	n (%)	n (%)	
Microscopy				
Negative	736 (68.2)	525 (70.9)	211 (62.4)	0.006
Positive	343 (31.8)	216 (29.1)	127 (37.6)	
Missing*	6	6	0	
STI/RTI				
Negative	384 (35.4)	260 (34.9)	124 (36.7)	0.599
Positive	700 (64.6)	486 (65.1)	214 (63.3)	
Missing	1	1	0	
Malaria & STI/RTI				
Coinfection ⁺				
No infection	177 (16.5)	128 (17.4)	49 (14.6)	0.071
Single infection [¥]	480 (44.8)	339 (46.2)	141 (41.8)	
Coinfection	414 (38.7)	267 (36.4)	147 (43.6)	
Missing*	14	13	1	

Table 3-1: Baseline characteristics of first ANC attendees of Nchelenge District, Zambia, according to recruitment site continued

NA: Not applicable, IQR: interquartile range, IRS: Indoor residual spraying

* Missing and NA values are only presented as numbers and were not included in the calculation of

percentages and in the Chi-squared test for association

[†]Infection with at least one STI/RTI and malaria

[¥]Single infection: Infection with at least one STI/RTI or malaria

^{*}Antimalarial treatment during pregnancy taken more than four weeks before recruitment

Less than 5 women took antibacterial agents during pregnancy but more than a month before

recruitment

3.4 Malaria and HIV Burden

3.4.1 The Prevalence of HIV infection

The prevalence of HIV infection among participants was 13.2% (95% CI, 11.3–15.3) and the highest burden was observed in multigravidae (15.9%) followed by secundigravidae (13.9%) and was lowest in primigravidae (8.0%), P = 0.018. A higher proportion of infection was noted among women aged above 25 years than women aged \leq 25 years (17.9% versus 9.4%, P < 0.001).

3.4.2 The prevalence of peripheral malaria

The prevalence of malaria infection at first ANC attendance measured by microscopy and by PCR was 31.8% (95% CI, 29.1-34.6) and 57.8% (95% CI, 54.9-60.8) respectively. Parasite density ranged from 64–24,760 parasites per μ l of blood with a geometric mean of 1,082 (95% CI, 962-1,217) asexual parasites per μ l of blood. Parasite density was highest among primigravidae with the geometric mean being significantly different across gravidae (*P* < 0.001). Of the 343 malaria positive samples detected by microscopy, seven were negative for *P. falciparum* by PCR (Table 3-2).

Of the 1,074 women whose results were available for malaria detected by PCR, 8.4% (n = 90, 95% CI, 6.7-10.2) were coinfected with HIV. The prevalence of malaria among HIV-infected and HIV-uninfected women detected by PCR was 65.2% (90/138) and 56.7% (531/936), respectively (P = 0.059). Among women who tested positive for HIV, combined with those with known HIV positive status, 42.7% were receiving antiretroviral therapy at the time of recruitment. The proportions of malaria infection detected by PCR among HIV-infected women on ARV treatment and those who were not on ARV treatment were 71.8% (56/78) and 56.7% (34/60) respectively, P = 0.064.

In the case of microscopy, 31.3% (294/939) of HIV-uninfected had parasitaemia compared to 35.0% (49/140) among HIV-infected women (P = 0.382). HIV-uninfected women had significantly lower parasite densities than HIV-infected women, 990 (95% CI, 876-1120) versus 1836 (95% CI, 1306-2580), P < 0.001. A lower proportion of malaria infection diagnosed by microscopy was observed among HIV-infected

women on ARV therapy than among those who were not on therapy, 24.6% (15/61) versus 43.0% (34/79) respectively, P = 0.023. Parasite density among HIV-infected women receiving ARV therapy was lower than that observed among those who were not on therapy at recruitment, 1,310 (95% CI, 693-2,476, n = 15) versus 2,131 (95% CI, 1,408-3,224, n = 34), respectively but this was not statistically significant (P = 0.189).

3.4.3 Placental malaria infection

Of the 729 women who were followed to delivery, placental tissue was successfully processed for 96% (n = 702) participants and 40.9% (n=287) of these had histopathological evidence of placental malaria (Table 3-3). The highest proportion (30.4%) was past infections detected by the presence of the pigment without the presence of parasites. The frequency of placental infection, past or active or both, was higher in primigravidae than multigravidae and this difference was statistically significant (52.7% versus 33.6%; P < 0.001). Further results on placental malaria infection 6.2).

Table 3- 2: Prevalence of HIV and *Plasmodium falciparum* malaria infection at first antenatal attendance and at delivery among women of Nchelenge district according to gravidity/parity

Infection	All gravidae	95%	Primigravidae	Secundigravidae	Multigravidae	P-value
Diagnostic method	N = 1085	Confidence	n=261	n=165	n=659	
	n(%) or n, geometric	Interval	n(%) or n, geometric	n(%) or n, geometric	n (%) or n, geometric	
	incan					
Negative	0.41 (86.8)	00 0 0 4 0	240 (02 0)	14296 1)	550 (84.0)	0.018
Negative	941 (86.8)	80.8-84.8	240 (92.0)	14286.1)	559 (84.9)	0.018
Positive	143 (13.2)	11.3-15.3	21 (8.0)	23 (13.9)	99 (15.1)	
Missing*	1		0	0	1	
Malaria (PCR)						
Negative	453 (42.2)	39.2-45.1	81 (31.3)	63 (38.2)	309 (47.5)	< 0.001
Positive	621 (57.8)	54.8-60.7	178 (68.7)	102 (61.8)	341 (52.5)	
Missing*	11		2		9	
Microscopy						
Negative	736 (68.2)	65.4-71.0	127 (49.3)	104 (63.0)	505 (77.1)	< 0.001
Positive	343 (31.8)	29.1-34.6	132 (51.0)	61 (37.0)	150 (22.9)	
Missing*	6		2	1	3	
Geometric mean of	343, 1082	962-1217	132, 1378	61, 1164	150, 848	
Parasite density			(1133-1677)	(897-1509)	(714-1010)	< 0.001
Placental malaria						
Negative	415 (59.1)	55.5-62.8	89 (47.3)	46 (50.0)	280 (66.4)	< 0.001
Positive	287 (40.9)	37.3-44.6	199 (52.7)	46 (50.0)	142 (33.6)	
Missing*	27		6	6	15	
Lost to follow-up*	356		67	67	222	

Of the 343 malaria positive samples detected by microscopy 7 were found negative by PCR specific for *P. falciparum* detection

The overall prevalence (40.9%) is a combination of different classes of pathology as follows: Presence of parasites only - 2.4%; Presence of parasites and pigment - 8.1%; Presence of pigment only – 30.4%.

*Missing and lost to follow-up. *P*-values are only presented as numbers and were not included in the calculation of percentages and in the Chi-squared test for association

Of the expected samples from 1085 participants, 4 dry blood spots and 4 slides were not collected. In addition to this 5 dry blood spots were misplaced

3.4.4 Monthly peripheral malaria infection among first ANC attendees

Figure 3.2 shows the prevalence of malaria infection among study participants each calendar month during the six-month recruitment period from November 2013 to April 2014. The malaria prevalence was slightly higher between February and April compared to the period between November and January.

Figure 3. 2: Prevalence of malaria in pregnant women of Nchelenge District and total monthly rainfall during the study recruitment period.



PCR: Polymerase chain reaction

Mm: Millimetres

3.5 Reproductive tract and curable sexually transmitted infections

The prevalence of STIs/RTIs among pregnant women in this study was high. Overall, infection with at least one STI/RTI was observed in 64.8% (95% CI, 61-67.4) of the participating women. The most prevalent infection was BV (48.3%) and the least prevalent was gonorrhoea (3.1%). With the exclusion of BV the prevalence of infection with at least one curable STI was 34.5% (95% CI, 31.7-37.4).

Contrary to what was observed in the case of HIV infection, a higher prevalence of STIs was observed among women aged ≤ 25 years than women aged above 25 years (38.7% versus 29.2%, *P* = 0.001). The prevalence of BV was higher among women aged ≤ 25 years than among older women, but this was not statistically significant (50.1 versus 45.8%, *P* = 0.130).

Table 3-4 presents the prevalence of individual and composite STIs/RTIs among participants according to HIV status. The prevalence of curable STIs/RTIs was higher among HIV-infected women than among HIV-uninfected women. The prevalence of BV, Chlamydia, gonorrhoea and syphilis was higher among HIV-infected women than among HIV-uninfected women. This was statistically significant in the cases of gonorrhoea, BV and syphilis but not in the case of Chlamydia. There was no statistically significant difference in the prevalence of trichomoniasis between HIV-uninfected women and HIV-infected women, 25.0% verse 23.8% (P = 0.752). With the exclusion of BV, prevalence of any STI was higher among HIV-infected than HIV uninfected women but this was not statistically significant (41.3 versus 33.5%, P = 0.069).

Of the 700 women who were infected with at least one STI/RTI at first ANC visit 14.0% (n = 98) complained of at least one symptom associated with curable STIs/RTIs (itching, discharge, burning session during urination or presence of sores/ulcers). About 9.4% (36/384) had similar complaints among uninfected women. A similar proportion was observed when only curable STIs were considered. Of the 374 women who tested positive to at least one STI, 14.2% (n = 53) indicated that they were experiencing at least one symptom indicative of infection with an STI at the time of recruitment.

Infection	All	95%	HIV-uninfected	HIV-infected	P-value
	(N = 1084)	Confidence	(n = 941)	(n = 143)	
	n (%)	interval [*]	n (%)	n (%)	
Bacterial vaginosis					
Normal	375 (34.7)	31.8-37.5	342 (36.5)	33 (23.1)	< 0.001
Intermediate	184 (17.0)	14.9-19.4	169 (18.0)	15 (10.5)	
BV	524 (48.3)	45.2-51.2	429 (45.8)	95 (66.4)	
Missing*	4		4	0	
Trichomoniasis					
Negative	814 (75.2)	72.6-77.8	705 (75.0)	109 (76.2)	0.752
Positive	269 (24.8)	22.3-27.5	235 (25.0)	34 (23.8)	
Missing*	1		1	0	
Chlamydia					
Negative	1027 (94.8)	93.5-96.2	894 (95.1)	133 (93.0)	0.291
Positive	56 (5.2)	3.9-6.7	46 (4.9)	10 (7.0)	
Missing*	1		1	0	
Gonorrhoea					
Negative	1049 (96.9)	95.8-97.9	915 (97.3)	134 (93.7)	0.020
Positive	34 (3.1)	2.2-4.4	25 (2.7)	9 (6.3)	
Missing*	1		1	0	
Confirmed syphilis					
Negative	1001 (92.9)	91.4-94.5	884 (94.2)	117 (84.2)	< 0.001
Positive ^a	76 (7.1)	5.6-8.7	54 (5.8)	22 (15.8)	
Missing*	7		3	4	
Composite STI					
Negative	710 (65.5)	62.7-68.3	625(66.5)	84 (58.7)	0.069
Positive	374 (34.5)	31.7-37.4	315 (33.5)	59 (41.3)	
Missing*	1		1	0	
Composite STI/RTI					
Negative	381 (35.2)	32.6-38.3	348 (37.0)	33 (23.1)	0.001
Positive	702 (64.8)	61.7-67.4	592 (63.0)	110 (76.9)	
Missing*	1		1	0	

 Table 3- 3: Prevalence of curable STIs/RTIs among first ANC attendees of Nchelenge district according to HIV status

HIV: Human Immunodeficiency Virus

Composite STI/RTI: sexually transmitted and reproductive tract infections (included bacterial vaginosis, Chlamydia, gonorrhoea, trichomoniasis and TPHA confirmed syphilis)

Composite STI: included Chlamydia, gonorrhoea, trichomoniasis and TPHA confirmed syphilis * Missing values are only presented as numbers and were not included in the calculation of percentages and in the Chi-squared test for association

^aThe figure represents the number of samples that were TPHA reactive among those who were seropositive by RPR calculated out of all the participants who were tested with RPR (1077). Note: Among the participants who were RPR reactive, 65% (76/117) were found reactive with TPHA

Overall the total number was 1084 instead of 1085 because the HIV status of one participant was not determined.

3.6 Malaria and curable STI/RTI coinfection

Out of the 1071 women tested for malaria and all the five curable STI/RTI, 38.7% (n= 414) were coinfected with malaria and STI/RTI (Table 3-5). The prevalence of coinfection was lower in the case of malaria diagnosed by microscopy (21.4%, 230/1076).

3.6.1 Malaria and STI/RTI coinfection according to parity

Mono-infection with malaria diagnosed by PCR did not differ significantly among the three gravidity groups (primigravidae, secundigravidae and multigravidae) but infection with at least one curable STI/RTI increased slightly with increasing gravidity (19.4%, 25.5% and 29.0%, respectively, P = 0.012). However, the prevalence of coinfection decreased with increasing gravidity (49.6%, 44.2% and 32.9% respectively, P < 0.001. This trend in the frequency of coinfection across gravidity groups persisted even when the malaria diagnosis was done by microscopy (33.7%, 28.5% and 14.7% respectively, P < 0.001).

The median ages of women in the three groups of infection were as follows; coinfection 22.5 years (IQR, 19-28), mono-infection (at least one STI/RTI or malaria) 25 years (IQR, 21-30) and uninfected 27 years (IQR, 21-31) (P < 0.001, kruskal-wallis equality-of-population rank test).

Category of	All (N =1085)	All	Primigravidae	Secundigravidae	Multigravidae	P-value
infection	n (%)	95% CI	n = 261	n = 165	(n = 648)	
			n (%)	n (%)	n (%)	
Coinfection [¥]	414 (38.7)	35.7 - 41.6	128 (49.6)	73 (44.2)	213 (32.9)	< 0.001
Malaria only	202 (18.8)	16.5 - 21.2	47 (18.2)	29 (17.5)	126 (19.4)	0.822
STI/RTI only	278 (26.0)	23.5 - 28.8	50 (19.4)	42 (25.5)	188 (29.0)	0.012
No infection	177 (16.5)	14.1 - 18.6	33 (12.8)	21 (12.7)	121 (18.7)	0.038
Missing*	14		3	0	11	
V						
Coinfection [*] bas	ed on malaria diagr	nosed by microsc	сору			
Coinfection [*] bas Category of	ed on malaria diagr All (N =1085)	All	opy Primigravidae	Secundigravidae	Multigravidae	<i>P</i> -value
Coinfection [*] bas Category of infection	ed on malaria diagr All (N =1085) n (%)	All 95% Cl	opy Primigravidae n = 261	Secundigravidae n = 165	Multigravidae (n = 648)	P-value
Coinfection [*] bas Category of infection	ed on malaria diagr All (N =1085) n (%)	All 95% Cl	opy Primigravidae n = 261 n (%)	Secundigravidae n = 165 n (%)	Multigravidae (n = 648) n (%)	<i>P</i> -value
Coinfection [*] bas Category of infection Coinfection [*]	ed on malaria diagr All (N =1085) n (%) 230 (21.4)	All 95% Cl 18.9 - 23.8	Primigravidae n = 261 n (%) 87 (33.7)	Secundigravidae n = 165 n (%) 47 (28.5)	Multigravidae (n = 648) n (%) 96 (14.7)	<i>P</i> -value < 0.001
Coinfection [*] bas Category of infection Coinfection [¥] Malaria only	ed on malaria diagr All (N =1085) n (%) 230 (21.4) 93 (8.6)	All 95% Cl 18.9 - 23.8 6.9 -10.3	Primigravidae n = 261 n (%) 87 (33.7) 39 (15.1)	Secundigravidae n = 165 n (%) 47 (28.5) 13 (7.7)	Multigravidae (n = 648) n (%) 96 (14.7) 41 (6.3)	P-value < 0.001
Coinfection [*] bas Category of infection Coinfection [¥] Malaria only STI/RTI only	ed on malaria diagr All (N =1085) n (%) 230 (21.4) 93 (8.6) 469 (43.5)	All 95% Cl 18.9 - 23.8 6.9 -10.3 40.6 - 46.5	Primigravidae n = 261 n (%) 87 (33.7) 39 (15.1) 91 (35.3)	Secundigravidae n = 165 n (%) 47 (28.5) 13 (7.7) 69 (41.8)	Multigravidae (n = 648) n (%) 96 (14.7) 41 (6.3) 309 (47.3)	P-value < 0.001
Coinfection [*] bas Category of infection Coinfection [*] Malaria only STI/RTI only No infection	ed on malaria diagr All (N =1085) n (%) 230 (21.4) 93 (8.6) 469 (43.5) 285 (26.4)	All 95% Cl 18.9 - 23.8 6.9 -10.3 40.6 - 46.5 23.8 - 29.1	Primigravidae n = 261 n (%) 87 (33.7) 39 (15.1) 91 (35.3) 41 (15.9)	Secundigravidae n = 165 n (%) 47 (28.5) 13 (7.7) 69 (41.8) 36 (21.8)	Multigravidae (n = 648) n (%) 96 (14.7) 41 (6.3) 309 (47.3) 208 (31.8)	P-value < 0.001

Table 3- 4: Prevalence of malaria and STI/RTI coinfection according to gravidity

*Missing values are only presented as numbers and were not included in the calculation of percentages and in the Chi-squared test for association Missing values were due missing samples, 14 samples could not be classified due to some missing results.

3.5.2 Malaria and STI/RTI coinfection according to HIV status

Table 3-5 presents the prevalence of individual STI/RTI and malaria coinfection as well as composite STI/RTI and malaria coinfection according to HIV status. The prevalence of coinfection with malaria and individual STIs/RTIs was higher among HIV-infected than HIV-uninfected women but this was only statistically significant in the case of syphilis and BV regardless of the diagnostic tool used for malaria detection. In the case of malaria diagnosed by PCR, coinfection with syphilis among HIV-infected and HIV-uninfected women was 8.8% versus 3.3%, respectively (P =0.002). Coinfection with BV and malaria was 45.7% versus 26.8% among HIV-infected and HIV-uninfected women respectively (P < 0.001). Coinfection with malaria and BV was the most prevalent (29.2%; 95% CI 26.5-32.0) and coinfection with malaria and gonorrhoea was the least prevalent (2.1%; 95% CI 1.3-3.1). STI/RTI among the 621 women who tested positive to malaria by PCR was as follows: 50.9% were infected with BV; 26.4% were infected with trichomoniasis; 6.0% were infected with Chlamydia; 3.5% were infected with gonorrhoea and 7.0% had syphilis. Composite STI/RTI and malaria coinfection was higher among HIV-infected than HIV-uninfected women, 50.0% versus 37%, P = 0.003 in case of Malaria diagnosed by PCR and 27.1% versus 20.5%, *P* = 0.073 in the case of malaria diagnosed by microscopy.

Table 3- 5: Prevalence of individual curable STI/RTI and malaria coinfection among first antenatal care attendees of Nchelenge District, Zambia

Malaria diagnosed by PCR						
Coinfection	N = 1084	95% CI	HIV-uninfected	HIV-infected	P-Value	
	n (%)		n = 941 n (%)	n = 143 n (%)		
Malaria and Bacterial vaginosis						
No	757 (70.8)	68.0-73.5	682 (73.2)	75 (54.4)	< 0.001	
Yes	313 (29.2)	26.5-32.0	250 (26.8)	63 (45.7)		
Missing*	14		9	5		
Malaria and Trichomonas vaginalis						
No	909 (84.7)	82.6-86.9	793 (84.8)	116 (84.1)	0.818	
Yes	164 (15.3)	13.3-17.6	142 (15.2)	22 (15.9)		
Missing*	11		6	5		
Malaria and Chlamydia trachomatis						
No	1036 (96.6)	95.5-97.6	905 (96.8)	131 (94.9)	0.263	
Yes	37 (3.4)	2.4-4.5	30 (3.2)	7 (5.1)		
Missing*	11		6	5		
Malaria and Neisseria gonorrhoeae						
No	1051 (97.9)	97.1-98.8	918 (98.2)	133 (96.4)	0.162	
Yes	22 (2.1)	1.2-2.9	17 (1.8)	5 (3.6)		
Missing*	11		6	5		
Malaria and Treponema Pallidum						
No	1029 (96.0)	94.8-97.2	904 (96.7)	125 (91.2)	0.002	
Yes	43 (4.0)	2.8-5.2	31 (3.3)	12 (8.8)		
Missing*	12		6	5		
Malaria and STI/RTI coinfection ^{\$}						
Coinfection	414 (38.7)	35.8-41.6	345 (37.0)	69 (50.0)	0.003	
Malaria only	202 (18.9)	16.6-21.3	182 (19.5)	20 (14.5)	0.154	
STI/RTI only	278 (26.0)	23.4-28.7	237 (25.5)	41 (29.7)	0.286	
No infection	177 (16.5)	14.3-18.8	169 (18.1)	8 (5.8)	< 0.001	
Missing*	13		8	5		

Malaria diagnosed by microscopy						
Coinfection	N = 1084	95% CI	HIV-uninfected	HIV-infected	P-Value	
	n (%)		n = 941 n (%)	n = 143 n (%)		
Malaria and Bacterial vaginosis						
No	896 (83.1)	80.8-85.3	789 (84.4)	104 (74.3)	0.003	
Yes	182 (16.9)	14.7-19.2	146 (15.6)	36 (25.7)		
Missing*	9		6	3		
Malaria and Trichomonas vaginalis						
No	983 (91.2)	89.5-92.9	851 (90.7)	132 (94.3)	0.166	
Yes	95 (8.8)	7.1-10.5	87 (9.3)	8 (5.7)		
Missing*	6		3	3		
Malaria and Chlamydia trachomatis						
No	1055 (97.9)	97.0-98.7	919 (98.0)	136 (97.1)	0.527	
Yes	23 (2.1)	1.2-3.0	19 (2.0)	4 (2.9)		
Missing*	6		3	3		
Malaria and Neisseria gonorrhoeae						
No	1069 (99.2)	98.6-97.7	931 (99.2)	138 (98.6)	0.408	
Yes	9 (0.83)	034-1.4	7 (0.8)	2 (1.42)		
Missing*	6		3	3		
Malaria and Treponema Pallidum						
No	1051 (97.8)	96.9-98.7	922 (98.4)	129 (93.5)	< 0.001	
Yes	24 (2.2)	1.3-3.1	15 (1.6)	9 (6.5)		
Missing*	9		4	5		
Malaria and STI/RTI coinfection ^{\$}						
Coinfection	230 (21.4)	18.9-23.8	192 (20.5)	38 (27.1)	0.073	
Malaria only	93 (8.6)	7.0-10.3	84 (9.0)	9 (6.5)	0.322	
STI/RTI only	467 (43.4)	40.4-46.4	395 (42.2)	72 (51.4)	0.039	
No infection	286 (26.6)	23.9-29.3	265 (28.3)	21 (15.0)	< 0.001	
Missing*	9					
*Missing values are only presented as numbers and were not included in the calculation of percentages						
\$ Coinfection: infection with malaria and at least one STI/RTI						

Table 3-5: Prevalence of individual curable STI/RTI and malaria coinfection among first antenatal care attendees of Nchelenge District, Zambia continued

3.7 Quality control assessment results

Results for quality control testing were highly concordant. There was a 96.4% (53/55) concordance in the Nugent score results obtained from the repeat reading of slides for detection of bacterial vaginosis. The two readings both conflicted between the normal and the intermediate class. Syphilis RPR results were 100% (55/55) concordant; TPHA testing was 100% (30/30) concordant. For *C. trachomatis,* agreement of test results between the in-house methods and the Seeplex[®] STI Master Panel 1 kit was 96.4% (54/55). In the case *of N. gonorrhoeae* and *T. vaginalis,* there was 100% concordance between the in house methods and the commercial kit. Repeat reading of slides for placental malarial diagnosis by histology was 95.7% (44/46) concordant.

4 Results (2): Risk factors for malaria, STI/RTI and their coinfection

4.1 Chapter introduction

This chapter presents risk factors for (i) malaria (ii) STI, (iii) BV and (iv) malaria and STI/RTI coinfection as identified by logistic regression models. The variables included in the analyses were guided by risk factors for infection as identified from literature in section 1.5.1. The tables presented include both univariate and multivariate analyses of risk factors for infection.

4.2 Risk factors for malaria infection

Table 4-1 presents risk factors for malaria infection. In the univariate analyses the association between malaria infection and the following factors was statistically significant (P < 0.1); recruitment site, age group, marital status, gravidity, bed net ownership, bed net usage, wealth quintile and HIV status. However, in the multivariate analysis, only three factors i.e. HIV status, site of recruitment and wealth quintile were independently associated with malaria infection (P < 0.05). Wealth quintile was not associated with malaria infection in a model with HIV infection and site of recruitment and was therefore excluded from the final model. In the final model, the risk of malaria infection was higher among women recruited at Nchelenge health centre (aOR = 1.81; 95% CI, 1.38-2.37, P < 0.001) and HIV-infected women across both health centres (aOR = 1.46; 95% CI 1.00-2.13, P = 0.045). There was no interaction between HIV status and site of recruitment (P = 0.751).
Potential risk factor	Number	%	Unadjusted OR	Adjusted OR	Adjusted OR for	<i>P</i> -value
					final model	
	1074	100	-	N = 1068	N = 1072	< 0.001 [¥]
Recruitment site			P < 0.001	P < 0.001		
Kashikishi	737	68.6	1.00	1.00	1.00	
Nchelenge	337	31.4	1.80 (1.37-2.35)	1.89 (1.43-2.50)	1.81 (1.38-2.37)	< 0.001
Age group			<i>P</i> < 0.001	<i>P</i> = 0.153	-	-
≤ 20	295	27.5	1.00	1.00		
21-25	304	28.3	0.51 (0.37-0.72)	0.63 (0.41-0.98)		
26-30	244	22.7	0.51 (0.36-0.73)	0.67 (0.41-1.09)		
≥ 30	231	21.5	0.45 (0.32-0.64)	0.58 (0.34 -0.96)		
Marital status			<i>P</i> = 0.114	-	-	-
Single	201	18.7	1.00			
Married	865	80.5	0.72 (0.53-0.99)			
Divorced/separated/	8	0.8	0.56 (0.14-2.30)			
widowed						
Gravidity			P < 0.001	P = 0.077	-	-
Primigravidae	259	24.1	1.00	1.00		
secundigravidae	165	15.4	0.74 (0.48-1.11)	0.80 (0.48-1.33)		
Multigravidae	650	60.5	0.50 (0.37-0.68)	0.58 (0.35-0.97)		
Bed net ownership			<i>P</i> = 0.015	<i>P</i> = 0.589	-	-
No	547	50.9	1.00	1.00		
Yes	527	49.1	0.74 (0.58-0.94)	1.13 (0.73-1.75)		
Bed net use on previous night			P = 0.006	<i>P</i> = 0.188	-	-
No	659	61.5	1.00	1.00		
Yes	412	38.5	0.75 (0.48-0.91)	0.74 (0.48-1.16)		
Missing*	3					

Table 4- 1: Predictors of malaria infection in pregnant women of Nchelenge District

Potential risk factor	Number	%	Unadjusted OR	Adjusted OR for multivariate	Adjusted OR for final model	P-value
				Analysis		
	1074	100	-	N = 1068	N = 1072	< 0.001 [¥]
IRS in past 12 months			<i>P</i> = 0.386	-	-	-
No	799	78.0	1.00			
Yes	225	22.0	1.19 (0.88-1.61)			
Unknown*	46					
Wealth quintile			<i>P</i> = 0.072	<i>P</i> = 0.031	-	-
Lowest	216	20.1	1.00	1.00		
Second	219	20.4	0.80 (0.54-1.17)	0.74 (0.50-1.09)		
Middle	209	19.5	0.91 (0.61-1.34)	0.77 (0.51-1.16)		
Fourth	214	20.0	1.08 (0.73-1.60)	0.89 (0.58-1.36)		
Highest	214	20.0	0.64 (0.44-0.95)	0.53 (0.35-0.81)		
Missing *	2					
HIV status			<i>P</i> = 0.057	1.00	1.00	
Negative	941	86.8	1.00	1.56 (1.05-2.32)	1.46 (1.00-2.13)	0.045
Positive	143	13.2	1.30 (0.91-1.88)			

Table 4-1: Predictors of malaria infection in pregnant women of Nchelenge District continued

-: Blank

[¥]*P*- value for overall final model

P-values appearing with unadjusted odds ratios are for each univariate model indicating associations of the variable with the outcome at P < 0.1.

P-values appearing with adjusted odds ratios indicate the independent association of each variable with the outcome at *P* < 0.05

Bold type- statistically significant outcome for multivariate analysis and final model.

Number of participants less than 1074 (available malaria test results by PCR) were due to missing values in individual variables.

*Missing/unknown values are only presented as numbers and were not included in the calculation of percentages and in the models

When wealth quintile was put in the model with HIV and site of recruitment it was not significantly associated with malaria infection and was therefore excluded from the final model.

There was no interaction between HIV infection and site of recruitment in the final model (*P* =0.751)

4.3 Risk factors for curable STIs/RTIs

Risk factors for STIs and BV which is not sexually transmitted were analysed separately and are presented in Tables 4-2 and Table 4-3.

4.3.1 Risk factors for STIs

Table 4-2 shows univariate and multivariate analyses of risk factors for composite STIs (Chlamydia, gonorrhoea, trichomoniasis and TPHA confirmed syphilis). In the univariate analysis infection with at least one STI was associated with age group, marital status, gravidity, number of life-time sexual partners, wealth quintile, HIV status and BV at P < 0.1. In multivariate analysis infection with at least one STI was associated with wealth quintile and infection with BV. Wealth quintile was not independently associated with infection with at least one STI in a model with BV and was therefore excluded from the final model. In the final model, infection with at least one STI was associated with BV. In comparison to uninfected women, women infected with BV were at a higher risk of being infected with at least one STI (aOR 1.44; 95% CI, 1.08-1.92, P = 0.012).

4.3.2 Risk factors for BV

Table 4-3 shows univariate and multivariate analyses of risk factors for BV. In the univariate analysis bacterial vaginosis was associated with age group, marital status, gravidity, HIV and infection with at least one STI. In the multivariate analysis infection with HIV and infection with at least one STI were independently associated with BV. In the final model HIV-infected women had a higher risk of infection with BV than HIV-uninfected women (aOR 1.87; 95% CI, 1.24-2.83, *P* = 0.003) and women infected with at least one STI had a higher risk of BV (aOR 1.40; 95% CI (1.07 -1.84, *P* = 0.01).

Variable	Number	%	Unadjusted OR	Adjusted OR	Final model	P-value
Category	1084	100	-	N = 1067	N = 1078	0.007 [¥]
Age group			<i>P</i> < 0.001	<i>P</i> = 0.101	-	-
≤ 20	296	27.3	1.00	1.00		
21-25	309	28.5	0.79 (0.56-1.09)	0.89 (0.58 -1.35)		
26-30	245	22.6	0.72 (0.51-1.02)	0.82 (0.51-1.35)		
≥ 30	234	21.6	0.45 (0.31-0.66)	0.56 (0.33-0.95)		
Maternal education			<i>P</i> = 0.893	-	-	-
0-6 years	426	39.3	1.00			
≥ 7 years	658	60.7	0.98 (0.76-1.27)			
Marital status			<i>P</i> = 0.022	<i>P</i> = 0.768	-	-
Single	203	18.7	1.00	1.00		
Married	873	80.4	0.64 (0.47-0.87)	0.87 (0.58-1.29)		
Separated/widowed/divorced	8	0.7	0.80 (0.19-3.4)	0.76 (0.13-4.40)		
Gravidity			<i>P</i> < 0.001	<i>P</i> = 0.151	-	-
Primigravidae	261	24.1	1.00	1.00		
secundigravidae	165	15.2	1.03 (0.69-1.52)	1.12 (0.69-1.81)		
Multigravidae	658	60.7	0.59 (0.43-0.79)	0.76 (0.46-1.24)		
No. of life time sexual			<i>P</i> = 0.037		-	-
partners	797	74.2	1.35 (1.02-1.79)	<i>P</i> = 0.05		
1 or 2	277	25.8	0.49 (0.42-0.56)	1.00		
3 or more	10			1.34 (1.00-1.80)		
Unknown						
Age at sexual debut			<i>P</i> = 0.771	-	-	-
≤15	245	26.3	1.00			
16-17	282	30.3	0.88 (0.61-1.26)			
≥ 18	404	43.4	0.93 (0.67 -1.31)			
Unknown*	153					

 Table 4- 2: Univariate and multivariate analyses of risk factors for STIs

Variable	Number	%	Unadjusted OR	Adjusted OR	Final model	P-value
Category	1084	100	-	N = 1067	N = 1078	0.007 [¥]
Wealth quintile			<i>P</i> = 0.018	<i>P</i> = 0.008	-	-
Lowest	217	20.0	1.00	1.00		
Second	221	20.4	1.13 (0.76-1.68)	1.00 (0.67-1.51)		
Middle	213	19.7	1.03 (0.69-1.54)	0.79 (0.51-1.20)		
Fourth	215	19.9	1.44 (0.97-2.13)	1.02 (0.67-1.55)		
Highest	216	20.0	0.71 (0.47-1.08)	0.52 (0.33-0.82)		
Missing*	2					
HIV status			<i>P</i> = 0.072	P = 0.064	-	-
Negative	940	86.8	1.00	1.00		
Positive	143	13.2	1.39 (0.97-1.99)	1.44 (0.98-2.10)		
Missing*	1					
Bacterial vaginosis			<i>P</i> = 0.034	<i>P</i> = 0.037		
Negative	375	34.7	1.00	1.00	1.00	
Intermediate	184	17.0	1.37 (0.95-2.00)	1.52 (1.04-2.23)	1.38 (0.95-2.00)	0.091
Positive	521	48.3	1.44 (1.08-1.91)	1.40 (1.04-1.87)	1.44 (1.08-1.92)	0.012
	4					

Table 4-2: Univariate and multivariate analyses of risk factors for STIs

-: Blank

[¥]*P*-value for overall final model

P-values appearing with unadjusted odds ratios are for each univariate model indicating associations of the variable with the outcome at *P* < 0.1.

P-values appearing with adjusted odds ratios indicate the independent association of each variable with the outcome at *P* < 0.05 Bold type- statistically significant outcome for multivariate analysis and final model.

When wealth quintile was put in the model with BV, it was not independently associated with infection with at least one STI and was therefore excluded from the final model.

*Missing/unknown values are only presented as numbers and were not included in the calculation of percentages and in the models

Potential risk factor	Number	%	Unadjusted OR	Adjusted OR	Final model	P-value
	1081	100	-	N= 1079	N = 1079	0.001 [¥]
Age group			<i>P</i> = 0.050	<i>P</i> = 0.120	-	-
≤ 20	294	27.3	1.00	1.00		
21-25	307	28.4	1.46 (1.04-2.03)	1.43 (0.93-2.20)		
26-30	246	22.8	1.43 (1.00-2.06)	1.45 (0.89-2.39)		
≥ 30	234	21.6	1.02 (0.72-1.46)	1.04 (0.63-1.72)		
Maternal education			P = 0.550	-	-	-
0-6 years	425	39.3	1.00			
≥ 7 years	656	60.7	1.08 (0.84-1.39)			
Marital status			<i>P</i> = 0.066	<i>P</i> = 0.127	-	-
Single	201	18.6	1.00	1.00		
Married	872	80.7	1.46 (1.06-1.99)	1.50 (1.01-2.24)		
Separated/widowed/divorced	8	0.7	1.20 (0.27-5.14)	1.19 (0.26-5.39)		
Gravidity			<i>P</i> = 0.191	<i>P</i> = 0.261	-	-
Primigravidae	259	23.9	1.00	1.00		
secundigravidae	164	15.2	1.47 (0.97-2.24)	0.99 (0.60-1.64)		
Multigravidae	658	60.9	1.14 (0.84-1.54)	0.73 (0.44-1.20)		
No. of life time sexual partners			<i>P</i> = 0.570	-	-	-
1 or 2	795	74.2	1.00			
3 or more	276	25.8	1.08 (0.81-1.45)			
Unknown*	10					
Age at sexual debut			<i>P</i> = 0.977	-	-	-
≤15	244	26.3	1.00			
16-17	282	30.4	1.04 (0.72-1.49)			
≥ 18	402	43.3	1.03 (0.74-1.44)			
Unknown*	153					

Table 4- 3: Univariate and multivariate analyses of risk factors for BV

Potential risk factor	Number	%	Unadjusted OR	Adjusted OR	Final model	P-value
	1081	100	-	N = 1079	N = 1079	0.001 [¥]
Wealth quintile			<i>P</i> = 0.484	-	-	-
Lowest	217	20.0	1.00			
Second	221	20.5	1.10 (0.75-1.62)			
Middle	214	19.8	1.39 (0.93-2.03)			
Fourth	213	19.7	1.36 (0.91-2.02)			
Highest	215	19.9	1.13 (0.76-1.67)			
Missing*	2					
HIV status			P = 0.001	<i>P</i> = 0.002		
Negative	937	86.8	1.00	1.00	1.00	
Positive	143	13.2	1.91 (1.27-2.89)	1.88 (1.24-2.86)	1.87 (1.24-2.83)	0.003
Missing*	1					
STI			<i>P</i> = 0.010	P = 0.016		
Negative	708	65.6	1.00	1.00	1.00	
Positive	372	34.4	1.42 (1.09-1.87)	1.40 (1.06-1.85)	1.40 (1.07-1.84)	0.015
Missing	1					

Table 4-3: Univariate and multivariate analyses of risk factors for BV continued

-: Blank

[¥]*P*-value for overall final model

P-values appearing with unadjusted odds ratios are for each univariate model indicating associations of the variable with the outcome at *P* < 0.1.

P-values appearing with adjusted odds ratios indicate the independent association of each variable with the outcome at *P* < 0.05

Bold type- statistically significant outcome for multivariate analysis and final model.

*Missing/unknown values are only presented as numbers and were not included in the calculation of percentages and in the models

4.4 Risk factors for malaria and curable STI/RTI coinfection

In the univariate analysis the following factors were associated with malaria and STI/RTI coinfection (P < 0.1); age group, gravidity, bed net ownership, bed net usage and HIV status (Table 4-3). None of the factors tested using the likelihood ratio test were independently associated with coinfection except HIV infection. HIV-infected women were at a higher risk of being coinfected with malaria and at least one curable STI/RTI than HIV-uninfected women [OR; 3.59 (95% CI, 1.73-7.48), P < 0.001].

Potential risk factor	Number	%	Unadjusted OR	Adjusted OR	Final model	P-value
	1071	100	-	N = 931	N = 1071	< 0.001 [¥]
Recruitment site			<i>P</i> = 0.273	-	-	-
Kashikishi	734	68.5	1.00			
Nchelenge	337	31.5	1.22(0.85-1.74)			
Age group			<i>P</i> = 0.026	<i>P</i> = 0.149	-	-
≤ 20	295	27.5	1.00	1.00		
21-25	304	28.3	0.84 (0.53-1.34)	0.95 (0.52-1.72)		
26-30	244	22.7	0.71 (0.44-1.13)	0.83 (0.43-1.61)		
≥ 30	231	21.5	0.51 (0.32-0.81)	0.58 (0.30-1.14)		
Marital status			<i>P</i> = 0.751	-	-	-
Single	201	18.8	1.00			
Married	862	80.5	0.91 (0.60-1.39)			
Divorced/separated/widowed	8	0.7	0.55 (0.11-2.84)			
Gravidity			<i>P</i> = 0.039	<i>P</i> = 0.681	-	-
Primigravidae	258	24.1	1.00	1.00		
Secundigravidae	165	15.4	1.95 (0.53-1.70)	0.96 (0.52-1.88)		
Multigravidae	648	60.5	0.63 (0.42-0.95)	0.77 (0.41-1.46)		
Bed net ownership			<i>P</i> = 0.064	<i>P</i> = 0.937	-	-
No	734	68.5	1.00	1.00		
Yes	337	31.5	0.74 (0.53-1.02)	0.83 (0.47-1.47)		
Bed net use on previous night			<i>P</i> = 0.097	<i>P</i> = 0.386	-	-
No	658	61.6	1.00	1.00		
Yes	410	38.4	0.76 (0.55-1.05)	0.98 (0.56-1.74)		
Missing*	3					

 Table 4- 4: Risk factors for malaria and curable STI/RTI coinfection in pregnant women of Nchelenge District

Potential risk factor	Number	%	Unadjusted OR	Adjusted OR	Final model	P-value
	1071	100	-	N = 931	N = 1071	< 0.001 [¥]
IRS in past 12 months			<i>P</i> = 0.193	-	-	-
No	798	74.7	1.00			
Yes	224	21.0	0.78 (0.53-1.15)			
Unknown*	46					
Missing*	3					
No. of life time sexual partners			<i>P</i> = 0.793	-	-	-
1 or 2	788	74.2	1.00			
3 or more	274	25.8	1.47(1.06-2.04)			
Unknown*	9					
Age at sex debut			<i>P</i> = 0.593	-	-	-
>14 years	97	10.4	1.00			
14 years & above	821	89.6	1.88 (0.49-1.60)			
Unknown*	153					
Wealth quintile			<i>P</i> = 0.166	-	-	-
Lowest	215	20.1	1.00			
Second	219	20.5	0.77 (0.46-1.26)			
Middle	208	19.5	1.07 (0.63-1.82)			
Fourth	214	19.9	1.35 (0.77-2.34)			
Highest	214	20.0	0.76 (0.46-1.26)			
Missing*	2					
HIV status			<i>P</i> < 0.001	P < 0.001		
Negative	933	87.1	1.00	1.00	1.00	
Positive	138	12.9	3.59 (1.73-7.48)	3.93 (1.87-8.27)	3.59 (1.73-7.48)	< 0.001
Missing*	1					

Table 4-4: Risk factors for malaria and curable STI/RTI coinfection in pregnant women of Nchelenge District continued

-: Blank; [¥]*P*- value for overall final model

P-values appearing with unadjusted odds ratios are for each univariate model indicating associations of the variable with the outcome at P < 0.1.

P-values appearing with adjusted odds ratios indicate the independent association of each variable with the outcome at P < 0.05.

Bold type- statistically significant outcome for multivariate analysis and final model.

*Missing/unknown values are only presented as numbers and were not included in the calculation of percentages and in the models

Only HIV infection was found to be associated with malaria and STI/RTI coinfection therefore was not adjusted for any other variable in the final model.

5 Results (3): The prevalence of DHFR and DHPS mutations associated with SP resistance and *in vivo* efficacy, parasitological failure and prophylactic failure of SP in pregnant Zambian women

5.1 Chapter introduction

The work in this chapter was included in the thesis in order to present information on the status of parasite resistance in the study population. The chapter gives results of the prevalence of SP resistance markers in a random sample of 96 women who tested positive at first ANC visit. The chapter also presents results from the determination of *in vivo* efficacy of SP and its treatment and prophylactic failure over a period of 28 days in pregnant women who were tested for peripheral parasitaemia at both day 0 and day 28.

5.2 The prevalence of DHFR and DHPS mutations associated with SP resistance

Only 96 samples were processed based on the required sample size for the characterisation of SP resistance markers in this study population. The pyrimethamine resistant DHFR Asn-108 was high among the 96 *P. falciparum* infections that were assayed (94.8%, n = 91). Mixed infections of Asn-108 occurring with the wild type Ser-108 were observed in 3.1% (n = 3) of the samples. High levels of the DHFR IIe-51 and Arg-59 mutants (Figure 5.1) as well as the DHPS Gly-437 and Glu-540 were found (Figure 5.2).



Figure 5. 1: Prevalence of *P. falciparum* DHFR mutations in pregnant women of Nchelenge district



Figure 5. 2: Prevalence of *P. falciparum* DHPS point mutations in pregnant women of Nchelenge district

Table 5-1 presents the prevalence of the double, triple, quintuple and sextuple mutants. The DHPS double (Gly-437 + Glu-540) mutant was found in 70.8% (n = 68) and the DHFR triple (Asn-108 + IIe-51 + Arg-59) mutant in 92.7% (n = 89) of the samples that were assayed. The quintuple mutant (DHFR triple + DHPS double) was detected in 68.8% (n = 66). The sextuple mutant (DHFR triple + DHPS double + Arg-581) was observed in 9.4% (n = 9) of the overall total with 2 of these occurring as mixed infections with the wild type.

Number of P.				Mutant						
falciparum infections	DHFR	Triple	DHPS	Double	DHFR	/DHPS Quintuple	DHF	R/DHPS Sextuple		
	n	% (95% CI)	n	% (95% CI)	n	% (95% CI)	n	% (95% CI)		
96	89	92.7 (85.3- 96.5)	68	70.8 (60.8 -79.2)	66	68.8 (58.6 -77.34)	9	9.4 (4.2-16.0)		
95% CI: 95% confidence interval DHFR: Dihydrofolate reductase DHPS: Dihydropteroate synthase										

Table 5- 1: Antifolate multiple mutants among first ANC attendees of Nchelenge District, Zambia

Table 5-2 presents results of DHFR and DHPS genotyping of 96 samples. It highlights the number of samples that carried a particular combination of point mutations associated with SP resistance.

Table 5-2: Results of DHFR and DHPS genotyping of samples from 96 first antenata	I
care attendees of Nchelenge District	

n	dhfr51	dhfr59	dhfr108	dhps437	dhps540	dhps581
1	-	-	-	-	-	-
2	+	-	+	-	-	-
2	+	+	+	-	+	-
1	+	-	+	+	+	-
1	+	+	-	+	+	-
7	+	+	+	-	-	-
16	+	+	+	+	-	-
57	+	+	+	+	+	-
9	+	+	+	+	+	+
- = wild type						
+ = Mutant or mixed genotype						

5.3 Therapeutic efficacy of IPTp-SP over a 28-day period

Therapeutic efficacy of IPTp-SP was assessed based on malaria diagnosed by microscopy only. Of the 343 women who tested positive for malaria by microscopy at day 0, 60.3% (n = 207) were screened for malaria at day 28 post IPTp-SP to assess the therapeutic efficacy and prophylactic effect of SP. The number of women was higher than the 138 asymptomatic malaria positive pregnant women required for determination of therapeutic efficacy as earlier mentioned in the methods section. Of these women 14.3% (n = 49) were HIV-infected and 85.7% (n = 294) were HIV uninfected.

Sulphadoxine-pyrimethamine cleared parasitaemia in 70.0% (n = 145) of the 207 pregnant women who tested positive for malaria at enrolment and were tested at day 28. Slightly below a third of these women, 30.0% (n = 62) were positive at day 0 and at day 28 and were classified as cases of therapeutic failure before PCR correction.

Among those with a negative malaria smear test result at day 0, 52.0% (383/736) were also screened at day 28. Of the 383 women who tested negative at day 0, 12.3% (n = 47) tested positive for malaria at day 28 and 87.7% (n = 336) of those who were negative at day 0 maintained the malaria negative status by day 28. The 47 women who tested negative at day 0 and positive at day 28 were classified as cases of prophylactic failure before PCR correction.

5.4 Classification of parasitological failure cases based on MSP-2 genotyping

Sixty out of the 62 pairs of samples from the participants, who tested positive by microscopy at day 0 and day 28, were successfully processed. One dried blood spot sample among those collected at day 28 was missing and amplification of another sample from another pair was unsuccessful. Of the 60 pairs that were processed 46.7% (95% CI, 33.7-59.7, n = 28) were classified as recrudescence and the rest, 53.3% (95% CI, 40.3-66.3, n = 32) as reinfection (Figure 5.3).

After PCR correction, therapeutic efficacy without recrudescence occurred in 84.4% (173/205) of the women who were infected at day 0. Failure of SP to prevent reinfection occurred in 15.6% (32/205) of participants.

The proportion of therapeutic failure observed among primigravidae, 24.1% (21/87), and secundigravidae, 24.1% (7/29) was similar, however what was observed among multigravidae was much lower (7.9% (7/89), P = 0.009). Younger women were more likely to experience failure. Median ages of women who experienced therapeutic failure and those whose parasitaemia cleared were 20 (IQR, 18-22) and 22 (IQR, 19-28) years respectively and the difference in the distribution of age in the two groups was statistically significant (P = 0.001).

The median age of women who experienced prophylactic failure was 24 (IQR, 20-29) years and that of women who did not experience prophylactic failure was 26 (IQR, 22-32) years with the distribution of age in the two groups being statistically significant, (P = 0.03).

Table 5- 3: Frequency of prophylactic and treatment failure of sulphadoxinepyrimethamine among HIV negative first ANC attendees of Nchelenge District, Zambia

Category	Total	Number (%)	95% Confidence interval				
PCR-uncorrected							
Therapeutic efficacy	207	145 (70.0)	63.8 - 76.3				
Therapeutic failure	207	62 (30.0)	23.7 - 36.2				
Prophylactic failure	338	47 (12.3)	9.0 -15.6				
PCR-corrected							
Therapeutic efficacy	205	173 (84.4)	79.4-89.4				
Therapeutic failure	205	32 (15.6)	10.6-20.6				
PCR: Polymerase chain reaction							

Two samples could not be differentiated into recrudescence or reinfection because one dried blood sample among those collected at day 28 was missing and amplification of another sample from another pair was unsuccessful, thus the reduction in the denominator after PCR correction in the case of therapeutic efficacy and failure





6 Results (4) The prevalence of adverse birth outcomes among delivering participants and predictors of adverse birth outcomes

6.1 Chapter introduction

This chapter presents infant and maternal characteristics. Characteristics of delivering women are presented overall and according to recruitment site and follow-up with comparisons of proportions in the groups. The chapter also presents the prevalence of ABO among participating women according to parity and infection status. Proportions of ABO are also presented for each placental malaria histopathology class. The chapter also presents risk factors for composite ABO and for individual ABO (PD, LBW and IUGR). Analysis of risk factors for stillbirth was not conducted because the number of women who delivered still born babies was small.

6.2 Maternal and infant characteristics

6.2.1 Baseline characteristics of delivering participants

Table 6-1 shows baseline characteristics of women who were successfully followedup to delivery and those who were not. There were no differences in proportions of characteristics among women who were followed up to delivery and those who were lost to follow-up except in the case of gravidity. The proportion of primigravidae among women who were followed up to delivery was higher than that among women who were lost to follow-up (26.6 versus 18.6%, P = 0.004). A lower proportion of secundigravidae was observed among those followed up to delivery than among women lost to follow-up (13.4 versus 18.6%, P = 0.025). In the case of multigravidae, a higher proportion was observed among women lost to follow-up than among those followed to delivery but this was not statistically significant (62.8 versus 60.0%, P = 0.375).

variable	All (n = 1085)	Followed-up	Lost to follow-up	P-value
	n (%) or Median	(n = 729)	(n = 356)	
	(IQR)	n (%) or Median	n (%) or Median	
		(IQR)	(IQR)	
Age (median)	25 (20-30)	24 (20-30)	25 (20-31)	0.510
Recruitment site				
Kashikishi	747 (68.9)	508 (69.7)	239 (67.1)	0.395
Nchelenge	338 (31.2)	221 (30.3)	117 (32.9)	
Marital status				
Single	203 (18.7)	143 (19.9)	60 (16.4)	0.304
Married	874 (80.6)	569 (79.3)	305 (83.1)	
Divorced/separated/	8 (0.7)	6 (0.8)	2 (0.5)	
widowed				
Years of schooling				
None to 6	426 (39.3)	273 (37.4)	153 (43.0)	0.080
7 and above	659 (60.7)	456 (62.6)	203 (57.0)	
Gravidity				
Primigravidae	261 (24.1)	194 (26.6)	67 (18.6)	0.004
Secundigravidae	165 (15.2)	98 (13.4)	67 (18.6)	0.025
Multigravidae	659 (60.7)	437 (60.0)	222 (62.8)	0.375
Bed net ownership				
No	551 (50.8)	373 (51.2)	178 (50.0)	0.718
Yes	534 (49.2)	358 (48.8)	178 (50.0)	
Bed net use				
No	666 (61.6)	452 (62.3)	214 (60.1)	
Yes	416 (38.5)	274 (37.7)	274 (37.7)	0.718
Missing*	3	0	3	
IRS in previous 12				
months	808 (78.1)	553 (78.8)	255 (76.8)	0.475
No	226 (21.9)	149 (21.2)	77 (23.2)	
Yes	51	27	24	
Unknown*				
Wealth quintile				
Lowest	217 (20.0)	137 (18.8)	80 (22.6)	0.174
Second	221 (20.4)	148 (20.3)	73 (20.6)	
Middle	214 (19.8)	148 (20.3)	66 (18.6)	
Fourth	215 (19.9)	138 (18.9)	77 (21.8)	
Highest	216 (19.9)	158 (21.7)	58 (16.4)	
HIV status				
Negative	941 (86.8)	635 (87.1)	306 (86.2)	0.678
Positive	143 (13.2)	94 (12.9)	49 (13.8)	
Missing*	1	0	1	
Past experience of				
miscarriage				
No	708 (85.1)	463 (86.5)	245 (84.8)	0.486
Yes	116 (14.1)	72 (13.5)	44 (15.2)	
NA (primigravidae)*	261	194	67	

Table 6- 1: Baseline characteristics of study participants according to follow-up to delivery

variable	e All (n = 1085) Followed-up Lost to follow-up n (%) or Median (n = 729) (n = 356)		P-value					
	(IQR)	n (%) or Median (IQR)	n (%) or Median (IQR)					
Prior delivery of								
stillborn								
No	755 (91.6)	489 (91.4)	266 (92.0)	0.752				
Yes	69 (8.4)	46 (8.6)	23 (8.0)					
NA (primigravidae)*	261	194	67					
Prior delivery of								
a Preterm baby				0.110				
No	784 (95.3)	505 (94.4)	279 (96.9)					
Yes	39 (4.7)	30 (5.6)	9 (3.1)					
Missing*	1	0	1					
NA (primigravidae)	261	194	67					
Malaria infection PCR								
Negative	453 (42.2)	304 (42.1)	149 (42.3)	0.944				
Positive	621 (57.8)	418 (57.9)	203 (57.7)					
Missing*	11	4	7					
Microscopy								
Negative	736 (68.2)	491 (67.6)	245 (69.4)	0.557				
Positive	343 (31.8)	235 (32.4)	108 (30.6)					
Missing*	6	3	3					
STI/RTI								
Negative	384 (35.4)	262 (35.9)	122 (34.4)	0.611				
Positive	700 (64.6)	467 (64.1)	233 (65.6)					
Missing*	1	0	1					
Malaria & STI/RTI								
Coinfection								
No infection	177 (16.5)	119 (16.5)	58 (16.6)	0.782				
Malaria	202 (18.8)	142 (19.7)	60 (17.1)					
STI/RTI	278 (26.0)	186 (25.8)	92 (26.3)					
Coinfection	414 (38.7)	274 (38.0)	140 (414)					
Missing*	14	8	6					
IRS: Indoor residual spra	aying	-	-					
*Missing values/NA prir	nigravidae are only p	presented as number	s and were not include	ed in the				
calculation of percentage	calculation of percentages and in the Chi-squared test for association							

Table 6-1: Baseline characteristics of study participants according to follow-up to delivery continued

6.2.2 Characteristics of delivering women and infants

There were no known maternal deaths among participants. Table 6-2 presents characteristics of the 729 participants followed up to delivery by recruitment site. There was no statistically significant difference in the place of delivery or type of delivery between the two sites. Slightly more deliveries were conducted by TBAs among women recruited at Nchelenge than those recruited at Kashikishi.

The clinical factors (level of Hb, treatment of RPR positives women, IPTp-SP doses taken, and use of cotrimoxazole and ART, and antimalarial treatment) did not differ between the two sites.

The median gestational age at delivery was 38 (IQR, 37-39) weeks and the mean birth weight was 2784g (95% CI, 2752-2816). Among the 729 participants who were followed-up to delivery, 98.4% (n = 717) had a singleton delivery and the rest delivered twins. Of the 717 singleton deliveries 50.9% (n = 365) were female and 49.1% (n = 352) were male babies. The mean birth weight of female babies was lower than that of male babies (2738g versus 2831g) and this difference was statistically significant (P = 0.004).

With the exclusion of BV, 18.8% (n = 46) of the 248 STI-infected women received antibacterial treatment during the pregnancy duration. Of the 467 women who were infected with at least one STI/RTI only 12.8% (n = 60) received antibacterial treatment during the pregnancy duration. Of the 60 women who received antibacterial treatment only eight received drugs other than benzathine penicillin G and 10 of the remaining 52 received both benzathine penicillin G and a different antibiotic during the pregnancy duration based on syndromic management assessment. Antibacterial agents that were used were metronidazole, erythromycin and spectinomycin. Some of the STI/RTI-infected women (n = 40) had been treated with antifungal agents by the time of delivery and only two of these also received an antibiotic. Antifungal agents that were recorded were nystatin and fluconazole.

Coverage of IPTp-SP among participants was high with 99.2% of delivering women receiving at least one dose of IPTp-SP during the pregnancy duration; 0.8% (n = 6), 17.0% (n = 124), 43.0% (n = 314), 38.3% (n = 279) and 0.7% (n = 5) of participants received, none, one, two, three and four doses respectively.

132

Among the 92 HIV-infected delivering women, 37.0% (n = 34) were on cotrimoxazole prophylaxis. Of the 88 HIV-infected women who took at least one dose of IPTp-SP, 36.4% (n = 32) also took cotrimoxazole during the pregnancy duration.

Variable	All (N = 729)	Kashikishi (n = 508)	Nchelenge (n =221)	P-value
	n (%)	n (%)	n (%)	
Number of babies				
One	717 (98.4)	501 (98.6)	216 (97.7)	0.388
Two	12 (1.6)	7 (1.4)	5 (2.3)	
Sex of baby				
Female	365 (50.9)	248 (49.5)	117 (54.2)	0.252
Male	352 (49.1)	253 (50.5)	99 (45.8)	0.232
Twins	12	7	5	
Place of delivery				
Hospital	683 (93.8)	479 (94.4)	204 (92.3)	0.220
Health Centre	20 (2.8)	14 (2.8)	6 (2.7)	0.320
Home	25 (3.4)	14 (2.8)	11 (5.0)	
Delivered by				
Family member	22 (3.0)	13 (2.5)	9 (4.1)	0.015
ТВА	14 (1.9)	5 (1.0)	9 (4.1)	0.015
Midwife	651 (89.3)	457 (90.0)	194 (87.7)	
Doctor	42 (5.7)	33 (6.5)	9 (4.1)	
Type of delivery				
Vaginal	687 (94.2)	475 (93.5)	212 (95.9)	
C-section	42 (5.8)	33 (6.5)	9 (4.1)	0.197
Haemoglobin level				
Normal	585 (84 0)	406 (83 9)	179 (84 4)	
Anaemia	111 (16.0)	78 (16 1)	33 (15 6)	0.855
Missing*	33	24	9	
Antimalarial treatment	55	27		
hesides IPTn-SP				
No	635 (87 5)	440 (87 1)	195 (88 2)	0.649
Yes	91 (12 5)	65 (12 9)	26 (11 8)	
Missing*	3	3	0	
Iron supplementation	5	5	0	
No	15 (2 1)	13 (2.6)	2 (0.9)	
Ves	714 (97 9)	495 (97 4)	219 (99 1)	0.254
	714 (37.37	455 (57.4)	215 (55.1)	
IPTp-SP doses taken				
0 to 1	130 (17.9)	97 (19.1)	33 (14.9)	0 174
2 or more	598 (82.1)	410 (80.9)	188 (85.1)	0.174
Missing*	1	1	0	
Cotrimoxazole use				
among HIV+				0.547
No	58 (63.0)	41 (61.2)	17 (68.0)	0.547
Yes	34 (37.0)	26 (38.8)	18 (32.0)	
Missing*	2	1	1	
NA (HIV negative)*	635	440	195	
ART among HIV positive				
women				0.222
No	36 (37.2)	24 (35.3)	12 (46.2)	0.555
Yes	58 (61.7)	44 (64.7)	14 (53.8)	
NA (HIV negative)*	635	440	195	

 Table 6- 2: Characteristics of delivering participants according to recruitment site

Variable	All (N = 729) n (%)	Kashikishi (n = 508) n (%)	Nchelenge (n = 221) n (%)	P-value
Treatment among RPR				
Reactive				
No	29 (35.8)	21 (38.9)	8 (29.6)	0.413
Yes	52 (64.2)	33 (61.1)	21 (70.4)	
Non-reactive	648	454	194	
Anti-bacterial [‡]				
treatment among all				
STI/RTI positives				
No	407 (87.2)	284 (87.1)	123 (87.2)	0 972
Yes	60 (12.8)	42 (12.9)	18 (12.8)	0.572
NA (Negative)*	262	182	80	

Table 6-2: Characteristics of delivering participants according to recruitment site continued

* Missing and NA (not applicable) values are only presented as numbers and were not included in the calculation of percentages and in the Chi-squared test for association

IPTp-SP: Intermittent preventive treatment in pregnancy with sulphadoxine-pyrimethamine ART: Antiretroviral therapy. Anaemia was defined as haemoglobin level < 11grams/decilitre [‡]Antibacterial treatment included benzathine penicillin G, metronidazole, erythromycin and spectinomycin.

6.3 Prevalence of adverse birth outcomes

Overall the prevalence of ABO among the 717 women with singleton deliveries was 35.0%. Women who experienced at least one ABO were younger than women who had normal birth outcome, median age 22 (IQR, 19-28) verses 25 (IQR, 21-31) years, with the age distribution being significantly different in the two groups, P < 0.001, Mann-Whitney U test. Preterm delivery and LBW were the most common ABOs, 22.1% and 21.9% respectively.

6.3.1 Prevalence of adverse birth outcome according to parity

The prevalence of composite ABO was 53.4% among primiparae, 31.2% among para II and 27.8% among multiparous women, P < 0.001. The prevalence of LBW, PD and IUGR were highest among primiparae followed by para II and lowest among multiparae as follows; PD 29.5%, 20.8% and 18.9%, P = 0.013; LBW 41.1%, 19.8% and 13.8%, P < 0.001; IUGR 31.6%, 13.2% and 9.3%, P < 0.001. None of the para II women experienced stillbirth. The prevalence of stillbirth was higher among primiparae than multiparae but the difference was not statistically significant, 2.6% versus 1.6%, P = 0.309 (Table 6-3).

Adverse	All	95% Confidence	Primiparae	Para II	Multiparae	P-value
	N = 717	interval [‡]	n = 193	n = 96	n = 428	
outcome	n (%)		n (%)	n (%)	n (%)	
PD						
No	558 (77.9)	74.9-81.0	136 (70.5)	76 (79.2)	346 (81.0)	0.013
Yes	158 (22.1)	19.2-25.3	57 (29.5)	20 (20.8)	81 (18.9)	
Missing*	1					
LBW						
No	559 (78.1)	75.0-81.1	113 (58.9)	77 (80.2)	369 (86.2)	< 0.001
Yes	157 (21.9)	19.0-25.0	79 (41.1)	19 (19.8)	59 (13.8)	
IUGR						
No	473 (84.8)	81.8-87.8	93 (68.4)	66 (86.8)	314 (90.8)	< 0.001
Yes	85 (15.2)	12.5-18.5	43 (31.6)	10 (13.2)	32 (9.3)	
Na (preterm)*	158					
Missing*	1					
Stillbirth						
No	705 (98.3)	97.4-99.3	188 (97.4)	96 (100)	421 (98.4)	0.309
Yes	12 (1.7) ⁺	1.0-2.9	5 (2.6)	0 (0.0)	7 (1.64)	
Composite ABO						
No	466 (65.0)	61.5-68.5	90 (46.6)	66 (68.8)	310 (71.4)	< 0.001
Yes	251 (35.0)	31.6-38.6	103 (53.4)	30 (31.2)	118 (27.6)	

Table 6-3: Frequency of adverse birth outcomes among participants with singleton deliveries according to parity

⁺The percentage of stillbirths (1.7%) consisted of 1.0% macerated and 0.7% fresh stillbirths. One woman among those with twin deliveries had spontaneous abortion at 26 weeks.

*Missing values are only presented as numbers and were not included in the calculation of percentages and in the Chi-squared test for association LBW: low birth weight

PD: preterm delivery

IUGR: Intrauterine growth retardation

ABO: Adverse birth outcome

6.3.2 Prevalence of adverse birth outcome among infected and uninfected women

Table 6-4 presents the prevalence of PD, LBW, IUGR and stillbirth) among women classified as infected (infected with at least one STI/RTI or malaria) or uninfected at first ANC attendance. Although not statistically significant, the prevalence of PD, LBW and IUGR was slightly higher among infected women than among uninfected women. The frequency of stillbirth was slightly higher among uninfected than infected women, 1.7% versus 1.5% but this could be due to chance (P = 0.880). The mean birth weight of babies born to uninfected women was higher than those born to infected women but this difference was not statistically significant (2821g versus 2774g, P = 0.285).

Table 6- 4: Prevalence of preterm delivery, low birth weight, intra-uterine growth retardation and stillbirth among women classified as infected and uninfected at first ANC visit

Adverse birth	All	Infected	Uninfected	P-Value
outcome	N = 709	N = 117	N = 592	
	n (%)	n (%)	n (%)	
PD				
No	550 (77.7)	93 (79.5)	457 (77.3)	0.608
Yes	158 (22.3)	24 (20.5)	134 (22.7)	
Missing*	1	0	1	
LBW				
No	551 (77.8)	95 (81.2)	456 (77.2)	0.337
Yes	157 (22.2)	22 (18.8)	135 (22.8)	
Missing*	1	0	1	
Stillbirth				
No	698 (98.5)	15 (98.3)	583 (98.5)	0.880
Yes	11(1.5)	2 (1.7)	9 (1.5)	
IUGR				
No	465 (84.5)	82 (88.2)	383 (83.8)	0.288
Yes	85 (15.5)	11 (11.8)	74 (16.2)	
NA (Pre-term)*	158			
Missing*	1			
All				
No	459 (64.7)	81 (69.2)	378 (63.9)	0.266
Yes	250 (35.3)	36 (30.8)	214 (36.1)	

N: Number, was equal to 709, 8 of the 717 participants with singleton deliveries could not be classified as infected or uninfected at recruitment due to missing results and negative outcomes in those available

LBW: low birth weight

PD: preterm delivery

IUGR: Intrauterine growth retardation

NA: Not applicable

* Missing values and NA(pre-term) are only presented as numbers and were not included in the calculation of percentages and in the Chi-squared test for association

6.4 Placental malaria burden

Of the 702 placental tissue samples, placental malaria was identified in 40.9% (n = 287) by histopathology, the majority of these, 30.3% (n = 213) were classified as past infections (class 4). Placental histopathology identified 8.1% (n = 57) chronic infections (class 2) and 2.4% (n = 17) active infections.

The median gestational age of women with placental malaria infection regardless of the pathology class was 38 weeks (IQR, 36-39) and that of uninfected women was 38 weeks (IQR, 37-39) with a statistically significant difference in the distribution of gestational age in the two groups, P = 0.017.

The mean birthweight, mean Hb, and median gestational age in the different placental pathology classes and comparison of each class with class 5 (the uninfected) are presented in Table 6-5.

The median gestational age of women with chronic placental malaria infection was lower than that of uninfected women and the distribution of gestational age was significantly different in the two groups 37 (IQR, 35-39) weeks and 38 (IQR, 37-39) weeks, P = 0.001. The mean birth weight of babies born to mothers with chronic placental malaria infection was significantly lower than the mean birthweight of babies born to uninfected mothers, 2537 grams versus 2834 grams, P = 0.001. Furthermore, the mean Hb of women with chronic placental malaria infection was significantly lower than that of uninfected women, 12.2g/dl versus 13.0g/dl, P = 0.024. Table 6- 5: Association between placental pathology class and birthweight, maternal haemoglobin and gestational age among study participants with singleton deliveries

Pathology class (Number)	Description	Mean Birth weight in grams (95% CI)	Probability compared with uninfected (class 5)	Maternal Haemoglobin Geometric Mean (95% CI)	Probability compared with uninfected (class 5)	Median gestational age (IQR)	Probability compared with uninfected (class 5)
1 (17) Active infection	Parasites, No pigment in monocytes or fibrin	2765 (2561-2969)	0.5022	13.2 (12.2-14.4)	0.731	38 (37-38)	0.284
2 (57) Chronic infection	Parasites, pigment in monocytes +/- fibrin	2537 (2408-2665)	< 0.001	12.2 (11.2-13.2)	0.026	37 (35-39)	0.001
3 (0)* Chronic infection	Parasites, pigment in Fibrin	-	-	-	-	-	-
4 (213) Past infection	No parasites, pigment only	2767 (2706-2828)	0.066	12.8 (12.5-13.3)	0.521	38 (37-39)	0.212
5 (415) No infection	No parasite or pigment (No infection)	2834 (2793-2874)	Reference group	13.0 (12.8-13.3)	Reference group	38 (37-39)	Reference group
-: Blank, no sp	ecimen was classifi	ed under class 3					

6.4.1 Frequency of adverse birth outcome in placental malaria histopathology classes

Table 6-6 presents the prevalence of LBW, PD, IUGR and stillbirth among 691 women with singleton deliveries and placental histology results. The highest proportion of each of these four ABO was observed among babies born to mothers with placental infection classified under pathology class 2 as follows: PD among women with chronic infection was 37.5% versus 23.5%, 24.0% and 18.1%, in women with active infection, past infection and no infection respectively, P = 0.007; LBW in women with chronic infection was 38.5% versus 17.7%, 24.0% and 18.1% in women with active infection, past infection and no infection respectively, P = 0.004; IUGR was 17.7% among women with chronic infection against 7.7% 16.3% and 14.9% in women with active infection, past infection and no infection respectively, P = 0.827 and the prevalence of stillbirth in women with chronic infection was 5.3% versus 0%, 1.9% and 1.6% in women with active infection, past infection and no infection respectively, P = 0.093. The overall prevalence of ABO was also highest among women with chronic placental malaria infection, 50.9%; and in the other classes the prevalence was 29.4% among women with active infection; 38.0% among women with past infection and 31.0% among uninfected women, P = 0.017.

Table 6-	6: Prevalence of low birth w	veight, preterm	delivery,	intrauterine	growth	retardation	and stillbirth	according to	pathology	class
among	participants with singleton d	leliveries								

Pathology class	All	1	2	3	4	5	
Adverse birth	All	Parasites, No	Parasites, pigment	Parasites,	No parasites,	No parasites,	P-Value
outcome	N = 691	pigment in	in monocytes +/-	pigment in Fibrin	pigment only	no pigment	
	n (%)	monocytes or fibrin	fibrin	(Chronic	(past infection)	(No infection)	
		(active infection)	(Chronic infection)	infection)	n (%)	n (%)	
		n (%)	n (%)	n (%)			
PD							
No	542 (78.4)	13 (76.5)	35 (62.5)	-	158 (76.0)	335 (81.9)	0.007
Yes	149 (21.6)	4 (23.5)	21 (37.5)		50 (24.0)	74 (18.1)	
Missing*	1	0	1		0	0	
LBW							
No	542 (78.4)	14 (82.3)	35 (61.4)	-	158 (76.0)	81.9 (335)	0.004
Yes	149 (21.6)	3 (17.7)	22 (38.6)		50 (24.0)	74 (18.1)	
IUGR							
No	458 (84.7)	12 (92.3)	28 (82.3)	-	133 (83.7)	285 (85.1)	0.827
Yes	83 (15.4)	1 (7.7)	6 (17.7)		26 (16.3)	50 (14.9)	
NA (preterm)*	150						
Stillbirth							
No	680 (98.4)	17 (100)	54 (94.7)	-	204 (98.1)	405 (99.4)	0.098
Yes	11 (1.6)	0 (0)	3 (5.3)		4 (1.9)	4 (1.6)	
Composite ABO							
No	51 (65.3)	12 (70.6)	28 (49.1)		129 (62.0)	282 (69.0)	0.017
Yes	240 (34.7)	5 (29.4)	29 (50.9)		79 (38.0)	127 (31.0)	

LBW: low birth weight, PD: preterm delivery, IUGR: Intrauterine growth retardation, NA: Not applicable

-: Blank, no specimen was classified under pathology class 3

*Missing and NA (preterm) values are only presented as numbers and were not included in the calculation of percentages and in the Chisquared test for association

6.5 Prevalence of adverse birth outcome according to antimalarial and antibiotic treatment

Table 6-7 shows proportions of ABO in different categories of STIs/RTIs and malaria treatment taken during the pregnancy duration. The proportion of LBW babies delivered among those who were uninfected, infected and treated and infected and untreated women was 20.9%, 10.0% and 24.4% (P = 0.039) respectively. The proportion of LBW babies delivered among women who received antimalarial treatment was higher than that observed among women who were not treated for malaria during pregnancy (34.4 versus 20.2, P = 0.002).
Factor	Number	Composite Adverse	Preterm delivery	Low birth weight	Intra-uterine birth
	in	birth outcome			retardation
	category	Number (%)	Number (%)	Number (%)	Number (%)
Syphilis		<i>P</i> = 0.840	<i>P</i> = 0.352	<i>P</i> = 0.449	<i>P</i> = 0.423
No	637	223 (35.0)	136 (21.4)	142 (22.3)	80 (16.0)
Yes-treated	52	17 (32.7)	13 (25.0)	8 (15.4)	3 (7.7)
No-untreated	28	11 (39.3)	9 (32.1)	7 (25.9)	2 (10.5)
STI/RTI including syphilis		<i>P</i> = 0.566	<i>P</i> = 0.912	<i>P</i> = 0.037	<i>P</i> = 0.261
No	259	88 (34.0)	50 (19.3)	54 (20.9)	33 (15.8)
Yes-treated	60	18 (30.0)	15 (25.0)	6 (10.0)	3 (6.7)
Yes- untreated	398	145 (36.4)	93 (23.4)	94 (24.4)	49 (16.2)
Antimalarial treatment*		<i>P</i> = 0.195	<i>P</i> = 0.912	<i>P</i> = 0.002	<i>P</i> = 0.217
No	464	213 (34.1)	137 (22.0)	126 (20.2)	71 (14.6)
Yes	250	37 (41.1)	20 (22.5)	31 (34.4)	14 (20.3)
*Antimalarial treatment (r	not prophyla	axis) during the pregnar	ncy duration		
Bold type: Significant outc	ome (Chi-sc	juare test)			

 Table 6- 7: Proportions of adverse birth outcome in antimalarial and antibacterial treatment categories

6.6 Protective and risk factors for composite adverse birth outcomes

In the univariate analyses the following factors were associated with ABO; height, age, parity, sex of the baby, placental malaria, hypertension and the number of doses of IPTp-SP taken during pregnancy. In the univariate analysis women who were mono-infected (malaria or at least one STI/RTI) or coinfection had 24% and 31% increased risk of experiencing an ABO, respectively compared to uninfected women but this was not statistically significant. None of the STIs/RTIs or malaria diagnosed at first ANC were associated with composite ABO. Women infected with STI/RTI and treated had 17% reduced risk and STI/RTI-infected and untreated women had 11% increased risk of experiencing an ABO than uninfected but this was not statistically significant.

In the multivariate analysis short stature (< 1.5 metres), parity and the number of doses of IPTp-SP were associated with ABO (Table 6-8). Women with a height < 1.5 metres had a higher risk of experiencing at least one ABO after adjusting for the number of doses of IPTp-SP taken during the pregnancy duration and parity (aOR 1.55; 95% CI, 1.10-2.18, P = 0.013). The risk of having an ABO among para II was less than half of the risk observed in primiparous (aOR 0.42; 95% CI, 0.28-0.63, P < 0.001) women and this was even much lower among multiparous women (aOR 0.32; 95% CI, 0.22-0.48, P < 0.001). After taking account of parity and height, women who had taken 2 or more doses of SP during their pregnancy had a lower risk of experiencing an adverse delivery outcome than women who had taken one dose or none (aOR 0.51; 95% CI, 0.34-0.76, P = 0.001). There was no interaction between parity and height (P = 0.036), parity and doses of IPTp-SP (P = 0.067), height and doses of IPTp-SP (P = 0.096).

To check if the effects of malaria, STI/RTIs and their coinfection on birth outcome were modified by the site of recruitment, these variables were individually included in the final model with the variable for recruitment site. There was no interaction between malaria and site of recruitment (P = 0.672), STI/RTI and recruitment site (P = 0.864) and coinfection and the site of recruitment (P = 0.809).

Potential risk factor	Number	%	Unadjusted OR	Adjusted OR	Adjusted OR	P-value
					(Final model)	
	717	100	-	N = 595	N = 715	< 0.001 [¥]
Recruitment site			<i>P</i> = 0.329	-	-	-
Kashikishi	501	69.9	1.00			
Nchelenge	216	30.1	0.84 (0.59-1.19)			
Age group			<i>P</i> < 0.001	<i>P</i> = 0.716	-	-
≤ 20	204	28.5	1.00	1.00		
21-25	205	28.6	0.52 (0.35-0.77)	1.26 (0.69-2.29)		
26-30	148	20.6	0.43 (0.27-0.66)	1.14 (0.56-2.35)		
≥ 30	160	22.3	0.34 (0.22-0.54)	1.01(0.48-2.10)		
Years of schooling			<i>P</i> = 0.247	-	-	-
0-6 years	266	37.1	1.00			
7 years and above	451	62.9	1.21 (0.88-1.66)			
Maternal height			<i>P</i> = 0.002	<i>P</i> = 0.013		
≥ 1.5m	503	70.2	1.00	1.00	1.00	
< 1.5m	213	29.8	1.69 (1.21-2.35)	1.57 (1.06-2.32)	1.55 (1.10-2.18)	0.013
Missing*	1					
Marital status			<i>P</i> = 0.03	<i>P</i> = 0.512	-	-
Single	142	19.8	1.00	1.00		
Married	569	79.4	0.57 (0.39- 0.82)	1.14 (0.66-1.97)		
Separated/widowed/divorced	6	0.8	0.24 (0.03-2.03)	-		
Parity			<i>P</i> < 0.001	<i>P</i> < 0.001		
Primiparae	193	27.0	1.00	1.00	1.00	
Para II	96	13.4	0.39 (0.24-0.67)	0.27 (0.13-0.57)	0.41 (0.28-0.63)	< 0.001
Multigravidae	428	59.6	0.33 (0.23-0.47)	0.31 (0.15-0.61)	0.32 (0.20-0.43)	< 0.001
Past experience of miscarriage			P = 0.023	-	-	-
No	457	87.2	1.00			
Yes	67	12.8	1.87 (1.10-3.19)			
Primigravidae*	193					

Table 6-8: Risk factors for composite adverse birth outcomes among study participants with singleton deliveries

Potential risk factor	Number	%	Unadjusted OR	Adjusted OR	Adjusted OR	P-value
					(Final model)	
	717	100	-	N = 595	N = 715	< 0.001 [¥]
Prior delivery of still born			P = 0.184	-	-	-
No	481	91.8	1.00			
Yes	43	8.2	1.57 (0.82-3.00)			
Primigravidae*	193					
Prior delivery of a Preterm baby			<i>P</i> = 0.733	-	-	-
No	495	94.5	1.00			
Yes	29	5.5	1.15 (0.51-2.59)			
Primigravidae*	193					
Peripheral malaria infection ^a			<i>P</i> = 0.181	-	-	-
Negative	301	42.4	1.00			
Positive	409	57.6	1.24 (0.90-1.69)			
Missing*	7					
Placental malaria infection			<i>P</i> = 0.015	<i>P</i> = 0.149	-	-
Negative	409	59.3	1.00	1.00		
Positive	282	40.7	1.48 (1.08-2.03)	1.28 (0.89-1.84)		
Missing*	26					
Syphilis			<i>P</i> = 0.842	-	-	-
Negative	637	88.8	1.00			
Positive-Treated	52	7.3	0.90 (0.49-1.65)			
Positive-Untreated	28	3.9	1.20 (0.55-2.61)			
Chlamydia trachomatis			<i>P</i> = 0.971	-	-	-
Negative	683	95.3	1.00			
Positive	34	4.7	1.01 (0.49-2.08)			
Neisseria gonorrhoeae			<i>P</i> = 0.893	-	-	-
Negative	695	36.9	1.00			
Positive	22	3.1	1.06 (0.44-2.57)			

 Table 6-8: Risk factors for composite adverse birth outcomes among study participants with singleton deliveries continued

Potential risk factor	Number	%	Unadjusted OR	Adjusted OR	Adjusted OR	P-value
					(Final model)	
	717	100	-	N = 595	N = 715	< 0.001 [¥]
Trichomonas vaginalis			<i>P</i> = 0. 121	-	-	-
Negative	547	76.3	1.00			
Positive	170	23.7	1.33 (0.93-1.89)			
Bacterial vaginosis			<i>P</i> = 0.539	-	-	-
Normal	246	34.4	1.00			
Intermediate	130	18.2	1.01 (0.65-1.58)			
Positive	339	47.4	0.84 (0.60-1.19)			
Missing*	2					
STI/RTI			<i>P</i> = 0.562	-	-	-
Uninfected [‡]	259	36.1	1.00			
Infected & treated	60	8.4	0.83 (0.45-1.53)			
Infected & untreated	398	55.5	1.11 (0.80-1.55)			
Malaria and STI/ RTI coinfection ^b			<i>P</i> = 0.507	-	-	-
No infection	117	16.5	1.00			
Single infection ^c	326	46.1	1.24 (0.79-1.96)			
Coinfection ^b	266	37.5	1.31 (0.82-2.09)			
Missing*	8					
Anaemia ^{\$}			<i>P</i> = 0.255	-	-	-
Normal	574	83.8	1.00			
Anaemic	111	16.2	1.28 (0.83-1.96)			
Missing*	32					
Hypertension			<i>P</i> = 0.027		-	-
Normal	608	97.8	1.00	1.00		
Hypertensive	14	2.2	3.36 (1.11-10.2)	0.73 (0.50-1.05)		
Missing*	95					
Sex of baby			<i>P</i> = 0.055	<i>P</i> = 0.103	-	-
Female	365	50.9	1.00	1.00		
Male	352	49.1	0.74 (0.54-1.00)	0.73 (0.50-1.05)		

Table 6-8: Risk factors for composite adverse birth outcomes among study participants with singleton deliveries continued

Potential risk factor	Number	%	Unadjusted OR	Adjusted OR	Final model	P-value
	717	100	-	N = 595	N = 715	< 0.001 [¥]
Antimalarial treatment			<i>P</i> = 0.199	-	-	-
No	464	65.0	1.00			
Yes	250	35.0	1.35 (0.86-2.11)			
Missing*	3					
Number of doses of IPTp-SP			<i>P</i> < 0.001	<i>P</i> = 0.015		
0 or 1	126	17.6	1.00	1.00	1.00	
2 or more	590	82.4	0.57 (0.38-0.84)	0.56 (0.36-0.89)	0.51 (0.34-0.76)	0.001
Missing*	1					

Table 6-8: Risk factors for composite adverse birth outcomes among study participants with singleton deliveries continued

^aMalaria infection at first ANC attendance; ^bInfection with at least one STI/RTI and malaria; ^cInfection with at least one STI/RTI or malaria IPTp-SP – Intermittent preventive treatment in pregnancy with sulphadoxine-pyrimethamine

^{*}*P-value* for overall final model

P-values appearing with unadjusted odds ratios are for each univariate model indicating associations of the variable with the outcome at P < 0.1. *P*-values appearing with adjusted odds ratios indicate the independent association of each variable with the outcome at P < 0.05.

Bold print: statistically significant outcome for multivariate analysis and final model

^{\$} Normal haemoglobin level (Hb) was defined as Hb \geq 11grams per decilitre and anaemia as Hb < 11 grams per decilitre.

Number of participants less than 715) (Number of singleton deliveries) are indicated in the first column. These were due to missing values in individual variables

*Missing/primigravidae values are only presented as numbers and were not included in the calculation of percentages and in the models When placental malaria was put in a model with maternal height, parity and number of doses of IPTP-SP, it was not independently associated with adverse birth outcomes.

[†]Antibacterial treatment included benzathine penicillin, metronidazole, erythromycin and spectinomycin.

⁺ Uninfected in relation to STI/RTI among those who were followed-up to delivery (malaria was not considered)

Past experience of a miscarriage was associated with adverse birth outcome in the univariate analysis but was not included in the multivariable model because that would have excluded all primiparous women from the model.

6.7 Risk factors for PD, LBW and IUGR

Univariate and multivariate analyses of factors associated with individual ABO (PD, LBW and IUGR) are reported in Table 6-9. The number of women who experienced stillbirth was small (12) and was not amenable to statistical analyses. None of the individual STIs/RTIs or malaria diagnosed at first ANC or placental malaria diagnosed at delivery were associated with any one of the individual ABO.

Potential risk factors for ABO that were associated with individual ABO at *P* < 0.1 and were included in the multivariate models for each outcome were as follows: PD was associated with maternal age group, maternal height, parity, placental malaria, anaemia, and the number of doses of IPTp-SP taken during pregnancy; LBW was associated with maternal age group, maternal height, parity, infection with trichomoniasis, HIV status, placental malaria and antimalarial treatment during pregnancy; IUGR was associated with maternal age group, maternal age group, maternal height, parity, and peripheral malaria diagnosed at first ANC attendance.

In the multivariate analysis only the number of doses of SP taken during the pregnancy duration was independently associated with PD; LBW was associated with marital status, maternal height, parity, HIV and antimalarial treatment during the pregnancy duration; and IUGR was associated with parity and height.

The final models for factors associated with PD, LBW and IUGR are shown in Table 6-10. In the final model women who took 2 or more doses of SP had a reduced risk of experiencing PD (OR 0.34; 95% CI, 0.23-0.52) than women who took one dose or none.

Marital status was not independently associated with LBW in a model with height, parity, HIV and antimalarial treatment during the pregnancy duration hence it was excluded from the final model. Compared to primiparae, Para II women had a reduced risk of delivering LBW babies (aOR 0.32; 95% CI; 0.18-0.58) and the risk was slightly lower among multiparous women (aOR 0.22; 95% CI, 0.15-0.34). Women with height < 1.5 metres had increased risk of delivering a LBW baby than women whose height was \geq 1.5 metres (aOR 2.10, 95% CI, 1.43-3.08). Women who had been treated for malaria were at increased risk of experiencing LBW than women who had not been treated for malaria during the pregnancy duration (aOR 2.01; 95%

CI, 1.20-3.5). Women infected with HIV had twice the risk of delivering LBW babies than HIV-uninfected (aOR 2.10; 95% CI, 1.23-3.55).

A reduced risk of IUGR was observed among para II (aOR 0.33; 95% CI, 0.16-0.71) and multiparae (aOR 0.23; 95% CI, 0.13-0.37) in comparison to primiparae and babies born to shorter women were at increased risk of experiencing IUGR (aOR 1.71; 95% CI, 1.04-2.83) than those born to women whose height was \geq 1.5 metres.

Potential risk factor	Preterm delivery		Low birth weight		Intra-uterine growth retardation	
	Unadjusted OR	Adjusted OR	Unadjusted OR	Adjusted OR	Unadjusted OR	Adjusted OR
	-	N = 659	-	N = 659	-	N = 540
Recruitment site	<i>P</i> = 0.648	-	<i>P</i> = 0.900	-	<i>P</i> = 0.910	-
Kashikishi	1.00		1.00		1.00	
Nchelenge	1.09 (0.74-1.60)		1.02 (0.69-1.51)		0.97(0.58-1.16)	
Age group	<i>P</i> = 0.045	P = 0.812	<i>P</i> < 0.001	<i>P</i> = 0.677	<i>P</i> < 0.001	<i>P</i> = 0.918
≤ 20	1.00	1.00	1.00	1.00	1.00	1.00
21-25	0.61 (0.38-0.97)	0.85 (0.44-1.65)	0.48 (0.31-0.74)	1.42 (0.74-2.71)	0.48 (0.27-0.85)	1.34 (0.60-2.88)
26-30	0.63 (0.38-1.04)	0.88 (0.39-1.96)	0.31 (0.18-0.53)	1.20 (0.51-2.78)	0.28 (0.13-0.58)	0.95 (0.33-2.72)
≥ 30	0.54 (0.31- 0.87)	0.69 (0.31-1.56)	0.24 (0.13-0.42)	1.02 (0.41-2.55)	0.22 (0.11-0.47)	0.86 (0.29-2.52)
Height	<i>P</i> = 0.053	<i>P</i> = 0.262	<i>P</i> < 0.001	<i>P</i> = 0.002	P = 0.019	<i>P</i> = 0.026
≥ 1.5m	1.00	1.00	1.00	1.00	1.00	1.00
< 1.5m	1.45 (0.99-2.11)	1.28 (0.83-1.98)	2.25 (1.56-3.26)	1.90 (1.24-2.91)	1.80 (1.11-2.91)	1.81 (1.07-3.06)
Parity	<i>P</i> = 0.015	P = 0.150	<i>P</i> < 0.001	<i>P</i> < 0.001	<i>P</i> < 0.001	<i>P</i> = 0.005
Primiparae	1.00	1.00	1.00	1.00	1.00	1.00
Para II	0.63 (0.35-1.12)	0.47 (0.21-1.02)	0.35 (0.20-0.63)	0.18 (0.09-0.37)	0.33 (0.15-0.70)	0.33 (0.13-0.86)
Multiparae	0.56 (0.38-0.83)	0.69 (0.34-1.39)	0.22 (0.15-0.34)	0.15 (0.06-0.34)	0.22 (0.13-0.37)	0.27 (0.11-0.66)
Bacterial vaginosis	<i>P</i> =0.783	-	<i>P</i> = 0.404	-	P = 0.175	-
Negative	1.00		1.00		1.00	
Intermediate	0.92 (0.54-1.55)		1.05 (0.64-1.73)		1.05 (0.57-1.96)	
Positive	1.08 (0.73-1.62)		0.80 (0.53-1.19)		0.65 (0.39-1.10)	
Trichomoniasis	P = 0.435	-	<i>P</i> = 0.069	<i>P</i> = 0.722	<i>P</i> = 0.132	-
Negative	1.00		1.00	1.00	1.00	
Positive	1.18 (0.78-1.77)		1.45 (0.97-2.16)	0.50 (0.10-2.49)	1.49 (0.89-2.5)	

Table 6-9: Univariate and multivariate analyses of factors associated with PD, LBW and IUGR

Potential risk factor	Potential risk factor Preterm delivery Low birth weight		Intra-uterine growth retardation			
	Unadjusted OR	Adjusted OR	Unadjusted OR	Adjusted OR	Unadjusted OR	Adjusted OR
	-	N = 659	-	N = 659	-	N = 540
Chlamydia	P = 0.834	-	P = 0.818	-	P = 0.982	-
Negative	1.00		1.00		1.00	
Positive	1.09 (0.48-2.46)		1.08 (048-2.48)		1.01 (0.34-3.01)	
Gonorrhoea	P = 0.559	-	P = 0.659	-	P = 0.752	-
Negative	1.00		1.00		1.00	
Positive	1.34 (0.51-3.47)		0.78 (0.26-2.35)		0.79 (0.18-3.54)	
Confirmed Syphilis	<i>P</i> = 0.380	-	<i>P</i> = 0.292	-	<i>P</i> = 0.270	-
Negative	1.00		1.00		1.00	
Positive-treated	1.02 (0.46-2.29)		0.57 (0.22-1.49)		0.41 (0.09-1.74)	
Positive-untreated	1.53 (0.58-4.07)		1.63 (0.61-4.36)		0.62 (0.05-3.15)	
STIs/RTIs	<i>P</i> = 0.387	-	<i>P</i> = 0.023	<i>P</i> = 0.127	<i>P</i> = 0.186	-
No	1.00		1.00	1.00	1.00	
Yes-treated	1.39 (0.72-2.70)		0.42 (0.17-1.03)	0.39 (0.15-1.04)	0.38 (0.11-1.30)	
Yes- untreated	1.28 (0.87-1.88)		1.23 (0.84-1.79)	0.91 (0.56-1.47)	1.02 (0.63-1.66)	
HIV status	<i>P</i> = 0.193	-	P = 0.035	<i>P</i> = 0.029	P = 0.48	-
Negative	1.00		1.00	1.00	1.00	
Positive	1.40 (0.85-2.31)		1.71 (1.05-2.77)	1.91 (1.04-3.49)	1.27 (0.65-2.50)	
Peripheral Malaria at	<i>P</i> = 0.989	-	<i>P</i> = 0.160	-	P = 0.081	<i>P</i> = 0.208
first ANC visit						
Negative	1.00		1.00		1.00	1.00
Positive	1.00 (0.70-1.43)		1.30 (0.90-1.87)		1.53 (0.94-2.48)	1.38 (0.82-2.34)
Placental malaria	<i>P</i> = 0.007	<i>P</i> = 0.069	<i>P</i> = 0.008	<i>P</i> = 0.157	<i>P</i> = 0.732	-
Negative	1.00	1.00	1.00	1.00	1.00	
Positive	1.64 (1.14-2.37)	1.46 (0.97-2.20)	1.64 (1.14-2.36)	1.37 (0.90-2.07)	1.08 (0.67-1.75)	

Table 6-9: Univariate and multivariate analyses of factors associated with PD, LBW and IUGR continued

Potential risk factor Preterm delivery			Low birth weight		Intra-uterine growth retardation	
	Unadjusted OR	Adjusted OR	Unadjusted OR	Adjusted OR	Unadjusted OR	Adjusted OR
	-	N = 659	-	N = 659	-	N = 540
Anaemia [¥]	<i>P</i> = 0.051	<i>P</i> = 0.177	<i>P</i> = 0.610	-	<i>P</i> = 0.413	-
Normal	1.00	1.00	1.00		1.00	
Anaemic	1.60 (1.01-2.52)	1.44 (0.85-2.46)	1.13 (0.70-1.84)		1.75 (0.37-1.52)	
Hypertension	<i>P</i> = 0.097	<i>P</i> = 0.245	<i>P</i> = 0.573	-	<i>P</i> = 0.112	-
Normal	1.00	1.00	1.00		1.00	
Hypertensive	2.57 (0.87-7.54)	1.99 (0.63-6.20)	1.41 (0.44-4.58)		3.55 (0.83-15.2)	
Sex of baby	<i>P</i> = 0.035	<i>P</i> = 0.083	<i>P</i> = 0.349	-	<i>P</i> = 0.739	-
Female	1.00	1.00	1.00		1.00	
Male	0.68 (0.48-0.98)	0.69 (0.46-1.05)	0.84 (0.59-1.20)		0.92 (0.58-1.47)	
Antimalarial	P = 0.912	-	<i>P</i> = 0.004	<i>P</i> = 0.016	<i>P</i> = 0.233	-
treatment	1.00		1.00	1.00	1.00	
NO	1.00		1.00	1.00	1.00	
Yes	1.03 (0.60-1.76)		2.08 (1.29-3.34)	2.00 (1.15-3.47)	1.49 (0.79-2.82)	
Number of doses of	<i>P</i> < 0.001	<i>P</i> < 0.001	<i>P</i> = 0.260	-	<i>P</i> = 0.104	-
IPTp-SP						
0 or 1	1.00	1.00	1.00		1.00	
2 or more	0.34 (0.23-0.52)	0.35 (0.22-0.56)	0.77 (0.49-1.21)		1.87 (0.83-4.24)	
Bold type indicates sta	atistically significant ou	tcome				
[¥] Normal haemoglobin	level (Hb) was defined	as Hb ≥ 11grams per d	ecilitre and anaemia a	s Hb < 11 grams per de	ecilitre.	

Only 12 women experienced stillbirth and were not amenable to statistical analysis because of the small number.

Risk factor	Preterm delivery	P-value	Low birth weight	<i>P</i> -value	Intra-uterine growth retardation	P-value
	N = 715	< 0.001 [¥]	N = 713	< 0.001 [¥]	N = 557	< 0.001 [¥]
Parity	-	-				
Primiparae			1.00		1.00	
Para II			0.32 (0.18-0.58)	< 0.001	0.33 (0.16-0.71)	0.005
Multigravidae			0.22 (0.15-0.34)	< 0.001	0.23 (0.13-0.37)	< 0.001
Maternal Height	-	-				
≥ 1.5m			1.00		1.00	
< 1.5m			2.10 (1.42-3.10)	< 0.001	1.71 (1.04-2.83)	0.035
HIV status	-	-			-	-
Negative			1.00			
Positive			2.10 (1.23-3.55)	0.006		
Antimalarial	-	-			-	-
Other than IPTp-SP						
No			1.00			
Yes			2.01 (1.20-3.35)	0.008		
Number of doses of			-	-	-	-
IPTp-SP	1.00					
0 or 1	0.34 (0.23-0.52)	< 0.001				
2 or more						

Table 6- 10: Final logistic regression models of factors associated with PD, LBW and IUGR

^{*}*P* value for overall final model

-: Blank-only variables independent associated with each adverse outcome in multivariate analysis are shown

Bold type indicates statistically significant outcome

When marital and HIV status were put in a model with parity and height in the analysis of risk factors for LBW, there were not independently associated with LBW therefore they were excluded from the final model.

7 Discussion

7.1 Losses to follow-up

The loss to follow in this study was much higher (33%) than was expected (10%) and provided for in the sample size mainly due to interruptions in the provision of transport that were experienced as a result of delays in the disbursement of project funds during the follow-up to delivery period. Monthly efforts to supplement the budget during the waiting period were not sufficient because provision of transport was only possible for about a week in each month. The extended periods of lack of funds for fuel resulted in high losses to follow-up as it was impossible to provide transport to some delivering women who called members of the team during the period when funds were unavailable. Long distances to the health facilities and lack of transport have been cited as some of the reasons why women deliver at home in Zambia [174, 210]. During this period, team members who were still involved in the study went without pay and this may have negatively influenced their attitude towards work although they were assured of backdated payment when funds became available.

Furthermore, before the unavailability of funds became a problem, the main field vehicle that was used to transport participants to and from the hospital was involved in an accident and was destroyed by an act of arson which resulted in a period of transport shortage before alternative transport arrangements could be made.

In the current study refusals were low (< 1%) and only one participant withdrew from the study, however a high proportion of participants (30%) were lost to follow-up. A higher proportion of loss to follow-up than that observed in the current study was noted in another study conducted in 2010-2011 in Mansa District by Tan *et al.*, in the same province where the current study was carried out. The study was designed to examine the efficacy of IPTp-SP and presence of quintuple mutant in pregnant women [114]. Participants were followed up weekly over a period of 5 weeks. Among 109 pregnant women who were recruited, 53 (58%) completed the study, 34 (31%) had incomplete follow-up and 17 (16%) were lost after day 0 despite the fact that there were no refusals and withdrawals [114]. The study by Tan *et al.*, differs from the current study in that the number of follow-up points was higher (weekly) over a shorter period of time (5 weeks) while in the current study follow-up was scheduled at day 28 (for some participants) and at delivery for all participants.

Weekly follow-up over a period of 5 weeks may be difficult to endure in comparison to follow-up at day 28 and at delivery although follow-up after longer periods of time may be difficult to remember. However in the current study global positioning system mapping of homes locations and checking on some participants via phone calls helped to reduce incidence of loss to follow-up although contact via phone was only possible in the case of participants with active phone lines over the study period. Factors associated with high losses to follow-up in this region are not clear but understanding these factors might be helpful to future studies. It is highly likely that the proportion of women lost to follow-up to delivery was increased by myths and traditional beliefs discussed in Section 1.8 that influence birthing practices in Zambia and cause women to prefer delivery at home in seclusion or away from hospital and their home villages.

7.2 Prevalence of malaria, STI/RTI and their coinfection

7.2.1 The malaria burden

This study provides a robust estimate of the prevalence of peripheral malaria infection among pregnant women of Nchelenge District. A high burden of *P. falciparum* malaria diagnosed by PCR was observed in this population. The prevalence of malaria detected by microscopy was much lower than that detected by PCR suggesting that a good proportion of this study population had sub-microscopic infections (smear negative but PCR positive). Sub-microscopic infections have been associated with placental malaria and decreased maternal Hb [211]. Although no association was found between sub-microscopic infections and adverse maternal and foetal outcomes, the importance of these infections in pregnancy cannot be ignored in malaria policy formulation because of their role in maternal morbidity and malaria transmission [211].

Seven samples that were positive by microscopy in this study were found negative by PCR. Because PCR diagnostic method is more precise than standard microscopy [212], we would expect that a malaria sample found positive by microscopy would also be positive by PCR. The observation above could be attributed to a number of factors including malaria infection due to other *Plasmodium* species and insufficient DNA extraction. A negative PCR test result may be an outcome if the amount of

blood collected on filter paper was less than 25µl, coupled with low parasitaemia. Negative PCR results for microscopic positive samples have been noted in other studies [98, 99].

The prevalence of malaria infection by microscopy was very similar to the prevalence of parasitaemia observed in children less than five-years-old in 2012 (32.1%, 114/356) in the MIS in Luapula Province [189]. Another study conducted in the same area among children less than 10 years-of-age reported a similar parasite prevalence, 30.2% (236/782) by microscopy [188]. The prevalence of placental malaria in the current study, 40.9% (95% CI, 37.3-44.6), is comparable to what was found in another study, 37% (95% CI: 32.5-41.5), conducted within Luapula province in Mansa District [90].

Analysis of monthly-reported district-level malaria cases among pregnant women showed that MiP in Zambia decreased between 2010 and 2013, although persistent hot spots were reported in the southeast and northeast of the country [213].

The prevalence of malaria among 375 pregnant women in the study conducted in Nchelenge District between February and April 2013 by Siame et al. was much lower than what was observed in the current study, 15.0% by microscopy and 22.0% by PCR [115]. The observed difference in malaria prevalence in the two studies could be attributed to a number of factors. Firstly, in the current study women attending the Nchelenge and Kashikishi health centres were recruited, whereas in the study by Siame *et al.* a third health centre, Kabuta, was included in addition to these two. Secondly, in the study by Siame et al. pregnant women who sought ANC services were included, regardless of visit number; in this study, only first ANC attendees who had not taken antimalarial treatment in the previous month before recruitment were enrolled. Some of the participants who were included in the study by Siame *et* al. had received IPTp-SP during their earlier ANC visit(s) as well as antimalarial treatment before enrolment, and it is unclear what time period may have elapsed between treatment and subsequent screening. The prevalence of malaria is likely lower in a group of women exposed to any IPTp-SP because even one dose of IPTp-SP has been shown to protect against maternal parasitaemia [154]. Furthermore, the last IRS exercise before the current study was conducted in November 2012 and no IRS was conducted in the community by the district health office between October

2012 and April 2014 [186]. The study by Siame *et al.*, was conducted in April 2013 a period during which the insecticide residue in the houses in the community would be expected to be more effective against mosquitoes in comparison to recruitment period of the current study which commenced about 11 months after the IRS exercise (November 2013 to April 2014).

The coverage of bed nets in this study was lower than that observed in the study by Siame *et al.*, (74% versus 49.2%). In the study by Siame *et al.*, bed net ownership among pregnant women was used to characterise the percentage of households with at least one bed net while in the current study bed net ownership describes the percentage of pregnant women who slept in a protected space. It is therefore expected that ownership of at least one bed net in a household would be higher than that observed when bed net ownership is limited to the participant's sleeping spaces. No association between malaria infection and bed net ownership was observed in the study by Siame *et al.*, just like in the current study.

There was a difference in malaria infection and parasite densities among HIVuninfected and HIV-infected women, a finding similar to other studies [214-220]. The proportion of malaria infection detected by microscopy and by PCR in this study was consistently higher in HIV-infected than HIV-uninfected women. HIV-infected women had a 46% increased risk of malaria infection detected by PCR at first ANC visit (P = 0.045). However, the association between HIV and malaria infection was not statistically significant.

Previous studies have reported stronger association between HIV and peripheral malaria detected by microscopy in pregnant women. A hospital based study in Zimbabwe in 2000-2001 reported a high odds of malaria parasitaemia in HIV-infected women compared to HIV-uninfected women (OR 3.96; 95% CI, 2.42-6.46) [219]; a study in Rwanda conducted between 1992 and 1993 reported that the risk of parasitaemia was moderately higher in HIV-infected compared to HIV-uninfected women (RR 1.4; 95% CI, 1.1-1.6, P = 0.016) [216]; a similar finding was reported from Kenya in a study conducted between 1996 and 1999 among HIV-infected women in the third trimester (RR 1.70; 95% CI, 1.52-1.90) and at delivery (RR 1.56; 95% CI, 1.34-1.81) [220]. The prevalence of malaria parasitaemia among HIV-infected

women at first ANC visit was higher than among HIV-uninfected women in a study in Malawi conducted between 1987 and 1990 (56.6% versus 43.6%, *P* < 0.001) [218]. The use of antiretroviral therapy was not stated in all the studies above and, considering the years in which these studies were conducted, it can be assumed that coverage of antiretroviral therapy among pregnant women should have been very low if not non-existent. In the current study, however, the coverage of antiretroviral therapy at first ANC visit was 42.7% among all HIV-positive women, newly-tested and known HIV-positive cases. Some of the women who were screened in the current study were receiving cotrimoxazole prophylaxis although coverage data of cotrimoxazole at recruitment was not collected. However, data from delivery showed that 37% of HIV-infected study participants had received cotrimoxazole prophylaxis during pregnancy. Thus, the relatively a weaker association between HIV-infected status and malaria parasitaemia observed in the current study compared to the previously reported studies could be due to the fact that most HIV-infected women were on ARV therapy and/or cotrimoxazole prophylaxis.

There was no difference in the distribution of parasite counts in HIV-infected women on antiretroviral therapy and women who were not on treatment. This could be because HIV in women who were not on antiretroviral therapy had not progressed to a state of highly compromised immunity since some women tested positive for the first time. Furthermore, eligibility to receive antiretroviral drugs in this community was linked to CD4 count. Therefore, if women known to be HIV-positive were not on therapy, it was likely that their CD4 count was above 500 cells per cubic millimetre of blood.

7.2.2 Prevalence of curable STIs/RTIs

The combined prevalence of STI/RTIs (64.6%) was higher than that of malaria as was suggested in the review by Chico *et al.*, [6]. Estimates of individual STIs/RTIs in the current study fall within the 95% upper and lower limits of pooled estimates for East and Southern Africa reported in the study by Chico *et al.*, except in the case of syphilis which was estimated at syphilis 4.5% (95% CI, 3.9-5.1). Although the prevalence of syphilis (7.1%) obtained in this study is higher than the estimated 95% CI upper limit for East and Southern Africa (3.9-5.1), it falls within the range of

estimates (lowest to highest prevalence, 0-14.6) obtained from studies conducted in Zambia and neighbouring countries.

There are a few published studies on STIs/RTI in Zambia. Two papers were published from the HIV Preventions Trials Network (HPTN) 024 study, one reported on syphilis and the other was on the rest curable STI/RTI [135, 140]. The HPTN-024 study (conducted between 2001 and 2003) showed a high prevalence of individual STIs/RTIs in HIV-infected pregnant women (N = 642) of Lusaka. The prevalence estimates of the same STIs/RTIs infections among HIV-infected pregnant women (n = 143) of Nchelenge district in this study were higher than those found in the HPTN-024 study. Comparison of the prevalence of infection in pregnant HIV-infected women of Lusaka in the HPTN-024 study versus what was observed in HIV-infected pregnant women in the current study are as follows; Chlamydia, 6.1% versus 7.0%; gonorrhoea, 1.6% versus 6.3%; trichomoniasis 20.9% versus 23.8%; syphilis, 13.0% versus 15.8% and BV, 40.0% versus 66.4%. The diagnostic tools that were used in the HPTN-024 study were Gram stain microscopy using the Nugent score, wet mount, an enzyme immunosorbent assay kit, culture (then gram stain appearance and positive oxidase reaction) and RPR (as well as *Treponema pallidum* particle agglutination or TPHA confirmation) for diagnosis of BV, trichomoniasis, Chlamydia, gonorrhoea and syphilis respectively. The methods that were used for BV and syphilis diagnosis in the current study were the same as in the HTPN-024 study and the other infections (Chlamydia, gonorrhoea and trichomoniasis) were detected using PCR. It is likely that the prevalence of STIs/RTIs was higher in the current study population than in the HTPN-024 study population but the differences can partially be attributed to the diagnostic tool (PCR) with higher sensitivity that was used for the detection of some of the infections in the current study.

The fact that high levels of asymptomatic STIs/RTIs exist in this population is not surprising because literature shows that high proportions of STIs/RTIs in women are asymptomatic [166, 221]. The number of women who complained of at least one symptom associated with STIs/RTIs among women with at least one curable STI/RTI was low (14.0%). The proportion remained unchanged when only curable STIs (excluding BV) were considered. Among 374 women who tested positive for at least one STI, only 14.2% (n = 53) indicated that they were experiencing at least one STI

symptom at recruitment. This implies that a large proportion of (86.8%) of curable STI-infected women were asymptomatic in agreement with what has been observed from literature. High proportions of gonococcal (80%) and Chlamydial (70-75%) infections have been shown to be asymptomatic [164] and approximately 60-70% of asymptomatic *T. vaginalis* infection and BV have been reported [165, 166].

The high burden of STI/RTI that are asymptomatic in this study population is a clear demonstration of missed treatment opportunities. Although the current ANC package for the management of STI/RTIs is partially effective, it is clear that a huge burden of STI/RTI remains untreated. In the interest of reducing STI/RTI morbidity, there is need to devise alternative strategies for the management of STIs/RTIs in pregnant women.

Provision of point of care tests for STI/RTIs would be beneficial for the reduction of STI/RTI morbidity in this setting. This is crucial especially when the role of STI/RTI in HIV transmission is considered. Different STIs/RTIs, BV [222-226], trichomoniasis [227-232], Chlamydia [233, 234], gonorrhoea [233, 235] and syphilis [233, 235, 236] have been individually identified as risk factors for HIV acquisition and/or transmission. Furthermore, studies have shown that treatment of Chlamydia, gonorrhoea and trichomoniasis reduced genital shedding of HIV in women [237, 238]. In this context screening and treatment of STI/RTI would reduce HIV transmission. In the current study higher proportions of syphilis, Chlamydia, gonorrhoea and BV were found among HIV-infected women than HIV-uninfected women although this was only statistically significant in the case of BV, gonorrhoea and syphilis. A statistically significant higher proportion of composite STI/RTI was also observed among HIV-infected women. In this context good diagnostics and appropriate treatment of STIs/RTIs have the potential to impact the incidence of HIV. The fact that the prevalence of STIs/RTIs was high in this resource poor setting supports calls for the development of a vaccine against five common STI pathogens (herpes simplex virus, C. trachomatis, N. gonorrhoea, T. vaginalis and T. pallidum) [239].

7.2.3 Prevalence of malaria and STI/RTI coinfection

This is the first study to report malaria and curable STI/RTI coinfection in pregnancy. The estimated prevalence of malaria and STI/RTI coinfection in this study was considerable, irrespective of whether malaria was diagnosed by PCR or microscopy, 38.7% and 21.4% respectively. Before this the only report of coinfection was from a sub-analysis of results from a study in Tanzania which showed that 48.3% of RPRpositive women had placental malaria [6, 36].

The prevalence of STIs/RTIs and their coinfection with malaria represent an important finding in the context of the ongoing search for alternative drugs to replace SP for use in IPTp necessitated by the emergence of SP resistant parasites in some settings. The study provides evidence in support of antimalarial and antibacterial drug combinations that may offer the added benefit of reducing the burden of curable STIs/RTIs in pregnancy, especially in poor resource settings such as this where STIs/RTIs are highly prevalent and routine screening may not be sustainable.

7.3 Risk factors for malaria, curable STIs/RTIs and their coinfection

7.3.1 Risk factors for malaria infection

In the univariate analysis, the risk of infection was strongly associated with gravidity and age, but this was not observed in the multivariate analysis. Studies have shown that in malaria-endemic areas, the prevalence of malaria, both clinical and asymptomatic, is highest in young women and primigravidae and paucigravidae (primigravidae and secundigravidae combined) [47, 240, 241]. This is due to acquisition of semi-immunity that is gravidity-dependent such that malaria infection tends to be less prevalent and less severe among multigravidae [42, 47, 240]. However, HIV infection has been shown to impair the ability of multigravidae to manage malaria infection [214, 220]. The fact that HIV prevalence was highest among multigravidae in this study population is a plausible explanation for the fact that there was no association between parity and malaria infection in multivariate analysis. The low coverage of IRS and bed net use in an area of intense transmission may partially explain the high prevalence of malaria infection. Despite the progress that has been made in scaling-up malaria control interventions, a high burden of malaria still remains in Nchelenge District [108, 242]. Several reasons may underlie this observation including population movements from high endemic areas (internally and across borders), increasing parasite resistance to insecticides [188, 242], as well as homes having been built close to water sources that serve as breeding grounds for *Anopheles* mosquitoes.

The risk of malaria infection was higher among women at Nchelenge health centre compared to Kashikishi health centre. Since there was no difference in sociodemographic characteristics and HIV infection at the two sites, this difference in the risk of malaria infection may be attributable to the intensity of malaria transmission within the catchment areas of Kashikishi and Nchelenge health centres.

Apart from providing health services to the people living close to the Lake Mweru, Nchelenge health centre also provides health services to people in villages that are close to a rubber plantation, the Kenani River, and other in-land water bodies in the Robert Mutepuka village area which provide more breeding sites for mosquitoes.

Malaria transmission intensity is known to decrease during the winter months, April to August, in Nchelenge District [18]. Thus, the estimated prevalence of malaria in pregnancy from this study conducted during the hot rainy season from November to April is probably higher than the prevalence at other times of the year.

Zambia has different malaria transmission zones ranging from very low, to lowmoderate and moderate-high [109]. Thus, the prevalence of malaria estimated in this study can only be cautiously extended to areas with similar transmission patterns.

7.3.2 Risk factors for curable STIs/RTIs infection

In the current study infection with any STI was associated with BV but not with HIV. Several studies have associated STIs with HIV [230, 235, 243]. Trichomoniasis[223], Chlamydia [223], gonorrhoea [223, 235, 244] and syphilis [156, 244] have been individually associated with HIV infection. In contrast to this, other studies have

found no such association between trichomoniasis [245], or Chlamydia [246], or gonorrhoea [246] or syphilis [226, 246] or composite STIs [247] and HIV. The lack of association between STIs and HIV in the current study can be partially explained by the fact that infection with any STI was more common in younger women (\leq 25 years) while HIV was concentrated in women aged above 25 years.

Bacterial vaginosis was associated with STIs and HIV infection in the current study. The association of BV with Chlamydia [248, 249], gonorrhoea [248, 249], trichomoniasis [250, 251] and HIV [222, 223, 252, 253] has been observed in other studies. Furthermore, HIV-infected women may have more severe or persistent BV [254, 255].

The number of life-time sexual partners, younger age at sexual debut and parity have been associated with STIs [130, 181, 256], however, these associations were not observed in the current study. Factors that relate to sexual behaviour were also not associated with STIs. This lack of association can be explained by the fact that in this study the number of sexual partners that was queried was that of a life time count rather than for a limited period of time (e.g. past one year or during pregnancy).

7.3.3 Risk factors for malaria and STI/RTI coinfection

Malaria and STIs/RTIs have both been associated with HIV infection [214-217, 219, 230, 235, 243, 257]. Consequently, it is not surprising that HIV was a risk factor for malaria and STI/RTI coinfection in this study population. Although the sample size was not large enough to conclude that the risk of malaria infection among HIV-infected women was statistically significant, HIV infection was associated with malaria and STI/RTI coinfection.

The fact that HIV was strongly associated with malaria and STI/RTI coinfection suggests that management and prevention of these infections in pregnant HIV-infected women should be prioritised in this setting through provision of point of care tests and aggressive treatment of infected women. This is particular true for syphilis as RPR results were not available on the same day as the ANC visit. The proportion of women testing positive for syphilis who receive treatment could be increased by the use of point of care tests rather than laboratory-based RPR tests as

demonstrated in several countries including Zambia [258]. The facilities relied on pregnant women returning to the facility for treatment with their partners, prompted by a note with their test results. However, among 729 delivering women only 52 of the 81 women who were RPR reactive at enrolment actually returned to the facility for care. It is likely that a higher proportion of these women could have received timely treatment if point of care tests for syphilis, now recommended by the WHO [259] for use in the ANC package, had been used. Overall, clinicians and staff involved in the provision of ANC services in this setting need to be proactive in the management of STIs/RTIs, especially in HIV-infected women. Furthermore, community education on the ways that infection with malaria, HIV and STIs/RTIs can be prevented cannot be over emphasised.

7.4 Prevalence of point mutations associated with parasite resistance to SP

Point mutations associated with resistance to SP in this study were highly prevalent. The prevalence of both the quintuple and the sextuple mutants was moderate but higher than what has been recorded in earlier studies both in Luapula Province and around Zambia [114]. In a study conducted in 2006 in the southern part of Zambia, the prevalence of the quintuple mutant was 6.5% in the general population. In another study conducted by Siame et al., in 2013 in Nchelenge District the prevalence of the quintuple and sextuple mutant among pregnant women was 17.0% and 3.0% respectively [115]. Another study, which documented the sextuple mutant for the first time in Zambia, conducted among pregnant women at two health centres in Mansa District (which is also located in Luapula province) between January 2010 and May 2011 reported the prevalence of the quintuple and sextuple mutant at 63.0% and 2.0%, respectively [114]. In the current study the prevalence of the quintuple and sextuple mutant was 68.8% and 9.4%, respectively. This suggests that there has been an increase in the prevalence of the 581G mutation in this population which has the potential to compromise the efficacy of IPTp-SP in this population. A systematic review and meta-analysis of IPTp-SP found that > 2 doses of IPTp-SP no longer protected against the incidence of LBW among multigravidae where the prevalence of 581G was > 10.1% among children [260]. It is therefore

important to monitor the prevalence of SP-resistance markers and the *in vivo* efficacy in Zambia especially in high transmission areas.

7.5 Therapeutic and prophylactic failure of sulphadoxine-pyrimethamine

There was 14.6% treatment failure among pregnant women of Nchelenge district 28 days post-IPTp-SP in this area with a moderate prevalence of the highly resistant quintuple mutant (68.8%) and presence of the sextuple mutant (9.4%). Sulphadoxine-pyrimethamine retained some efficacy in parasite clearance among pregnant women with asymptomatic malaria in this study population. Similar observations have been has been made elsewhere i.e. SP retaining some efficacy despite a moderate to high prevalence of the quintuple mutant and presence of the sextuple mutant [114, 261].

The risk of having parasitaemia at both day 0 and day 28 was higher among primigravidae than secundigravidae and multigravidae. This can be explained by the fact that malaria prevalence is highest in younger women and primigravidae [47, 240, 241].

Prophylactic failure could not be estimated accurately because it is impossible to know the number of women who tested negative by microscopy at day 0 and were later exposed to the malaria parasite after IPTp-SP administration. Some women may have been infected and SP cleared the parasitaemia and others may never have been infected. Nonetheless, estimating prophylactic failure based on the number of malaria positive women among those who tested negative at day 0 is a good proxy for the estimate of prophylactic failure.

Less than half of the cases (47%) with positive malaria outcome on both day 0 and day 28 were due to recrudescence by MSP2 genotyping. Only MSP2 genotyping was carried out to distinguish recrudescence from reinfection in the current study. Although genotyping for the three genes (MSP1, MSP2 and GLURP) was not done in this study, the results from MSP2 genotyping are reliable since MSP2 provides a more accurate measure of treatment outcomes in comparison to MSP1 and GLURP [262].

7.6 Characteristics of delivering women, Prevalence of adverse birth outcomes and risk factors for adverse birth outcomes

7.6.1 Characteristics of women according to follow-up

The distribution of characteristics between women who were followed up to delivery and those who were lost were similar except in the case of gravidity. The proportion of primigravidae was higher among women followed to delivery than among those lost to follow-up. This is contrary to what has been observed in other studies where women lost to follow-up to delivery were more likely to be primigravidae [36, 263, 264]. The observation of a higher proportion of primigravidae among women followed up than among those who were lost to follow-up in the current study can partially be explained by the fact that younger Zambian women prefer to deliver at a health facility due to lack of birthing experience and for fear of labour complications [175].

7.6.2 Prevalence of adverse birth outcomes

There was considerable prevalence of ABO (35.0%) in this study population. The prevalence of stillbirth and LBW in the current study was 1.7% (95% CI, 1.0-2.9) and 21.9% (95% CI, 19.2-25.3), respectively. This is similar to what was recorded in 2013 at St Paul's Hospital, the only hospital in the district where prevalence estimates of stillbirth and LBW were 2.4% (95% CI, 1.8-3.0%) and 21.5% (19.9-25.3%), respectively. The routine data system at the hospital does not contain estimates of PD nor IUGR. The rate of stillbirths observed in this study (1.7%) was also comparable to what was observed in the ZEPRS (24 per 1000 births). Differences in individual ABO between results from this study and others conducted in Zambia were noted in the prevalence of LBW and PD. In this study the prevalence of LBW was 21.9% which was much higher than what was reported in the 2013-14 ZDHS and the ZEPRS, both of which reported 10% prevalence. And the prevalence of PD recorded in the ZEPRS was 54% higher than what was recorded in this study (34% versus 22.1%). The estimate for the prevalence of LBW reported in the ZDHS survey was obtained from Lusaka, Copperbelt and Southern Provinces while the ZEPRS was conducted in 25 health facilities across Lusaka. The results seem to suggest

differences in the prevalence of these ABO in different regions of Zambia. Understanding factors that are associated with ABO in different regions is required in order to target the most affected populations with efforts aimed at reducing the prevalence of ABO.

7.6.3 Risk factors for adverse birth outcome

There was no difference in the proportions of women who delivered in the different places (hospital, health centre or home) between the two recruitment sites, but a higher proportion of women recruited at Nchelenge were assisted by TBAs than women at recruited at Kashikishi health centre. The simple explanation for this is that at the time of the study Nchelenge Health Centre did not offer maternity services and the alternative place of delivery besides St. Paul's Missionary Hospital for women who attend ANC at Nchelenge was Kabwali Health Centre, which was the other centre where some of the study participants delivered. In comparison to Kashikishi Health Centre, Kabwali was under staffed (six midwives versus one) and relied on TBAs to a greater extent to perform some of the deliveries at the health centre. In the current study adverse ABO was associated with short stature and primiparity. Both short stature [36, 265, 266] and primiparity [36, 265-268] have been associated with ABO in other studies.

In the current study peripheral malaria was diagnosed at first ANC attendance and placental malaria was diagnosed at delivery. No association between malaria (both peripheral and placental), STIs/RTIs or their coinfection and ABO was observed in the current study. Studies have associated malaria [36, 49, 50, 55-58] and individual STIs/RTIs of interest in the current study i.e BV [12, 23, 24, 37-41, 65, 66], trichomoniasis [19-21, 33], Chlamydia [11, 12, 28-31, 33-35, 62-64], gonorrhoea[17, 18, 32, 33] and syphilis[9, 10, 17, 18, 26, 69] with ABO. Interestingly, women who indicated that they had received antimalarial treatment besides IPTp-SP at the time of delivery were at a higher risk of experiencing LBW. It is assumed that women who were treated with antimalarial drugs experienced clinical malaria at some point during pregnancy which may have had an impact on the weight of the babies.

In this study no association was found between composite STIs/RTIs (BV, trichomoniasis, Chlamydia, gonorrhoea and syphilis) and composite ABO (PD, LBW, IUGR and stillbirth). Similarly, no association was found between individual STIs/RTIs or malaria and PD or LBW or IUGR. The similarity in these finding underscores the observation that there was no association between STIs/RTIs or malaria or their coinfection and ABO in this study population.

Although the prevalence of ABO was high in the study population (35%), the difference in the proportions of ABO between unexposed and exposed (infected with at least one curable STI/RTI and/or malaria diagnosed at first ANC) women was minimal (31.8% versus 36.2%).; a larger sample size of 4,469 pregnant women would be required to detect a statistically significant difference between unexposed and exposed women under the prevailing conditions. This study was underpowered for the detection of such a small difference in the outcome between infected and uninfected women.

Various ANC activities contributed to the reduction of ABO attributable to curable STIs/RTIs and malaria in pregnancy. The lack of association between infections and ABO could partially be attributed to the interventions in the ANC package including IPTp-SP, iron and folic acid supplementation and syphilis treatment. Administration of IPTp-SP reduces the adverse effects of malaria in pregnancy including third-trimester maternal anaemia, placental parasitaemia and the incidence of LBW [150, 153, 217, 261]. In the current study IPTp-SP coverage of at least one dose was high with 99% among women who were followed to delivery. The coverage for at least one dose of IPTp-SP was higher than the recorded provincial coverage of 89.7%. The increase in the coverage of IPTp-SP among participants can be attributed to the fact that one of the objectives of the study was to assess the *in vivo* efficacy of SP and, thus, the administration of SP at the first and second visit as well as attendance of ANC may have influenced the IPTp-SP coverage.

It was noted that 36.4% of the 88 HIV-infected women who took IPTp-SP also took cotrimoxazole during the pregnancy duration. It is unclear whether these women took both cotrimoxazole and SP at the same point in time. However, clinicians in this setting are aware of toxicity concerns of administration of both drugs at the same time.

Treatment of BV [23, 68, 73-76], trichomoniasis [75], Chlamydia [70, 71], syphilis [60, 61] and composite STIs/RTIs[67] has been shown to improve birth outcome. Conflicting results have been observed in the case of treatment of BV [23, 73, 74, 76]. Studies in the US showed that treatment of BV reduced PD [23, 73, 74] and similar results were found in a study in the UK [76]. However, no benefit of treating asymptomatic BV was observed in a review [77]. Results from two studies looking at treatment of trichomoniasis and BV in pregnancy showed no benefit in terms of reducing ABO [75, 78]. Others found increased risk of ABO in treated women compared to untreated women [79, 80]. In contrast to this, a sub-analysis of data from another study showed that women randomised to the antibiotic arm were not more likely to deliver a preterm infant or to deliver an infant with a lower mean birth weight [84].

The coverage of syphilis treatment among RPR reactive women (64%) was quite low. All the participants were followed up within three days of screening and given a slip to return to the health centre together with their sexual partner for treatment. Counsellors were involved in the note delivery process to ensure that those who had doubts were given adequate information on the benefits of seeking treatment. However, some of the participants did not return for treatment after being given the first note or at day 28 follow-up or after a second note was delivered to them/their homes inviting them to return to the health centre for treatment. The success of the treatment exercise was partially dependent on the cooperation of individual participants. Unfortunately, the reasons for choosing to stay away are unknown but establishing this would be useful in future. The proportion of women testing positive for syphilis who receive treatment could be increased by the use of point of care tests rather than laboratory-based RPR tests as demonstrated in several countries including Zambia [258], therefore the Zambian Ministry of Health should consider amending the practise of syphilis screening during ANC.

In this study the prevalence of STIs/RTIs was high and treatment of participants based on syndromic management was low due the fact that a high proportion of infections (86%) were asymptomatic. Furthermore, only a small proportion (12.8%) of the 467 who tested positive for at least one STI/RTI and 18.6% of 248 STI-infected women received antibacterial treatment during the pregnancy duration. Some of the

women who were treated based on syndromic management received antifungal drugs. It is important to note that the assessment of participants who presented with one or more symptoms indicative of an STI/RTI for the decision on treatment was not part of this study. In this study all the participants were asked via the questionnaire if they were experiencing any signs or symptoms associated with STIs/RTIs at the time of recruitment. Those who responded positively were directed to their regular clinician for further assessment and decisions on treatment. The drugs that were given were recorded on the ANC cards and for the purpose of the study drugs taken throughout the pregnancy duration were recorded at delivery. As a result of this arrangement, it is difficult to comment on the performance of the syndromic management since the study did not assess how this was done or the competence of the clinicians who were attending to the participants.

The administration of antifungal drugs to women who tested positive for at least one STI/RTI suggests that some of the women may have had fungal infections which may have been much more obvious than the STI/RTI in terms of signs and/or symptoms.

It was surprising that infection with at least one STI/RTI, which was highly prevalent in this study population, was not associated with ABO despite a low proportion of infected women being treated. One possible explanation is the fact that the uptake of SP was high in the study population. Studies have suggested that SP could offer additional protection against infections other than malaria resulting in the protective effect against ABO which becomes more apparent in areas where parasites have lost sensitivity [260, 269].

Aside from being underpowered, this study differs from other studies that have associated STIs/RTIs with ABO in several ways. Firstly, some of the prior studies were conducted in the year 1990 or earlier [11-13, 20, 24]. Under the circumstances, one might expect that STIs/RTIs in pregnancy were not managed as well at that time compared to more recent years as interventions and implementation have evolved. Therefore, differences in proportions of ABO among infected and uninfected women may have been greater historically and, consequently, more easily be detected. Secondly, among the prospective studies involving screening of women and follow-up to delivery [20, 24, 30, 37] or record based retrospective studies, much larger

groups of women/records ranging between 3540 and 354200 were enrolled/included in the studies. Thirdly, studies from both the developing and developed world used the case-control design [17, 19, 33, 35].

Syphilis was associated with ABO in Kenya at a time when screening and treatment was policy in many countries but was not implemented in poor settings [60]. In a study in Tanzania by Watson-Jones *et al.*, syphilis was strongly associated with stillbirth, PD and LBW [10]. Notable differences in relation to syphilis between the study by Watson-Jones *et al.* and the current study include; (i) the women in the study in Tanzania had not been previously screened and treated for syphilis while in the current study screening was done and about 64% of delivering participants who tested positive to RPR received treatment, (ii) although the prevalence of RPR-positives in the study by Watson-Jones *et al.*, was slightly lower than that observed in the current study (8% versus 10.9%), the proportions of high-titre active syphilis among RPR-seropositive women in the study by Watson-Jones *et al.*, was higher than that observed in the current study (52.9% versus 36.7%). High-titre active syphilis was shown to pose the greatest risk to the foetus in the same study in Tanzania [10].

Malaria was strongly associated with LBW, PD and IUGR but not stillbirth in a study conducted between 1997 and 2000 in Mwanza, Tanzania [36]. The Mwanza study was conducted before the WHO recommended IPTp-SP and therefore participants did not receive IPTp-SP unlike in the current study. Another study, conducted in November to December 2006 in Ghana, found an association between malaria and stillbirth [50]. The coverage of IPTp-SP among 746 delivering women was low in comparison to what was observed in this study; 47.3% of delivering women had taken two or more SP doses during the pregnancy duration, while in the current study 82.1% of participants had received two or more doses of SP. In another study in Sudan carried out between November 2006 and February 2008, malaria was associated with stillbirth, however, the coverage of IPTp-SP was not mentioned [49].

Although placental malaria diagnosed by histology at delivery was not associated with ABO, the proportions of PD and LBW were significantly higher among women with chronic placental malaria infection (pathology class 2) than in active, past and no infection groups. Furthermore, in comparison to the uninfected women, those with chronic placental malaria infection had the lower mean maternal Hb, a lower median gestational age and the mean birthweight of the babies was lower as observed previously [191].

Infection with HIV indirectly affects birth outcomes as it is a risk factor for maternal anaemia [270]. Infection with HIV was associated with LBW in the current study. No association was found between HIV and composite ABO or the other individual ABO (IUGR and PD). Other studies also conducted in sub-Saharan Africa have associated HIV with ABO [271, 272] in contrast to others [36]. In the current study 42.7% of participants were on ARV therapy at recruitment and the coverage increased to 61.7% by the time of delivery. Studies have associated highly active antiretroviral therapy with higher rates of ABO [273-276], others found no association [271, 277, 278] or recorded favourable pregnancy outcome [279].

Having taken two or more doses of IPTp-SP was protective against ABO in the current study demonstrating the effectiveness of this strategy in this setting with moderate prevalence of the quintuple mutant. Previous studies in areas of high prevalence of the quintuple mutant have shown declining protective effect of IPTp-SP or that it is harmful [2, 280, 281]. However, in areas of low and moderate levels of SP resistance, IPTp-SP has been shown to be effective in reducing the incidence of ABO [90, 282]. In the same province where the current study was conducted, in Mansa District, an area of moderate prevalence of the quintuple mutant [114], each dose of IPTp-SP contributed to a 46% and 37% decrease in the frequency of LBW among paucigravidae and multigravidae women presenting for delivery at a rural health clinic between December 2009 and December 2010 [90].

In the univariate analysis hypertension was strongly associated with ABO (OR 3.36, 95% CI 1.1-10). Blood pressure readings were taken at recruitment and before delivery and hypertension was defined as two high BP readings. It was not always possible to take BP readings before delivery due to some deliveries occurring at home or in cases where women made it to the hospital so close to delivery resulting in a large number of missing values. However, the risk factors for ABO that were identified when hypertension was included were the same as when hypertension

was excluded in the multivariate model. Urine testing was not conducted and therefore preeclampsia was not adjusted for in this study.

This study demonstrates the existence of high prevalence of malaria and curable STIs/RTIs as well as their coinfection in a sub-Saharan setting. In this setting two or more doses of IPTp-SP was associated with a decrease in the risk of ABO. Although IPTp-SP continues to offer protection against the effects of malaria in some settings, the emergence of parasites resistant to SP has undermined the efficacy of SP to protect against the malaria attributable fraction of LBW [157]. Given the existence of malaria and curable STIs/RTIs confection, the high levels of asymptomatic curable STIs/RTIs, the limited accuracy of syndromic management and emerging parasite resistance warrants trials to explore antibacterial and antimalarial combinations for the replacement IPTp-SP. A review of studies on the efficacy of azithromycin against curable STIs/RTIs concluded that azithromycin-based combination therapy therapies are among leading potential candidates to replace SP for IPTp owing to the effectiveness of azithromycin against T. pallidum, N. gonorrhoea and C. trachomatis [283]. In a study in Malawi, adding azithromycin to monthly SP regimen appeared to offer further benefit to foetal and neonatal growth [284]. Such a regimen is likely to benefit pregnancy outcomes as well as reduce STI/RTI morbidity.

7.7 Potential study biases, limitations, Conclusions and recommendations

7.7.1 Potential study biases and limitations

Non-attendance of ANC may be a potential source of bias if the women who stayed away from ANC had a different profile from those that attended ANC. However, it is unlikely the results would differ much because in this part of Zambia ANC coverage is about 95% and refusal rate among eligible women was less than 1%.

The general follow-up of study participants was lower than was expected and this is a source of concern when bias resulting from losses to follow-up is considered. However, women who were followed-up to delivery were generally similar to those who were lost to follow-up with respect to socio-demographic characteristics and exposure to infection and thus were representative of the study population. Although malaria transmission occurs all year-round, higher numbers of malaria cases are recorded in the rainy season, which was the season in which this study was conducted. Therefore, the estimated prevalence observed of malaria alone and malaria and STI/RTI coinfection may be higher than it would have been had the study been carried out at another time of the year.

Classification of participants as infected or uninfected was based on results for malaria diagnosed by PCR and all the STIs/RTIs at first ANC visit. Repeat reading of a proportion of slides for BV and RPR/TPHA testing were conducted for quality control. In the case of Chlamydia, gonorrhoea and trichomoniasis PCR was repeated using a different assay (Seeplex[®] STI Master Panel 1 by Seegene). The results obtained from repeat diagnosis/reading were virtually reproducible (> 95%). Therefore, it is expected that bias due to misclassification was minimal in this context. On the other hand, misclassification of some of the infected participants as uninfected was reduced by diagnosis of malaria and some curable STIs/RTIs by PCR methods which generally have higher sensitivities than traditional methods.

Diagnosis of STIs/RTIs was limited to the first ANC visit and women were classified as infected or uninfected based on results from the screening done at first ANC. This potentially may have resulted in the misclassification of individuals who were classified as uninfected but acquired an infection later in pregnancy and which may have resulted in the reduction of the association between infection and ABO towards the null. A second screening for STIs/RTIs later in pregnancy is recommended in future studies although this would be much more expensive and labour intensive.

Gestation age was determined using ultrasound which is a reliable method for establishing the length of pregnancy. One of the inclusion criteria in the current study was women with gestation \leq 32 weeks. The measurements of gestational age by ultrasound in the second part of pregnancy are less reliable than in the first trimester due to normal biological variation [285]. Under/over estimation of gestational age at recruitment would result in misclassification of participants in terms of outcomes that are dependent on gestational age i.e. PD and IUGR. Although the recruitment period would be substantially increased, setting a lower gestation

age as an exclusion criterion would minimise the level of misclassification in this respect and is therefore suggested for future studies.

Another study limitation is in the kind of information that was required for the study. Some of the questions that were posed to the women especially those on sexual behaviour required sharing of information that can be considered confidential and as a result the responses to these questions may not be reflective of the truth. Furthermore some of the responses to questions on socio-economic status, utilisation of malaria interventions, sexual behaviour etc. may have been influenced by social desirability.

The study was under powered to detect risk factors with ORs > 2 and was therefore limited in this respect.

7.7.2 Conclusions

The prevalence of *P. falciparum* malaria detected by PCR in pregnant women of Nchelenge District was high. HIV infection increased the risk of malaria infection, although the sample size was not large enough to conclude this risk was statistically significant. The high prevalence of malaria infection that was observed in this population suggests that past prevention efforts have had limited impact in pregnant women.

The combined prevalence of STIs/RTIs was higher than that of malaria and was concentrated in younger and HIV-infected women. The prevalence of malaria and STI/RTI coinfection was considerable in this study population and the risk of infection was higher in HIV-infected women.

Both quintuple and sextuple mutants were observed in this study population. Sulphadoxine-pyrimethamine retains partial efficacy in clearing parasites in pregnant women in this area with moderate prevalence of the highly resistant quintuple mutant.

The prevalence of ABO was considerable and was higher in shorter and primiparous women. However, there was no association found between infection with malaria, STI/RTI or both and ABO in this study. Two or more doses of IPTp-SP was protective against ABO, therefore IPTp-SP remains a viable option for Zambia in this context.

However, there is need to investigate alternative drugs that are effective against malaria and common STI/RTIs to replace SP for IPTp.

7.7.3 Recommendations

The findings from this study are not enough to influence change of policy; however they can be used to recommend better coverage of the current policy.

The prevalence of malaria and STIs/RTIs in this study group was generally high and ownership of bed nets and usage were also observed to be low. Malaria control policies should therefore focus on providing uninterrupted interventions such as robust vector control, distribution of bed nets, yearly IRS spraying, IPTp-SP provision and sensitising of the community about the importance of using available interventions which are vital to reducing the malaria burden in pregnancy. Health service provision administrators should ensure uninterrupted availability of test kits for syphilis and treatment drugs. Further sensitisation of community members on the effects of syphilis in pregnancy could improve the numbers of women and their partners who seek appropriate treatment for syphilis since effective treatment requires commitment from infected pregnant women. Provision of point of care tests for syphilis and prompt treatment of infected women and their partners should be considered in this setting especially among HIV positive women. Provision of point of care tests for syphilis has the potential to increase the proportion of women receiving treatment among those testing positive. Increased frequency of community education on the prevention of infection with STIs via safe sex practises and behaviour change such as the use of condoms and maintaining one sexual partner can help reduce the burden of STIs/RTIs. Dissemination of information to the community can be enhanced via other channels such as the community radio and in schools.

7.8 Future work

The results in the current thesis show that the prevalence of malaria and STI/RTI coinfection is considerable. Although this has only been demonstrated in one setting, the fact that malaria and STI/RTI have overlapping prevalence in sub-Saharan Africa
implies that a number of settings have cases of malaria and STI/RTI coinfection. Evidence suggesting that IPTp-SP may not be beneficial and that it may be harmful in areas where parasites express the 581G DPHS mutation has emerged in Muheza, Tanzania [2], but this has not observed elsewhere [286]. It has been reported that IPTp-SP was not associated with decreased odds of placental malaria or with increased mean maternal haemoglobin or mean birthweight. Furthermore IPTp-SP was associated with decreased cord haemoglobin level and increased foetal anaemia. Given the compromised efficacy of IPTp-SP [2, 13, 157, 280] coupled with and the limitations of syndromic management in the detection of STIs/RTIs, there is need for trials for antibiotic and antimalarial combinations for IPTp. Therefore, future work will involve evaluation of the effect of an antibiotic and antimalarial combination on birth outcomes in an environment with high prevalence of SP resistance markers in comparison to the current strategy of IPTp-SP.

A possible combination of an antimalarial and antibiotic is that of azithromycin and dihydroartemisinin-piperaquine. This is because azithromycin has been shown to efficacious against T. pallidum, C. trachomatis and N. gonorrhoea and azithromycin combination therapies are among the leading candidates to replace SP for IPTp and may offer the added benefit of reducing morbidity and the effects of STIs/RTIs on pregnancy outcome [283]. In the case of an antimalarial, research has shown that dihydroartemisinin-piperaquine could be a good replacement for SP as IPTp. Two trials have evaluated the use of dihydroartemisinin-piperaquine for the prevention of malaria in pregnancy. One was conducted in an area of western Kenya with high levels of resistance to SP. In this trial, HIV-uninfected pregnant women were randomly assigned to receive intermittent screening and treatment with dihydroartemisinin-piperaquine or IPTp with a median of three doses of either dihydroartemisinin-piperaquine or SP [269]. Intermittent screening and treatment with dihydroartemisinin-piperaquine was not found to be a suitable alternative to IPTp-SP. However, IPTp with dihydroartemisinin-piperaquine was associated with a lower prevalence of malaria infection at delivery (15 [3%] of 457 versus 47 [10%] of 459 of women; relative risk 0.32, 95% CI 0.18-0.56; *P* < 0.0001); a lower incidence of malaria infection (19.2 versus 54.4 events per 100 person-years; incidence rate ratio

0.28, 95% CI, 0.22-0.36; *P* < 0.0001) and clinical malaria during pregnancy (37.9 versus 6.1 events; 95% CI, 0.16, 0.08-0.33; P < 0.0001) than was for SP.

The second trial conducted in Tororo, Uganda, an area of high malaria-transmission and intensity, was a double-blind, randomised, three-group controlled trial SP, three-dose dihydroartemisinin-piperaquine, comparing and monthly dihydroartemisinin-piperaquine as IPT for malaria in pregnancy [287]. Participants were HIV-uninfected pregnant adolescents or women at least 16 years of age (primigravid or multigravid), who were between 12 and 20 weeks of gestation. In this trial the prevalence of histopathologically confirmed placental malaria was significantly higher in the SP group (50.0%) than in the three-dose dihydroartemisinin-piperaquine group (34.1%, P = 0.03), or the monthly dihydroartemisinin-piperaquine group (27.1%, P = 0.001). The prevalence of composite ABO was marginally lower in the monthly dihydroartemisininpiperaquine group (9.2%) than in the SP group (18.6%, P = 0.05) or the three-dose dihydroartemisinin–piperaquine group (21.3%, P = 0.02).

Adding an antibiotic to dihydroartemisinin–piperaquine is likely to be more beneficial than using this antimalarial alone. As earlier mentioned, studies have suggested that SP could offer additional protection against infections other than malaria resulting in the protective effect against ABO [260, 269]. In view of this, it is likely that if SP is withdrawn and replaced by an efficacious antimalarial without any antibacterial activities, it may not be as effective as IPTp-SP has been noted to be in reducing ABO.

Given that malaria and STI/RTI co-infections are very common and both infections are associated with ABOs, future interventions to reduce ABOs should be effective against both infections. As mentioned earlier a combination of azithromycin and an antimalarial could be considered. A clinical trial to assess the effect of SP alone, SP + azithromycin or dihydroartemisinin-piperaquine + azithromycin on ABOs is needed. In addition, as the most common RTI is BV a trial to evaluate the effects of adding metronidazole to SP or SP is also needed to inform the future guidelines to control adverse birth outcomes attributable to malaria and STI/RTIs.

7.9 Reflexivity

According to socially constructed knowledge claims, personal, cultural and historical experiences of the researcher influence the interpretation hence the need for the researcher to acknowledge this and account for it in their study [288-290]. Specifically, there is need for reflexivity from the researcher on how their socio-cultural and historical contexts shape their theories and research practises. Although this was not qualitative research, this section was included to give the reader some understanding of my background which shapes the way in which the research was conducted and my interpretation of the results. Reflexivity helps "researchers to recognise that their own background shapes their interpretation and 'position themselves' in research to acknowledge how their interpretation flows from their own personal, cultural and historical background" [291].

This section identifies the role of my personal experiences in the study which may be in the design, conduct, analysis, interpretation of results and writing of the research results [292]. I have written this section in first person to show my perspectives, experiences and roles in certain aspects of the research project.

I know that my personal experience influenced the design of this study in one way or another. A good example is in how my decision on the study design was influenced by my personal experience as a mother. I was faced with deciding whether to conduct a case-control study in which samples were to be collected from delivering women or find an alternative design. I knew that obtaining cervico-vaginal samples from delivering women might be inappropriate although I had come across a study in which vaginal and cervical samples were collected within 24 hours of delivery and at 7 days postpartum for identification of organisms including *Chlamydia trachomatis* and *Neisseria gonorrhoeae* by the culture method [17]. This was because I strongly felt that this would be too much to ask from a participant during such an intimate time and that I would personally find it difficult to agree to such a thing having gone through the birthing process. I also thought about collection of cervico-vaginal samples a week after delivery which I dismissed for the same reason. Due to my experience in molecular diagnostic methods (which I acquired during my

Master of Science training in Medical Parasitology), I was aware of possible challenges I would have faced if I had opted to isolate genetic material from cervicovaginal swabs stained with blood. As a result of this background I settled for a cohort study in which samples were collected at first antenatal care attendance and participants were followed up to delivery. My past training and experience also made other aspects of the study manageable. For instance my undergraduate training in Microbiology made the processing, reading of slides for diagnosis of BV and interpretation easy for me.

Having grown up in both the rural and urban settings of Zambia, I understood the importance of recognising community leaders which was cardinal to the success of the study. As a result of this I made courtesy calls on the chief and the village head men in the areas where the two recruitment sites were located. I also recognised the authority of the administrators at the district health office and the district commissioner who appreciated this and were supportive to the project throughout the period that I worked in the area.

Furthermore, I decided to engage the local people as much as possible in the research activities. The regular ANC staff, midwives and TBAs were trained and involved in the project because I understood that the local people were likely to co-operate with a research team that included people within their community based on information I had gathered prior to the commencement of the project. I was also aware of the fact that involving members of community who were familiar with participants might result in participants withholding certain information during one-on-one interviews while completing the study questionnaire. To deal with this, research team members administering questionnaires were instructed to declare personal knowledge of or familiarity with the participants and pass on such participants to another member of the team.

Some of the community members thought I was linked to the Tropical Disease Research Centre which has been conducting research in the area for several years and has a good reputation in the community. I believe this also influenced the willingness of women to take part in the study to some extent.

My gender and the fact that I had a few months old baby at the time I started my field research may also have influenced the attitude of expecting mothers towards me. Women seemed comfortable talking to me and seeking clarification on certain aspects of the study. Based on some of their questions and comments, it was clear that some of the community members thought that I am a powerful person or someone connected to politically powerful people. This may have influenced some of their responses to questions bordering on economic status or bed net ownership and usage to some extent as some may have wanted to portray a slightly different picture with the hope of getting help or may have wanted to be desirable.

The field work also had a personal impact on me as I leant to relate to and engage the community. Visiting the homes of participants was made much easier by the warmth of community members who were generally welcoming. Co-ordinating the day to day running of project activities also helped me to build my interpersonal skills. A good example of a time when I had to closely deal with the community is when the project vehicle was involved in an accident involving a drunken cyclist and two children. Our driver tried to avoid a drunken cyclist who hit his bicycle on the front passenger's door by swerving to the opposite side of the road into an area with tall grass along the road. Sadly, two little girls were playing in the grass and both were hit. One sustained injuries and the other died instantly. The people who witnessed the accident understood what happened. After about an hour while waiting for the police some youths who were not at the scene at the time of the accidents got excited and set the vehicle ablaze. This brought the project to a temporal halt. As the principle investigator I had to deal with the police, the insurance company, the community leaders, the district commissioner's office, the family of the deceased and the injured girl and cyclist. However, in dealing with all these people, I generally received support from the community including the community radio station staff that contacted me and gave me an opportunity to explain our side of the story as well as inform the community about the project progress and the way forward. I had some community leaders who came to apologise for the behaviour of the youths who took the law into their own hands and they also offered to help the police to bring the perpetuators to justice. In no time

the project was up and running and I found alternative transport under very difficult circumstances. This whole experience taught me the importance of recognition of authority when working with communities. Managing the field activities and team members financially was also crucial to the project. Despite the limited budget that I had to work with, I managed to achieve most of the objectives for the study. Fieldwork did not always run as smoothly as I would have liked but I did everything to the best of my ability under the circumstances, I believe that a good job was done.

8.0 References

- Feng G, Simpson JA, Chaluluka E, Molyneux ME, Rogerson SJ: Decreasing burden of malaria in pregnancy in Malawian women and its relationship to use of intermittent preventive therapy or bed nets. *PLoS One* 2010, 5:e12012.
- Harrington WE, Mutabingwa TK, Kabyemela E, Fried M, Duffy PE: Intermittent treatment to prevent pregnancy malaria does not confer benefit in an area of widespread drug resistance. *Clin Infect Dis* 2011, 53:224-230.
- Kayentao K, Garner P, van Eijk AM, Naidoo I, Roper C, Mulokozi A, MacArthur JR, Luntamo M, Ashorn P, Doumbo OK, ter Kuile FO: Intermittent preventive therapy for malaria during pregnancy using 2 vs 3 or more doses of sulfadoxine-pyrimethamine and risk of low birth weight in Africa: systematic review and meta-analysis. *Jama* 2013, 309:594-604.
- World Health Organization: Management of patients with sexually transmitted diseases.
 Report of a WHO Study Group. In World Health Organ Tech Rep Ser, vol. 810, 1991/01/01 edition. pp. 1-1031991:1-103.
- Pettifor A, Walsh J, Wilkins V, Raghunathan P: How effective is syndromic management of STDs?: A review of current studies. Sex Transm Dis 2000, 27:371-385.
- Chico RM, Mayaud P, Ariti C, Mabey D, Ronsmans C, Chandramohan D: Prevalence of Malaria and Sexually Transmitted and Reproductive Tract Infections in Pregnancy in Sub-Saharan Africa. JAMA 2012, Vol 307
- Mullick S, Watson-Jones D, Beksinska M, Mabey D: Sexually transmitted infections in pregnancy: prevalence, impact on pregnancy outcomes, and approach to treatment in developing countries. Sex Transm Infect 2005, 81:294-302.
- World Health Organization: Global prevalence and incidence of selected curable sexually transmitted infections. 2001.
- McDermott J, Steketee R, Larsen S, Wirima J: Syphilis-associated perinatal and infant mortality in rural Malawi. *Bull World Health Organ* 1993, 71:773-780.
- 10. Watson-Jones D CJ, Balthazar G, Weiss H, Rusizoka M, Ndeki L, Whitehouse A, Balira R, Todd J, Ngeleja D, Ross D, Buvé A, Hayes R, Mabey D Syphilis in pregnancy in Tanzania. I.
 Impact of maternal syphilis on outcome of pregnancy. J Infect Dis 2002, 186:940-947.
- Investigators of the Johns Hopkins Study of Cervicitis and Adverse Pregnancy Outcome: Association of Chlamydia trachomatis and Mycoplasma hominis with intrauterine growth retardation and preterm delivery. *Am J Epidemiol* 1989, **129**:1247–1257.

- 12. Gravett MG, Nelson HP, DeRouen T, Critchlow C, Eschenbach DA, Holmes KK: Independent associations of bacterial vaginosis and Chlamydia trachomatis infection with adverse pregnancy outcome. JAMA 1986, 256:1899-1903.
- Harrison HR, Alexander ER, Weinstein L, Lewis M, Nash M, Sim DA: Cervical Chlamydia trachomatis and mycoplasmal infections in pregnancy. Epidemiology and outcomes. JAMA 1983, 250:1721-1727.
- Plummer FA, Laga M, Brunham RC, Piot P, Ronald AR, Bhullar V, Mati JY, Ndinya-Achola JO,
 Cheang M, Nsanze H: Postpartum upper genital tract infections in Nairobi, Kenya:
 epidemiology, etiology, and risk factors. J Infect Dis 1987, 156:92-98.
- Laga M, Plummer FA, Nzanze H, Namaara W, Brunham RC, Ndinya-Achola JO, Maitha G, Ronald AR, D'Costa LJ, Bhullar VB, et al.: Epidemiology of ophthalmia neonatorum in Kenya. *Lancet* 1986, 2:1145-1149.
- Lacey CJ, Milne JD: Preterm labour in association with Neisseria gonorrhoeae: case reports.
 Br J Vener Dis 1984, 60:123-124.
- Elliott B, Brunham RC, Laga M, Piot P, Ndinya-Achola JO, Maitha G, Cheang M, Plummer FA:
 Maternal gonococcal infection as a preventable risk factor for low birth weight. J Infect Dis 1990, 161:531-536.
- 18. Donders GG, Desmyter J, De Wet DH, Van Assche FA: **The association of gonorrhoea and syphilis with premature birth and low birthweight.** *Genitourin Med* 1993, **69:**98-101.
- 19. Sutton MY, Sternberg M, Nsuami M, Behets F, Nelson AM, St Louis ME: Trichomoniasis in pregnant human immunodeficiency virus-infected and human immunodeficiency virusuninfected congolese women: prevalence, risk factors, and association with low birth weight. *Am J Obstet Gynecol* 1999, **181**:656-662.
- 20. Cotch MF, Pastorek JG, 2nd, Nugent RP, Hillier SL, Gibbs RS, Martin DH, Eschenbach DA, Edelman R, Carey JC, Regan JA, et al: Trichomonas vaginalis associated with low birth weight and preterm delivery. The Vaginal Infections and Prematurity Study Group. Sex Transm Dis 1997, 24:353-360.
- Minkoff H, Grunebaum AN, Schwarz RH, Feldman J, Cummings M, Crombleholme W, Clark L, Pringle G, McCormack WM: Risk factors for prematurity and premature rupture of membranes: a prospective study of the vaginal flora in pregnancy. *Am J Obstet Gynecol* 1984, 150:965-972.
- Hardy PH, Hardy JB, Nell EE, Graham DA, Spence MR, Rosenbaum RC: Prevalence of six sexually transmitted disease agents among pregnant inner-city adolescents and pregnancy outcome. Lancet 1984, 2:333-337.

- McGregor JA, French JI, Parker R, Draper D, Patterson E, Jones W, Thorsgard K, McFee J:
 Prevention of premature birth by screening and treatment for common genital tract infections: results of a prospective controlled evaluation. *Am J Obstet Gynecol* 1995, 173:157-167.
- Hillier SL, Nugent RP, Eschenbach DA, Krohn MA, Gibbs RS, Martin DH, Cotch MF, Edelman R, Pastorek JG, 2nd, Rao AV, et al.: Association between bacterial vaginosis and preterm delivery of a low-birth-weight infant. The Vaginal Infections and Prematurity Study Group. N Engl J Med 1995, 333:1737-1742.
- Riduan JM, Hillier SL, Utomo B, Wiknjosastro G, Linnan M, Kandun N: Bacterial vaginosis and prematurity in Indonesia: association in early and late pregnancy. *Am J Obstet Gynecol* 1993, 169:175-178.
- Arnesen L, Martinez G, Mainero L, Serruya S, Duran P: Gestational syphilis and stillbirth in
 Latin America and the Caribbean. Int J Gynaecol Obstet 2015, 128:241-245.
- 27. Gomez GB, Kamb ML, Newman LM, Mark J, Broutet N, Hawkes SJ: Untreated maternal syphilis and adverse outcomes of pregnancy: a systematic review and meta-analysis. Bull World Health Organ 2013, 91:217-226.
- 28. Odendaal HJ, Schoeman J, Grove D, de Jager M, Theron GB, Orth H, McMahon C, Chalkley LJ:
 The association between Chlamydia trachomatis genital infection and spontaneous
 preterm labour. South African Journal of Obstetrics and Gynaecology 2006, 12:146-149.
- Nakubulwa S, Kaye DK, Bwanga F, Tumwesigye NM, Mirembe FM: Genital infections and risk of premature rupture of membranes in Mulago Hospital, Uganda: a case control study. BMC Res Notes 2015, 8:573.
- Rours GI, Duijts L, Moll HA, Arends LR, de Groot R, Jaddoe VW, Hofman A, Steegers EA, Mackenbach JP, Ott A, et al: Chlamydia trachomatis infection during pregnancy associated with preterm delivery: a population-based prospective cohort study. *Eur J Epidemiol* 2011, 26:493-502.
- Baud D, Goy G, Vasilevsky S, Osterheld MC, Roth-Kleiner M, Croxatto A, Greub G: Roles of bovine Waddlia chondrophila and Chlamydia trachomatis in human preterm birth. New Microbes New Infect 2015, 3:41-45.
- 32. Liu B, Roberts CL, Clarke M, Jorm L, Hunt J, Ward J: Chlamydia and gonorrhoea infections and the risk of adverse obstetric outcomes: a retrospective cohort study. *Sex Transm Infect* 2013, **89:**672-678.

- 33. Johnson HL, Ghanem KG, Zenilman JM, Erbelding EJ: Sexually transmitted infections and adverse pregnancy outcomes among women attending inner city public sexually transmitted diseases clinics. Sex Transm Dis 2011, 38:167-171.
- 34. Blas MM, Canchihuaman FA, Alva IE, Hawes SE: Pregnancy outcomes in women infected with Chlamydia trachomatis: A population-based cohort study in Washington State. Sexually Transmitted Infections 2007, 83:314-318.
- 35. Silveira MF, Ghanem KG, Erbelding EJ, Burke AE, Johnson HL, Singh RH, Zenilman JM:
 Chlamydia trachomatis infection during pregnancy and the risk of preterm birth: A casecontrol study. International Journal of STD and AIDS 2009, 20:465-469.
- Watson-Jones D, Weiss HA, Changalucha JM, Todd J, Gumodoka B, Bulmer J, Balira R, Ross D,
 Mugeye K, Hayes R, Mabey D: Adverse birth outcomes in United Republic of Tanzania- impact and prevention of maternal risk factors. Bull World Health Organ 2007, 85:9-18.
- 37. Svare JA, Schmidt H, Hansen BB, Lose G: Bacterial vaginosis in a cohort of Danish pregnant women: prevalence and relationship with preterm delivery, low birthweight and perinatal infections. *BJOG* 2006, **113**:1419-1425.
- 38. Oakeshott P, Kerry S, Hay S, Hay P: Bacterial vaginosis and preterm birth: a prospective community-based cohort study. *Br J Gen Pract* 2004, **54:**119-122.
- Leitich H, Bodner-Adler B, Brunbauer M, Kaider A, Egarter C, Husslein P: Bacterial vaginosis as a risk factor for preterm delivery: a meta-analysis. *Am J Obstet Gynecol* 2003, 189:139-147.
- 40. Leitich H, Kiss H: Asymptomatic bacterial vaginosis and intermediate flora as risk factors for adverse pregnancy outcome. *Best Pract Res Clin Obstet Gynaecol* 2007, **21:**375-390.
- Daskalakis G, Papapanagiotou A, Mesogitis S, Papantoniou N, Mavromatis K, Antsaklis A:
 Bacterial vaginosis and group B streptococcal colonization and preterm delivery in a lowrisk population. *Fetal Diagn Ther* 2006, 21:172-176.
- 42. Brabin BJ: An analysis of malaria in pregnancy in Africa. Bull World Health Organ 1983,61:1005-1016.
- 43. Meeusen EN, Bischof RJ, Lee CS: **Comparative T-cell responses during pregnancy in large animals and humans.** *Am J Reprod Immunol* 2001, **46:**169-179.
- Fried M, Duffy PE: Adherence of Plasmodium falciparum to chondroitin sulfate A in the human placenta. *Science* 1996, 272:1502-1504.
- 45. Garnham PCC: **The placenta in malaria with special reference to reticuloendothelial immunity.** *Trans R Soc Trop Med Hyg* 1938, **32:**13–22.

- Poespoprodjo JR, Fobia W, Kenangalem E, Lampah DA, Warikar N, Seal A, McGready R,
 Sugiarto P, Tjitra E, Anstey NM, Price RN: Adverse pregnancy outcomes in an area where
 multidrug-resistant plasmodium vivax and Plasmodium falciparum infections are endemic.
 Clin Infect Dis 2008, 46:1374-1381.
- 47. McGregor IA: **Epidemiology, malaria and pregnancy.** *Am J Trop Med Hyg* 1984, **33:**517-525.
- 48. McGregor IA, Wilson ME, Billewicz WZ: Malaria infection of the placenta in The Gambia,
 West Africa; its incidence and relationship to stillbirth, birthweight and placental weight.
 Trans R Soc Trop Med Hyg 1983, 77:232-244.
- Bader E, Alhaj AM, Hussan AA, Adam I: Malaria and stillbirth in Omdurman Maternity
 Hospital, Sudan. Int J Gynaecol Obstet 2010, 109:144-146.
- Yatich NJ, Funkhouser E, Ehiri JE, Agbenyega T, Stiles JK, Rayner JC, Turpin A, Ellis WO, Jiang Y, Williams JH, et al: Malaria, intestinal helminths and other risk factors for stillbirth in Ghana. Infect Dis Obstet Gynecol 2010, 2010:350763.
- 51. Greenwood AM, Armstrong JR, Byass P, Snow RW, Greenwood BM: Malaria
 chemoprophylaxis, birth weight and child survival. *Trans R Soc Trop Med Hyg* 1992, 86:483-485.
- 52. Guyatt HL, Snow RW: Malaria in pregnancy as an indirect cause of infant mortality in sub-Saharan Africa. *Trans R Soc Trop Med Hyg* 2001, **95:**569-576.
- 53. Murphy SC, Breman JG: Gaps in the childhood malaria burden in Africa: cerebral malaria, neurological sequelae, anemia, respiratory distress, hypoglycemia, and complications of pregnancy. *Am J Trop Med Hyg* 2001, **64:**57-67.
- 54. Steketee RW, Nahlen BL, Parise ME, Menendez C: **The burden of malaria in pregnancy in** malaria-endemic areas. *Am J Trop Med Hyg* 2001, **64:**28-35.
- 55. Steketee RW, Wirima JJ, Hightower AW, Slutsker L, Heymann DL, Breman JG: **The effect of** malaria and malaria prevention in pregnancy on offspring birthweight, prematurity, and intrauterine growth retardation in rural Malawi. *Am J Trop Med Hyg* 1996, **55:**33-41.
- 56. van den Broek NR, Jean-Baptiste R, Neilson JP: Factors associated with preterm, early preterm and late preterm birth in Malawi. *PLoS One* 2014, **9:**e90128.
- 57. Mohammed AH, Salih MM, Elhassan EM, Mohmmed AA, Elzaki SE, El-Sayed BB, Adam I:
 Submicroscopic Plasmodium falciparum malaria and low birth weight in an area of unstable malaria transmission in Central Sudan. *Malar J* 2013, 12:172.
- 58. Asundep NN, Jolly PE, Carson AP, Turpin CA, Zhang K, Wilson NO, Stiles JK, Tameru B: Effect of Malaria and Geohelminth Infection on Birth Outcomes in Kumasi, Ghana. *Int J Trop Dis Health* 2014, **4**:582-594.

- 59. Ratnam AV, Din SN, Hira SK, Bhat GJ, Wacha DS, Rukmini A, Mulenga RC: **Syphilis in** pregnant women in Zambia. Br J Vener Dis 1982, **58:**355-358.
- Temmerman M, Gichangi P, Fonck K, Apers L, Claeys P, Van Renterghem L, Kiragu D, Karanja G, Ndinya-Achola J, Bwayo J: Effect of a syphilis control programme on pregnancy outcome in Nairobi, Kenya. Sex Transm Infect 2000, 76:117-121.
- 61. Watson-Jones D, Gumodoka B, Weiss H, Changalucha J, Todd J, Mugeye K, Buve A, Kanga Z, Ndeki L, Rusizoka M, et al: **Syphilis in pregnancy in Tanzania. II. The effectiveness of antenatal syphilis screening and single-dose benzathine penicillin treatment for the prevention of adverse pregnancy outcomes.** J Infect Dis 2002, **186**:948-957.
- 62. Sweet RL, Landers DV, Walker C, Schachter J: Chlamydia trachomatis infection and pregnancy outcome. *Am J Obstet Gynecol* 1987, **156**:824-833.
- 63. Sozio J, Ness RB: Chlamydial lower genital tract infection and spontaneous abortion. *Infect Dis Obstet Gynecol* 1998, **6:**8-12.
- 64. Kovacs L, Nagy E, Berbik I, Meszaros G, Deak J, Nyari T: The frequency and the role of
 Chlamydia trachomatis infection in premature labor. Int J Gynaecol Obstet 1998, 62:47-54.
- Meis PJ, Goldenberg RL, Mercer B, Moawad A, Das A, McNellis D, Johnson F, Iams JD, Thom
 E, Andrews WW: The preterm prediction study: significance of vaginal infections. National
 Institute of Child Health and Human Development Maternal-Fetal Medicine Units
 Network. Am J Obstet Gynecol 1995, 173:1231-1235.
- 66. Jacobsson B, Pernevi P, Chidekel L, Jorgen Platz-Christensen J: **Bacterial vaginosis in early** pregnancy may predispose for preterm birth and postpartum endometritis. *Acta Obstet Gynecol Scand* 2002, **81**:1006-1010.
- Gray RH, Wabwire-Mangen F, Kigozi G, Sewankambo NK, Serwadda D, Moulton LH, Quinn TC, O'Brien KL, Meehan M, Abramowsky C, et al: Randomized trial of presumptive sexually transmitted disease therapy during pregnancy in Rakai, Uganda. *Am J Obstet Gynecol* 2001, 185:1209-1217.
- McGregor JA, French JI, Richter R, Vuchetich M, Bachus V, Seo K, Hillier S, Judson FN, McFee J, Schoonmaker J, et al.: Cervicovaginal microflora and pregnancy outcome: results of a double-blind, placebo-controlled trial of erythromycin treatment. *Am J Obstet Gynecol* 1990, 163:1580-1591.
- 69. Temmerman M, Njagi E, Nagelkerke N, Ndinya-Achola J, Plummer FA, Meheus A: Mass antimicrobial treatment in pregnancy. A randomized, placebo-controlled trial in a population with high rates of sexually transmitted diseases. J Reprod Med 1995, 40:176-180.

- Ryan GM, Jr., Abdella TN, McNeeley SG, Baselski VS, Drummond DE: Chlamydia trachomatis infection in pregnancy and effect of treatment on outcome. *Am J Obstet Gynecol* 1990, 162:34-39.
- Cohen I, Veille JC, Calkins BM: Improved pregnancy outcome following successful treatment of chlamydial infection. JAMA 1990, 263:3160-3163.
- 72. Rastogi S, Das B, Salhan S, Mittal A: Effect of treatment for Chlamydia trachomatis during pregnancy. *Int J Gynaecol Obstet* 2003, **80:**129-137.
- Hauth JC, Goldenberg RL, Andrews WW, DuBard MB, Copper RL: Reduced incidence of preterm delivery with metronidazole and erythromycin in women with bacterial vaginosis. N Engl J Med 1995, 333:1732-1736.
- Koumans EH, Lane SD, Aubry R, Demott K, Webster N, Levandowski BA, Berman S,
 Markowitz LE: Evaluation of Syracuse Healthy Start's program for abnormal flora
 management to reduce preterm birth among pregnant women. Matern Child Health J
 2011, 15:1020-1028.
- 75. Okun N, Gronau KA, Hannah ME: Antibiotics for bacterial vaginosis or Trichomonas vaginalis in pregnancy: a systematic review. *Obstet Gynecol* 2005, **105:**857-868.
- 76. Lamont RF, Duncan SL, Mandal D, Bassett P: Intravaginal clindamycin to reduce preterm birth in women with abnormal genital tract flora. *Obstet Gynecol* 2003, **101:**516-522.
- 77. Nygren P, Fu R, Freeman M, Bougatsos C, Klebanoff M, Guise JM: Evidence on the benefits and harms of screening and treating pregnant women who are asymptomatic for bacterial vaginosis: an update review for the U.S. Preventive Services Task Force. *Ann Intern Med* 2008, **148:**220-233.
- 78. Ross SM, van Middelkoop A: Trichomonas infection in pregnancy--does it affect perinatal outcome? S Afr Med J 1983, 63:566-567.
- Klebanoff MA, Carey JC, Hauth JC, Hillier SL, Nugent RP, Thom EA, Ernest JM, Heine RP,
 Wapner RJ, Trout W, et al: Failure of metronidazole to prevent preterm delivery among
 pregnant women with asymptomatic Trichomonas vaginalis infection. N Engl J Med 2001,
 345:487-493.
- Kigozi GG, Brahmbhatt H, Wabwire-Mangen F, Wawer MJ, Serwadda D, Sewankambo N,
 Gray RH: Treatment of Trichomonas in pregnancy and adverse outcomes of pregnancy: a
 subanalysis of a randomized trial in Rakai, Uganda. Am J Obstet Gynecol 2003, 189:1398 1400.
- 81. Carey JC, Klebanoff MA, Hauth JC, Hillier SL, Thom EA, Ernest JM, Heine RP, Nugent RP,
 Fischer ML, Leveno KJ, et al: Metronidazole to prevent preterm delivery in pregnant

women with asymptomatic bacterial vaginosis. National Institute of Child Health and Human Development Network of Maternal-Fetal Medicine Units. *N Engl J Med* 2000, **342:**534-540.

- McDonald HM, O'Loughlin JA, Vigneswaran R, Jolley PT, Harvey JA, Bof A, McDonald PJ:
 Impact of metronidazole therapy on preterm birth in women with bacterial vaginosis flora (Gardnerella vaginalis): a randomised, placebo controlled trial. *Br J Obstet Gynaecol* 1997, 104:1391-1397.
- 83. Morales WJ, Schorr S, Albritton J: Effect of metronidazole in patients with preterm birth in preceding pregnancy and bacterial vaginosis: a placebo-controlled, double-blind study. Am J Obstet Gynecol 1994, 171:345-347; discussion 348-349.
- Stringer E, Read JS, Hoffman I, Valentine M, Aboud S, Goldenberg RL: Treatment of
 trichomoniasis in pregnancy in sub-Saharan Africa does not appear to be associated with
 low birth weight or preterm birth. SAMJ South African Medical Journal 2010, 100:58-64.
- 85. Guyatt HL, Snow RW: Impact of malaria during pregnancy on low birth weight in sub-Saharan Africa. *Clin Microbiol Rev* 2004, **17:**760-769, table of contents.
- Besai M, ter Kuile FO, Nosten F, McGready R, Asamoa K, Brabin B, Newman RD:
 Epidemiology and burden of malaria in pregnancy. *Lancet Infect Dis* 2007, 7:93-104.
- 87. Kelsey JL, Whittemore AS, Evans AS, Thompson WD: *Methods in Observational Epidemiology.* 2nd edition edn. New York, NY: Oxford University Press; 1996.
- 88. Rogerson SJ, Chaluluka E, Kanjala M, Mkundika P, Mhango C, Molyneux ME: Intermittent sulfadoxine-pyrimethamine in pregnancy: effectiveness against malaria morbidity in Blantyre, Malawi, in 1997-99. *Trans R Soc Trop Med Hyg* 2000, 94:549-553.
- 89. Valente B, Campos PA, do Rosario VE, Varandas L, Silveira H: Prevalence and risk factors of Plasmodium falciparum infections in pregnant women of Luanda, Angola. *Trop Med Int Health* 2011, 16:1206-1214.
- 90. Mace KE, Chalwe V, Katalenich BL, Nambozi M, Mubikayi L, Mulele CK, Wiegand RE, Filler SJ, Kamuliwo M, Craig AS, Tan KR: Evaluation of sulphadoxine-pyrimethamine for intermittent preventive treatment of malaria in pregnancy: a retrospective birth outcomes study in Mansa, Zambia. *Malar J* 2015, 14:69.
- Barat L, Chipipa J, Kolczak M, Sukwa T: Does the availability of blood slide microscopy for malaria at health centers improve the management of persons with fever in Zambia? *Am J Trop Med Hyg* 1999, 60:1024-1030.

- 92. Reyburn H, Mbatia R, Drakeley C, Carneiro I, Mwakasungula E, Mwerinde O, Saganda K, Shao J, Kitua A, Olomi R, et al: Overdiagnosis of malaria in patients with severe febrile illness in Tanzania: a prospective study. *Bmj* 2004, 329:1212.
- 93. Reyburn H, Ruanda J, Mwerinde O, Drakeley C: **The contribution of microscopy to targeting** antimalarial treatment in a low transmission area of Tanzania. *Malar J* 2006, **5**:4.
- 94. Snounou G, Viriyakosol S, Jarra W, Thaithong S, Brown KN: Identification of the four human malaria parasite species in field samples by the polymerase chain reaction and detection of a high prevalence of mixed infections. *Mol Biochem Parasitol* 1993, 58:283-292.
- 95. Rantala AM, Taylor SM, Trottman PA, Luntamo M, Mbewe B, Maleta K, Kulmala T, Ashorn P, Meshnick SR: **Comparison of real-time PCR and microscopy for malaria parasite detection in Malawian pregnant women.** *Malar J* 2010, **9:**269.
- 96. Mercereau-Puijalon O, Fandeur T, Bonnefoy S, Jacquemot C, Sarthou JL: A study of the genomic diversity of Plasmodium falciparum in Senegal. 2. Typing by the use of the polymerase chain reaction. *Acta Trop* 1991, **49:**293-304.
- 97. Roper C, Elhassan IM, Hviid L, Giha H, Richardson W, Babiker H, Satti GM, Theander TG, Arnot DE: Detection of very low level Plasmodium falciparum infections using the nested polymerase chain reaction and a reassessment of the epidemiology of unstable malaria in Sudan. *Am J Trop Med Hyg* 1996, **54:**325-331.
- 98. Coleman RE, Sattabongkot J, Promstaporm S, Maneechai N, Tippayachai B, Kengluecha A, Rachapaew N, Zollner G, Miller RS, Vaughan JA, et al: Comparison of PCR and microscopy for the detection of asymptomatic malaria in a Plasmodium falciparum/vivax endemic area in Thailand. *Malar J* 2006, 5:121.
- Singh B, Cox-Singh J, Miller AO, Abdullah MS, Snounou G, Rahman HA: Detection of malaria in Malaysia by nested polymerase chain reaction amplification of dried blood spots on filter papers. *Trans R Soc Trop Med Hyg* 1996, 90:519-521.
- Jelinek T, Proll S, Hess F, Kabagambe G, von Sonnenburg F, Loscher T, Kilian AH: Geographic differences in the sensitivity of a polymerase chain reaction for the detection of Plasmodium falciparum infection. *Am J Trop Med Hyg* 1996, 55:647-651.
- 101. Kain KC, Brown AE, Mirabelli L, Webster HK: **Detection of Plasmodium vivax by polymerase** chain reaction in a field study. *J Infect Dis* 1993, **168**:1323-1326.
- 102. Farnert A, Arez AP, Correia AT, Bjorkman A, Snounou G, do Rosario V: Sampling and storage of blood and the detection of malaria parasites by polymerase chain reaction. *Trans R Soc Trop Med Hyg* 1999, 93:50-53.

- Makler MT, Palmer CJ, Ager AL: A review of practical techniques for the diagnosis of malaria. Ann Trop Med Parasitol 1998, 92:419-433.
- 104. Rogerson SJ, Mkundika P, Kanjala MK: Diagnosis of Plasmodium falciparum malaria at delivery: comparison of blood film preparation methods and of blood films with histology. J Clin Microbiol 2003, 41:1370-1374.
- 105. Anchang-Kimbi JK, Achidi EA, Nkegoum B, Sverremark-Ekstrom E, Troye-Blomberg M:
 Diagnostic comparison of malaria infection in peripheral blood, placental blood and
 placental biopsies in Cameroonian parturient women. *Malar J* 2009, 8:126.
- Ministry of Health: Health Management Information System (HMIS). Lusaka, Zambia: Ministry of Health; 2008.
- 107. CDC and USAID: Maternal Child Health Intergrated programme: A malaria in pregnancy case study:Zambia's Successes and Remaining Challenges for Malaria in Pregnancy Programming. CDC and USAID; 2010.
- 108. Ministry of Health: Zambia National Malaria Indicator Survey 2010. Lusaka, Zambia: Ministry of Health, Central Statistics Office, PATH Malaria Control and Evaluation partnership in Africa (MACEPA), the United States President's Malaria Initiative, the World Bank and World Health Organisation; 2012.
- Masaninga F, Chanda E, Chanda-Kapata P, Hamainza B, T. MH, M. K, Kapelwa W, J. C, Govere J, Fall IS, Babaniyi O: Review of the malaria epidemiology and trends in Zambia. Asian Pacific Journal of Tropical Biomedicine 2012:1-5.
- 110. Anon: A Six-year Strategic plan: a roadmap for impact on malaria in Zambia 2006-2011.Ministry of Health, Government of the Republic of Zambia; 2006.
- 111. Ministry of Health: **Guidelines for the diagnosis and treatment of Malaria in Zambia.**, Third edition. pp. 41-42. Lusaka: Ministry of Health; 2010:41-42.
- 112. Chanda PHM, Mharakurwa S, Shinondo C, Roper C, Pota H: Frequency of plasmodium falciparum dihydrofolate reductase and synthase resistance markers in six districts in Zambia. Medical Journal of Zambia 2007, 34:58-61.
- 113. Mkulama MA, Chishimba S, Sikalima J, Rouse P, Thuma PE, Mharakurwa S: Escalating
 Plasmodium falciparum antifolate drug resistance mutations in Macha, rural Zambia.
 Malar J 2008, 7:87.
- Tan KR, Katalenich BL, Mace KE, Nambozi M, Taylor SM, Meshnick SR, Wiegand RE, Chalwe
 V, Filler SJ, Kamuliwo M, Craig AS: Efficacy of sulphadoxine-pyrimethamine for intermittent
 preventive treatment of malaria in pregnancy, Mansa, Zambia. *Malar J* 2014, 13:227.

- 115. Siame MN, Mharakurwa S, Chipeta J, Thuma P, Michelo C: High prevalence of dhfr and dhps molecular markers in Plasmodium falciparum in pregnant women of Nchelenge district, Northern Zambia. *Malar J* 2015, 14:190.
- 116. Peeling RW, Ye H: Diagnostic tools for preventing and managing maternal and congenital syphilis: an overview. *Bull World Health Organ* 2004, **82:**439-446.
- 117. Kuypers J, World Health Organisation: *Laboratory tests for detection of reproductive tract infections. In: jejeeboy S, Koenig M, Elias C, eds. Reproductive Tract Infections and other Gynaecological Disorders.* Cambridge: Cambridge University Press; 2003.
- 118. Kuypers J, Gaydos CA, Peeling RW: Principles of Laboratory Diagnosis. In: Holmes KK SF,
 Stamm WE, PIOT P, Wasserheit JN, Corey L, Cohen M,, ed. Sexually Transmitted Diseases. .
 4th ed edn. New York: McGraw Hill; 2008.
- 119. Amsel R, Totten PA, Spiegel CA, Chen KC, Eschenbach D, Holmes KK: Nonspecific vaginitis.
 Diagnostic criteria and microbial and epidemiologic associations. *Am J Med* 1983, 74:14-22.
- 120. Nugent RP, Krohn MA, Hillier SL: **Reliability of diagnosing bacterial vaginosis is improved by** a standardized method of gram stain interpretation. *J Clin Microbiol* 1991, **29**:297-301.
- 121. Dadhwal V, Hariprasad R, Mittal S, Kapil A: **Prevalence of bacterial vaginosis in pregnant** women and predictive value of clinical diagnosis. *Arch Gynecol Obstet* 2010, **281:**101-104.
- 122. Schwebke JR, Hillier SL, Sobel JD, McGregor JA, Sweet RL: Validity of the vaginal gram stain for the diagnosis of bacterial vaginosis. *Obstet Gynecol* 1996, 88:573-576.
- 123. Goyal R, Sharma P, Kaur I, Aggarwal N, Talwar V: Bacterial vaginosis and vaginal anaerobes in preterm labour. J Indian Med Assoc 2004, **102:**548-550, 553.
- 124. Thomason JL, Anderson RJ, Gelbart SM, Osypowski PJ, Scaglione NJ, el Tabbakh G, James JA:
 Simplified gram stain interpretive method for diagnosis of bacterial vaginosis. *Am J Obstet Gynecol* 1992, 167:16-19.
- 125. Latif AS, Mason PR, Marowa E, Gwanzura L, Chingono A, Mbengeranwa OL: Risk factors for gonococcal and chlamydial cervical infection in pregnant and non-pregnant women in Zimbabwe. Cent Afr J Med 1999, 45:252-258.
- 126. Mayaud P, Grosskurth H, Changalucha J, Todd J, West B, Gabone R, Senkoro K, Rusizoka M, Laga M, Hayes R, et al.: Risk assessment and other screening options for gonorrhoea and chlamydial infections in women attending rural Tanzanian antenatal clinics. *Bull World Health Organ* 1995, **73:**621-630.
- 127. Romoren M, Sundby J, Velauthapillai M, Rahman M, Klouman E, Hjortdahl P: **Chlamydia and** gonorrhoea in pregnant Batswana women: time to discard the syndromic approach? *BMC Infect Dis* 2007, **7:**27.

- 128. Kurewa NE, Mapingure MP, Munjoma MW, Chirenje MZ, Rusakaniko S, Stray-Pedersen B: The burden and risk factors of Sexually Transmitted Infections and Reproductive Tract Infections among pregnant women in Zimbabwe. BMC Infect Dis 2010, 10:127.
- 129. Mwakagile D, Swai AB, Sandstrom E, Urassa E, Biberfeld G, Mhalu FS: **High frequency of** sexually transmitted diseases among pregnant women in Dar es Salaam, Tanzania: need for intervention. *East Afr Med J* 1996, **73:**675-678.
- Hokororo A, Kihunrwa A, Hoekstra P, Kalluvya SE, Changalucha JM, Fitzgerald DW, Downs JA:
 High prevalence of sexually transmitted infections in pregnant adolescent girls in Tanzania:
 a multi-community cross-sectional study. Sex Transm Infect 2015.
- 131. Cossa HA, Gloyd S, Vaz RG, Folgosa E, Simbine E, Diniz M, Kreiss JK: Syphilis and HIV infection among displaced pregnant women in rural Mozambique. *Int J STD AIDS* 1994, 5:117-123.
- Pham L, Woelk GB, Ning Y, Madzime S, Mudzamiri S, Mahomed K, Williams MA:
 Seroprevalence and risk factors of syphilis infection in pregnant women delivering at
 Harare Maternity Hospital, Zimbabwe. Cent Afr J Med 2005, 51:24-30.
- 133. Yahya-Malima KI, Evjen-Olsen B, Matee MI, Fylkesnes K, Haarr L: HIV-1, HSV-2 and syphilis among pregnant women in a rural area of Tanzania: prevalence and risk factors. BMC Infect Dis 2008, 8:75.
- 134. Urassa WK, Kapiga SH, Msamanga GI, Antelman G, Coley J, Fawzi WW: Risk factors for syphilis among HIV-1 infected pregnant women in Dar es Salaam, Tanzania. Afr J Reprod Health 2001, 5:54-62.
- Potter D, Goldenberg RL, Read JS, Wang J, Hoffman IF, Saathoff E, Kafulafula G, Aboud S,
 Martinson FE, Dahab M, Vermund SH: Correlates of syphilis seroreactivity among pregnant
 women: the HIVNET 024 Trial in Malawi, Tanzania, and Zambia. Sex Transm Dis 2006,
 33:604-609.
- 136. Swai RO, Somi GG, Matee MI, Killewo J, Lyamuya EF, Kwesigabo G, Tulli T, Kabalimu TK, Ng'ang'a L, Isingo R, Ndayongeje J: Surveillance of HIV and syphilis infections among antenatal clinic attendees in Tanzania-2003/2004. BMC Public Health 2006, 6:91.
- 137. Kwiek JJ, Mwapasa V, Alker AP, Muula AS, Misiri HE, Molyneux ME, Rogerson SJ, Behets FM, Meshnick SR: Socio-demographic characteristics associated with HIV and syphilis seroreactivity among pregnant women in Blantyre, Malawi, 2000-2004. Malawi Med J 2008, 20:80-85.

- Manyahi J, Jullu BS, Abuya MI, Juma J, Ndayongeje J, Kilama B, Sambu V, Nondi J, Rabiel B,
 Somi G, Matee MI: Prevalence of HIV and syphilis infections among pregnant women
 attending antenatal clinics in Tanzania, 2011. BMC Public Health 2015, 15:501.
- 139. Menendez C, Castellsague X, Renom M, Sacarlal J, Quinto L, Lloveras B, Klaustermeier J, Kornegay JR, Sigauque B, Bosch FX, Alonso PL: Prevalence and risk factors of sexually transmitted infections and cervical neoplasia in women from a rural area of southern Mozambique. Infect Dis Obstet Gynecol 2010, 2010.
- Aboud S, Msamanga G, Read JS, Mwatha A, Chen YQ, Potter D, Valentine M, Sharma U,
 Hoffmann I, Taha TE, et al: Genital tract infections among HIV-infected pregnant women in
 Malawi, Tanzania and Zambia. Int J STD AIDS 2008, 19:824-832.
- 141. Rogerson SJ, van den Broek NR, Chaluluka E, Qongwane C, Mhango CG, Molyneux ME:
 Malaria and anemia in antenatal women in Blantyre, Malawi: a twelve-month survey. Am J Trop Med Hyg 2000, 62:335-340.
- 142. Wort UU, Warsame M, Brabin BJ: **Birth outcomes in adolescent pregnancy in an area with intense malaria transmission in Tanzania.** *Acta Obstet Gynecol Scand* 2006, **85:**949-954.
- Serra-Casas E, Menendez C, Dobano C, Bardaji A, Quinto L, Ordi J, Sigauque B, Cistero P,
 Mandomando I, Alonso PL, Mayor A: Persistence of Plasmodium falciparum parasites in
 infected pregnant Mozambican women after delivery. *Infect Immun* 2011, 79:298-304.
- Taylor SM, van Eijk AM, Hand CC, Mwandagalirwa K, Messina JP, Tshefu AK, Atua B, Emch M, Muwonga J, Meshnick SR, Ter Kuile FO: Quantification of the burden and consequences of pregnancy-associated malaria in the Democratic Republic of the Congo. J Infect Dis 2011, 204:1762-1771.
- 145. Campos PA, Valente B, Campos RB, Goncalves L, Rosario VE, Varandas L, Silveira H:
 Plasmodium falciparum infection in pregnant women attending antenatal care in Luanda,
 Angola. *Rev Soc Bras Med Trop* 2012, 45:369-374.
- 146. Kabyemela ER, Fried M, Kurtis JD, Mutabingwa TK, Duffy PE: Decreased susceptibility to
 Plasmodium falciparum infection in pregnant women with iron deficiency. J Infect Dis
 2008, 198:163-166.
- 147. Mosha D, Chilongola J, Ndeserua R, Mwingira F, Genton B: Effectiveness of intermittent preventive treatment with sulfadoxine-pyrimethamine during pregnancy on placental malaria, maternal anaemia and birthweight in areas with high and low malaria transmission intensity in Tanzania. *Trop Med Int Health* 2014, **19:**1048-1056.
- 148. World Health Organisation: A Strategic Framework for Malaria Prevention and Control during Pregnancy in the African Region Brazzaville: WHO Regional Office for Africa; 2004.

- 149. Njagi JK, Magnussen P, Estambale B, Ouma J, Mugo B: **Prevention of anaemia in pregnancy** using insecticide-treated bednets and sulfadoxine-pyrimethamine in a highly malarious area of Kenya: a randomized controlled trial. *Trans R Soc Trop Med Hyg* 2003, **97**:277-282.
- 150. Kayentao K, Kodio M, Newman RD, Maiga H, Doumtabe D, Ongoiba A, Coulibaly D, Keita AS, Maiga B, Mungai M, et al: Comparison of intermittent preventive treatment with chemoprophylaxis for the prevention of malaria during pregnancy in Mali. J Infect Dis 2005, 191:109-116.
- 151. Schultz LJ, Steketee RW, Macheso A, Kazembe P, Chitsulo L, Wirima JJ: **The efficacy of** antimalarial regimens containing sulfadoxine-pyrimethamine and/or chloroquine in preventing peripheral and placental Plasmodium falciparum infection among pregnant women in Malawi. *Am J Trop Med Hyg* 1994, **51**:515-522.
- 152. Shulman CE, Dorman EK, Cutts F, Kawuondo K, Bulmer JN, Peshu N, Marsh K: Intermittent sulphadoxine-pyrimethamine to prevent severe anaemia secondary to malaria in pregnancy: a randomised placebo-controlled trial. *Lancet* 1999, **353**:632-636.
- 153. ter Kuile FO, van Eijk AM, Filler SJ: Effect of sulfadoxine-pyrimethamine resistance on the efficacy of intermittent preventive therapy for malaria control during pregnancy: a systematic review. JAMA 2007, 297:2603-2616.
- 154. Fried M, Nosten F, Brockman A, Brabin BJ, Duffy PE: Maternal antibodies block malaria.
 Nature 1998, 395:851-852.
- 155. Mehaffey PC, Barrett MS, Putnam SD, Jones RN: Antigonococcal activity of 11 drugs used for therapy or prophylaxis of malaria. *Diagn Microbiol Infect Dis* 1995, 23:11-13.
- 156. Sombie I, Meda N, Cartoux M, Tiendrebeogo S, Ouangre A, Yaro S, Ky-Zerbo O, Dao B, Van de Perre P, Mandelbrot L, Dabis F: Seroprevalence of syphilis among women attending urban antenatal clinics in Burkina Faso, 1995-8. The DITRAME Study Group. Diminunation de la TRAnsmission Mere-Enfant. Sex Transm Infect 2000, 76:314-316.
- 157. Chico RM, Chandramohan D: Intermittent preventive treatment of malaria in pregnancy: at the crossroads of public health policy. *Trop Med Int Health* 2011, **16:**774-785.
- 158. Vallely A, Vallely L, Changalucha J, Greenwood B, Chandramohan D: Intermittent preventive treatment for malaria in pregnancy in Africa: what's new, what's needed? *Malar J* 2007, 6:16.
- 159. Gesase S, Gosling RD, Hashim R, Ord R, Naidoo I, Madebe R, Mosha JF, Joho A, Mandia V,
 Mrema H, et al: High resistance of Plasmodium falciparum to
 sulphadoxine/pyrimethamine in northern Tanzania and the emergence of dhps resistance
 mutation at Codon 581. *PLoS One* 2009, 4:e4569.

- 160. Moore JM, Nahlen BL, Lal AA, Udhayakumar V: **Immunologic memory in the placenta: a lymphocyte recirculation hypothesis.** *Med Hypotheses* 2000, **54:**505-510.
- 161. Chico RM, Cano J, Ariti C, Collier TJ, Chandramohan D, Roper C, Greenwood B: Influence of Malaria Transmission Intensity and the 581G Mutation on the Efficacy of Intermittent Preventive Treatment in Pregnancy: Systematic Review and Meta-analysis. Trop Med Int Health 2015.
- 162. Centers for Disease Control and Prevention: Sexually transmitted diseases treatment guidelines: MMWR Recommendations. vol. Report 51. pp. 1–78.: Centers for Disease Control and Prevention. ; 2002:1–78.
- 163. World Health Organization: Guidelines for the Management of Sexually TransmittedInfections. Geneva: World Health Organisation; 2001.
- 164. World Health Organization: Global Prevalence and Incidence of selected curable Sexually
 Transmitted Infection: Overview and Estimates. Geneva: World Health Organisation; 1991.
- 165. Msuya SE, Uriyo J, Stray-Pedersen B, Sam NE, Mbizvo EM: **The effectiveness of a syndromic approach in managing vaginal infections among pregnant women in northern Tanzania.** *East Afr J Public Health* 2009, **6:**263-267.
- Baisley K, Changalucha J, Weiss HA, Mugeye K, Everett D, Hambleton I, Hay P, Ross D, Tanton
 C, Chirwa T, et al: Bacterial vaginosis in female facility workers in north-western Tanzania:
 prevalence and risk factors. Sex Transm Infect 2009, 85:370-375.
- 167. Central Statistical Office (CSO) [Zambia], Ministry of Health (MOH) [Zambia], ICF
 International: Zambia Demographic and Health Survey 2013-14. Rockville, Maryland, USA:
 Central Statistical Office, Ministry of Health, and ICF International 2014.
- 168. Central Statistical Office (CSO) MoHM TDRCT, University of Zambia, Macro International Inc:
 : Zambia Demographic and Health Survey 2007. Calverton, Maryland, USA: CSO and Macro International Inc.; 2009.
- 169. Minstry of Health: NATIONAL HIV/AIDS/STI/TB POLICY. Lusaka2002.
- 170. Kyei NN, Chansa C, Gabrysch S: Quality of antenatal care in Zambia: a national assessment.
 BMC Pregnancy Childbirth 2012, 12:151.
- 171. Taylor SD: Culture and Customs of Zambia. Connecticut: Greenwood Press; 2006.
- 172. Maimbolwa MC, Yamba B, Diwan V, Ransjo-Arvidson AB: Cultural childbirth practices and beliefs in Zambia. *J Adv Nurs* 2003, **43:**263-274.
- 173. Nsemukila BG, Phiri DS, Diallo HM, Banda SS, Benaya WK, Kitahara N: A Study of Factors
 Associated with Maternal Mortality in Zambia. Lusaka: Ministry of Health; 1998.

- Mwewa D, Michelo C: Factors associated with home deliveries in a low income rural setting-observations from Nchelenge district, Zambia. *Medical Journal of Zambia* 2010, 37:234-239.
- 175. Sialubanje C, Massar K, van der Pijl MS, Kirch EM, Hamer DH, Ruiter RA: Improving access to skilled facility-based delivery services: Women's beliefs on facilitators and barriers to the utilisation of maternity waiting homes in rural Zambia. *Reprod Health* 2015, **12:**61.
- 176. No more Traditional Birth Attendants training, Ministry of Health [https://www.lusakatimes.com/2010/12/22/traditional-birth-attendants-training-ministryhealth/]
- 177. Chi BH, Vwalika B, Killam WP, Wamalume C, Giganti MJ, Mbewe R, Stringer EM, Chintu NT,
 Putta NB, Liu KC, et al: Implementation of the Zambia electronic perinatal record system
 for comprehensive prenatal and delivery care. Int J Gynaecol Obstet 2011, 113:131-136.
- 178. Olusanya BO, Ofovwe GE: Predictors of preterm births and low birthweight in an inner-city hospital in sub-Saharan Africa. *Matern Child Health J* 2010, **14:**978-986.
- 179. Chico RM, Chandramohan D: Azithromycin plus chloroquine: combination therapy for protection against malaria and sexually transmitted infections in pregnancy. *Expert Opin Drug Metab Toxicol* 2011, 7:1153-1167.
- 180. Iriemenam NC, Shah M, Gatei W, van Eijk AM, Ayisi J, Kariuki S, Vanden Eng J, Owino SO, Lal AA, Omosun YO, et al: Temporal trends of sulphadoxine-pyrimethamine (SP) drug-resistance molecular markers in Plasmodium falciparum parasites from pregnant women in western Kenya. *Malar J* 2012, **11**:134.
- 181. Walsh MS, Hope E, Isaia L, Righarts A, Niupulusu T, Temese SV, Iosefa-Siitia L, Auvaa L,
 Tapelu SA, Motu MF, et al: Prevalence of Chlamydia trachomatis infection in Samoan
 women aged 18 to 29 and assessment of possible risk factors: a community-based study.
 Trans R Soc Trop Med Hyg 2015, 109:245-251.
- 182. Manyando C, Mkandawire R, Puma L, Sinkala M, Mpabalwani E, Njunju E, Gomes M, Ribeiro I, Walter V, Virtanen M, et al: Safety of artemether-lumefantrine in pregnant women with malaria: results of a prospective cohort study in Zambia. *Malar J* 2010, 9:249.
- 183. World Health Organization: Standard protocol for measuring efficacy of antimalarial drugs in high transmission settings. Geneva2003.
- 184. Central Statistical Office (CSO): 2010 Census of Population and Housing. pp. 12. Lusaka:Central Statistical Office; 2011:12.
- 185. Nchelenge District Health Management Team: Nchelenge District Population-2014. 2014.
- 186. Nchelenge District Health Management Team: 2014.

- 187. Ministry of Health: Zambia National Malaria Indicator Survey 2015. Lusaka, Zambia: Ministry of Health, Government of the Replic of Zambia; 2015.
- 188. Nambozi M, Malunga P, Mulenga M, Van Geertruyden JP, D'Alessandro U: Defining the malaria burden in Nchelenge District, northern Zambia using the World Health Organization malaria indicators survey. *Malar J* 2014, 13:220.
- 189. Ministry of Health: Zambia National Malaria Indicator Survey 2012. Ministry of Health, Central Statistics Office, PATH Malaria Control and Evaluation partnership in Africa (MACEPA), the United States President's Malaria Initiative, the World Bank and World Health Organisation; 2014.
- 190. Maternal, newborn, child and adolescent health [http://www.who.int/maternal_child_adolescent/epidemiology/stillbirth/en/]
- 191. Rogerson SJ, Pollina E, Getachew A, Tadesse E, Lema VM, Molyneux ME: Placental monocyte infiltrates in response to Plasmodium falciparum malaria infection and their association with adverse pregnancy outcomes. *Am J Trop Med Hyg* 2003, **68**:115-119.
- 192. Chaudhry U, Saluja D: Detection of Neisseria gonorrhoeae by PCR using orf1 gene as target. Sex Transm Infect 2002, **78**:72.
- 193. Schachter J: DFA, EIA, PCR, LCR and other technologies: what tests should be used for diagnosis of chlamydia infections? *Immunol Invest* 1997, **26**:157-161.
- 194. Mahony JB, Luinstra KE, Sellors JW, Chernesky MA: Comparison of plasmid- and chromosome-based polymerase chain reaction assays for detecting Chlamydia trachomatis nucleic acids. J Clin Microbiol 1993, 31:1753-1758.
- 195. Tam JE, Davis CH, Thresher RJ, Wyrick PB: Location of the origin of replication for the 7.5-kb Chlamydia trachomatis plasmid. *Plasmid* 1992, **27**:231-236.
- Santos C, Teixeira F, Vicente A, Astolfi-Filho S: Detection of Chlamydia trachomatis in endocervical smears of sexually active women in Manaus-AM, Brazil, by PCR. Braz J Infect Dis 2003, 7:91-95.
- 197. Kengne P, Veas F, Vidal N, Rey JL, Cuny G: **Trichomonas vaginalis: repeated DNA target for highly sensitive and specific polymerase chain reaction diagnosis.** *Cell Mol Biol (Noisy-legrand)* 1994, **40:**819-831.
- Jamali R, Zareikar R, Kazemi A, Yousefee S, Ghazanchaei A, Estakhri R, Asgharzadeh M:
 Diagnosis of Trichomonas Vaginalis Infection Using PCR Method Compared To Culture and
 Wet Mount Microscopy. Int Med J June 2006, 5.

- 199. Crucitti T, Van Dyck E, Tehe A, Abdellati S, Vuylsteke B, Buve A, Laga M: Comparison of culture and different PCR assays for detection of Trichomonas vaginalis in self collected vaginal swab specimens. *Sex Transm Infect* 2003, **79:**393-398.
- 200. Kain KC, Lanar DE: Determination of genetic variation within Plasmodium falciparum by using enzymatically amplified DNA from filter paper disks impregnated with whole blood. *J Clin Microbiol* 1991, **29**:1171-1174.
- 201. Snounou G, Viriyakosol S, Zhu XP, Jarra W, Pinheiro L, do Rosario VE, Thaithong S, Brown KN:
 High sensitivity of detection of human malaria parasites by the use of nested polymerase
 chain reaction. *Mol Biochem Parasitol* 1993, 61:315-320.
- Johnston SP, Pieniazek NJ, Xayavong MV, Slemenda SB, Wilkins PP, da Silva AJ: PCR as a confirmatory technique for laboratory diagnosis of malaria. J Clin Microbiol 2006, 44:1087-1089.
- 203. Duraisingh MT, Curtis J, Warhurst DC: Plasmodium falciparum: detection of polymorphisms in the dihydrofolate reductase and dihydropteroate synthetase genes by PCR and restriction digestion. *Exp Parasitol* 1998, **89:**1-8.
- 204. Snounou G, Beck HP: **The use of PCR genotyping in the assessment of recrudescence or reinfection after antimalarial drug treatment.** *Parasitol Today* 1998, **14:**462-467.
- 205. Snounou G, Zhu X, Siripoon N, Jarra W, Thaithong S, Brown KN, Viriyakosol S: Biased distribution of msp1 and msp2 allelic variants in Plasmodium falciparum populations in Thailand. *Trans R Soc Trop Med Hyg* 1999, **93:**369-374.
- 206. Lauritsen JM, Bruus M: EpiData (version 3.1). A comprehensive tool for validated entry and documentation of data. . Odense, Denmark: The EpiData Association; 2003-2005.
- 207. StataCorp: Stata Statistical Software: Release 13. College Station, TX: StataCorp LP; 2013.
- Braveman PA, Cubbin C, Egerter S, Chideya S, Marchi KS, Metzler M, Posner S:
 Socioeconomic status in health research: one size does not fit all. Jama 2005, 294:2879-2888.
- 209. World Health Organization: Guidelines set new definitions, update treatment for hypertension. In *Bulletin of the World Health Organization*, vol. 77 (3)1999.
- Sialubanje C, Massar K, Hamer DH, Ruiter RA: Reasons for home delivery and use of traditional birth attendants in rural Zambia: a qualitative study. *BMC Pregnancy Childbirth* 2015, 15:216.
- 211. Cohee LM, Kalilani-Phiri L, Boudova S, Joshi S, Mukadam R, Seydel KB, Mawindo P, Thesing P, Kamiza S, Makwakwa K, et al: Submicroscopic malaria infection during pregnancy and the impact of intermittent preventive treatment. *Malar J* 2014, **13**:274.

- 212. Farcas GA, Zhong KJ, Mazzulli T, Kain KC: **Evaluation of the RealArt Malaria LC real-time PCR** assay for malaria diagnosis. *J Clin Microbiol* 2004, **42:**636-638.
- 213. Kamuliwo M, Kirk KE, Chanda E, Elbadry MA, Lubinda J, Weppelmann TA, Mukonka VM, Zhang W, Mushinge G, Mwanza-Ingwe M, Haque U: Spatial patterns and determinants of malaria infection during pregnancy in Zambia. *Trans R Soc Trop Med Hyg* 2015, 109:514-521.
- Steketee RW, Wirima JJ, Bloland PB, Chilima B, Mermin JH, Chitsulo L, Breman JG:
 Impairment of a pregnant woman's acquired ability to limit Plasmodium falciparum by
 infection with human immunodeficiency virus type-1. Am J Trop Med Hyg 1996, 55:42-49.
- 215. ter Kuile FO, Parise ME, Verhoeff FH, Udhayakumar V, Newman RD, van Eijk AM, Rogerson
 SJ, Steketee RW: The burden of co-infection with human immunodeficiency virus type 1
 and malaria in pregnant women in sub-saharan Africa. Am J Trop Med Hyg 2004, 71:41-54.
- Ladner J, Leroy V, Simonon A, Karita E, Bogaerts J, De Clercq A, Van De Perre P, Dabis F,
 Pregnancy, Group HIVS: HIV infection, malaria, and pregnancy: a prospective cohort study
 in Kigali, Rwanda. Am J Trop Med Hyg 2002, 66:56-60.
- 217. Parise ME, Ayisi JG, Nahlen BL, Schultz LJ, Roberts JM, Misore A, Muga R, Oloo AJ, Steketee RW: Efficacy of sulfadoxine-pyrimethamine for prevention of placental malaria in an area of Kenya with a high prevalence of malaria and human immunodeficiency virus infection. *Am J Trop Med Hyg* 1998, **59:**813-822.
- 218. Steketee RW, Wirima JJ, Slutsker L, Breman JG, Heymann DL: Comparability of treatment groups and risk factors for parasitemia at the first antenatal clinic visit in a study of malaria treatment and prevention in pregnancy in rural Malawi. Am J Trop Med Hyg 1996, 55:17-23.
- 219. Ticconi C, Mapfumo M, Dorrucci M, Naha N, Tarira E, Pietropolli A, Rezza G: Effect of maternal HIV and malaria infection on pregnancy and perinatal outcome in Zimbabwe. J Acquir Immune Defic Syndr 2003, 34:289-294.
- van Eijk AM, Ayisi JG, ter Kuile FO, Misore AO, Otieno JA, Rosen DH, Kager PA, Steketee RW,
 Nahlen BL: HIV increases the risk of malaria in women of all gravidities in Kisumu, Kenya.
 Aids 2003, 17:595-603.
- 221. Msuya SE, Uriyo J, Hussain A, Mbizvo EM, Jeansson S, Sam NE, Stray-Pedersen B: Prevalence of sexually transmitted infections among pregnant women with known HIV status in northern Tanzania. *Reprod Health* 2009, 6:4.

- 222. Cohen CR, Duerr A, Pruithithada N, Rugpao S, Hillier S, Garcia P, Nelson K: Bacterial vaginosis and HIV seroprevalence among female commercial sex workers in Chiang Mai, Thailand. *AIDS* 1995, **9**:1093-1097.
- 223. Fonck K, Kaul R, Keli F, Bwayo JJ, Ngugi EN, Moses S, Temmerman M: Sexually transmitted infections and vaginal douching in a population of female sex workers in Nairobi, Kenya. *Sex Transm Infect* 2001, **77:**271-275.
- Myer L, Denny L, Telerant R, Souza M, Wright TC, Jr., Kuhn L: Bacterial vaginosis and susceptibility to HIV infection in South African women: a nested case-control study. J Infect Dis 2005, 192:1372-1380.
- 225. Myer L, Kuhn L, Stein ZA, Wright TC, Jr., Denny L: Intravaginal practices, bacterial vaginosis, and women's susceptibility to HIV infection: epidemiological evidence and biological mechanisms. *Lancet Infect Dis* 2005, **5:**786-794.
- Taha TE, Hoover DR, Dallabetta GA, Kumwenda NI, Mtimavalye LA, Yang LP, Liomba GN,
 Broadhead RL, Chiphangwi JD, Miotti PG: Bacterial vaginosis and disturbances of vaginal
 flora: association with increased acquisition of HIV. AIDS 1998, 12:1699-1706.
- 227. Chesson HW, Blandford JM, Pinkerton SD: Estimates of the annual number and cost of new HIV infections among women attributable to trichomoniasis in the United States. Sex Transm Dis 2004, 31:547-551.
- 228. Lazenby GB: Trichomonas vaginalis screening and prevention in order to impact the HIV pandemic: Isn't it time we take this infection seriously? *Infect Dis Rep* 2011, **3:**e4.
- 229. McClelland RS, Sangare L, Hassan WM, Lavreys L, Mandaliya K, Kiarie J, Ndinya-Achola J, Jaoko W, Baeten JM: Infection with Trichomonas vaginalis increases the risk of HIV-1 acquisition. J Infect Dis 2007, 195:698-702.
- 230. Rottingen JA, Cameron DW, Garnett GP: A systematic review of the epidemiologic interactions between classic sexually transmitted diseases and HIV: how much really is known? Sex Transm Dis 2001, 28:579-597.
- 231. Shafir SC, Sorvillo FJ, Smith L: Current issues and considerations regarding trichomoniasis and human immunodeficiency virus in African-Americans. *Clin Microbiol Rev* 2009, 22:37-45, Table of Contents.
- Sorvillo F, Smith L, Kerndt P, Ash L: Trichomonas vaginalis, HIV, and African-Americans.
 Emerg Infect Dis 2001, 7:927-932.
- 233. Laga M, Manoka A, Kivuvu M, Malele B, Tuliza M, Nzila N, Goeman J, Behets F, Batter V,
 Alary M, et al.: Non-ulcerative sexually transmitted diseases as risk factors for HIV-1
 transmission in women: results from a cohort study. AIDS 1993, 7:95-102.

- Plummer FA, Simonsen JN, Cameron DW, Ndinya-Achola JO, Kreiss JK, Gakinya MN, Waiyaki
 P, Cheang M, Piot P, Ronald AR, et al.: Cofactors in male-female sexual transmission of
 human immunodeficiency virus type 1. J Infect Dis 1991, 163:233-239.
- 235. Kapiga SH, Shao JF, Lwihula GK, Hunter DJ: Risk factors for HIV infection among women in
 Dar-es-Salaam, Tanzania. J Acquir Immune Defic Syndr 1994, 7:301-309.
- 236. de Vincenzi I: A longitudinal study of human immunodeficiency virus transmission by
 heterosexual partners. European Study Group on Heterosexual Transmission of HIV. N Engl
 J Med 1994, 331:341-346.
- 237. McClelland RS, Wang CC, Mandaliya K, Overbaugh J, Reiner MT, Panteleeff DD, Lavreys L, Ndinya-Achola J, Bwayo JJ, Kreiss JK: **Treatment of cervicitis is associated with decreased** cervical shedding of HIV-1. *Aids* 2001, 15:105-110.
- 238. Wang CC, McClelland RS, Reilly M, Overbaugh J, Emery SR, Mandaliya K, Chohan B, Ndinya-Achola J, Bwayo J, Kreiss JK: The effect of treatment of vaginal infections on shedding of human immunodeficiency virus type 1. J Infect Dis 2001, 183:1017-1022.
- 239. Rees H, Holmes K: The STI vaccine roadmap-a long overdue intervention. Vaccine 2014,
 32:1638-1639.
- 240. Menendez C: Malaria during pregnancy: a priority area of malaria research and control. *Parasitol Today* 1995, **11**:178-183.
- 241. Walker-Abbey A, Djokam RR, Eno A, Leke RF, Titanji VP, Fogako J, Sama G, Thuita LH, Beardslee E, Snounou G, et al: Malaria in pregnant Cameroonian women: the effect of age and gravidity on submicroscopic and mixed-species infections and multiple parasite genotypes. Am J Trop Med Hyg 2005, 72:229-235.
- Mukonka VM, Chanda E, Haque U, Kamuliwo M, Mushinge G, Chileshe J, Chibwe KA, Norris DE, Mulenga M, Chaponda M, et al: High burden of malaria following scale-up of control interventions in Nchelenge District, Luapula Province, Zambia. *Malar J* 2014, 13:153.
- 243. Wasserheit JN: Epidemiological synergy. Interrelationships between human immunodeficiency virus infection and other sexually transmitted diseases. Sex Transm Dis 1992, 19:61-77.
- Zuma K, Gouws E, Williams B, Lurie M: Risk factors for HIV infection among women in
 Carletonville, South Africa: migration, demography and sexually transmitted diseases. Int J
 STD AIDS 2003, 14:814-817.
- 245. Klinger EV, Kapiga SH, Sam NE, Aboud S, Chen CY, Ballard RC, Larsen U: A Community-based study of risk factors for Trichomonas vaginalis infection among women and their male partners in Moshi urban district, northern Tanzania. *Sex Transm Dis* 2006, **33**:712-718.

- Patterson TL, Semple SJ, Staines H, Lozada R, Orozovich P, Bucardo J, Philbin MM, Pu M,
 Fraga M, Amaro H, et al: Prevalence and correlates of HIV infection among female sex
 workers in 2 Mexico-US border cities. J Infect Dis 2008, 197:728-732.
- 247. Dallabetta GA, Miotti PG, Chiphangwi JD, Saah AJ, Liomba G, Odaka N, Sungani F, Hoover DR:
 High socioeconomic status is a risk factor for human immunodeficiency virus type 1 (HIV-1)
 infection but not for sexually transmitted diseases in women in Malawi: implications for
 HIV-1 control. J Infect Dis 1993, 167:36-42.
- 248. Ness RB, Kip KE, Soper DE, Hillier S, Stamm CA, Sweet RL, Rice P, Richter HE: Bacterial vaginosis (BV) and the risk of incident gonococcal or chlamydial genital infection in a predominantly black population. *Sex Transm Dis* 2005, **32:**413-417.
- Wiesenfeld HC, Hillier SL, Krohn MA, Landers DV, Sweet RL: Bacterial vaginosis is a strong predictor of Neisseria gonorrhoeae and Chlamydia trachomatis infection. *Clin Infect Dis* 2003, 36:663-668.
- 250. Martin HL, Richardson BA, Nyange PM, Lavreys L, Hillier SL, Chohan B, Mandaliya K, Ndinya-Achola JO, Bwayo J, Kreiss J: Vaginal lactobacilli, microbial flora, and risk of human immunodeficiency virus type 1 and sexually transmitted disease acquisition. J Infect Dis 1999, 180:1863-1868.
- 251. Moodley P, Connolly C, Sturm AW: Interrelationships among human immunodeficiency virus type 1 infection, bacterial vaginosis, trichomoniasis, and the presence of yeasts. J Infect Dis 2002, 185:69-73.
- 252. Guedou FA, Van Damme L, Mirembe F, Solomon S, Becker M, Deese J, Crucitti T, Alary M: Intermediate vaginal flora is associated with HIV prevalence as strongly as bacterial vaginosis in a cross-sectional study of participants screened for a randomised controlled trial. Sex Transm Infect 2012, 88:545-551.
- 253. Sewankambo N, Gray RH, Wawer MJ, Paxton L, McNaim D, Wabwire-Mangen F, Serwadda D,
 Li C, Kiwanuka N, Hillier SL, et al: HIV-1 infection associated with abnormal vaginal flora
 morphology and bacterial vaginosis. Lancet 1997, 350:546-550.
- Jamieson DJ, Duerr A, Klein RS, Paramsothy P, Brown W, Cu-Uvin S, Rompalo A, Sobel J:
 Longitudinal analysis of bacterial vaginosis: findings from the HIV epidemiology research study. Obstet Gynecol 2001, 98:656-663.
- Warren D, Klein RS, Sobel J, Kieke B, Jr., Brown W, Schuman P, Anderson J, Cu-Uvin S, Mayer
 K, Jamieson DJ, et al: A multicenter study of bacterial vaginosis in women with or at risk for
 human immunodeficiency virus infection. *Infect Dis Obstet Gynecol* 2001, 9:133-141.

- 256. Wangnapi RA, Soso S, Unger HW, Sawera C, Ome M, Umbers AJ, Ndrewei N, Siba P, Li Wai Suen CS, Vallely A, et al: Prevalence and risk factors for Chlamydia trachomatis, Neisseria gonorrhoeae and Trichomonas vaginalis infection in pregnant women in Papua New Guinea. Sex Transm Infect 2015, 91:194-200.
- 257. Kapiga S, Kelly C, Weiss S, Daley T, Peterson L, Leburg C, Ramjee G: Risk factors for incidence of sexually transmitted infections among women in South Africa, Tanzania, and Zambia: results from HPTN 055 study. Sex Transm Dis 2009, 36:199-206.
- 258. Mabey DC, Sollis KA, Kelly HA, Benzaken AS, Bitarakwate E, Changalucha J, Chen XS, Yin YP, Garcia PJ, Strasser S, et al: Point-of-care tests to strengthen health systems and save newborn lives: the case of syphilis. PLoS Med 2012, 9:e1001233.
- 259. World Health Organization: The use of Rapid Syphilis Tests. Geneva: WHO; 2006.
- 260. Chico RM, Moss WJ: Prevention of malaria in pregnancy: a fork in the road? *Lancet* 2015.
- 261. Gutman J, Mwandama D, Wiegand RE, Abdallah J, Iriemenam NC, Shi YP, Mathanga DP, Skarbinski J: In vivo efficacy of sulphadoxine-pyrimethamine for the treatment of asymptomatic parasitaemia in pregnant women in Machinga District, Malawi. Malar J 2015, 14:197.
- 262. Cattamanchi A, Kyabayinze D, Hubbard A, Rosenthal PJ, Dorsey G: Distinguishing recrudescence from reinfection in a longitudinal antimalarial drug efficacy study: comparison of results based on genotyping of msp-1, msp-2, and glurp. *Am J Trop Med Hyg* 2003, 68:133-139.
- 263. Clerk CA, Bruce J, Affipunguh PK, Mensah N, Hodgson A, Greenwood B, Chandramohan D: A randomized, controlled trial of intermittent preventive treatment with sulfadoxine-pyrimethamine, amodiaquine, or the combination in pregnant women in Ghana. J Infect Dis 2008, 198:1202-1211.
- 264. Kalilani L, Mofolo I, Chaponda M, Rogerson SJ, Meshnick SR: **The effect of timing and** frequency of Plasmodium falciparum infection during pregnancy on the risk of low birth weight and maternal anemia. *Trans R Soc Trop Med Hyg* 2010, **104**:416-422.
- 265. Ekeus C, Lindblad F, Hjern A: Short stature, smoking habits and birth outcome in international adoptees in Sweden. *Acta Obstet Gynecol Scand* 2008, **87:**1309-1314.
- 266. Zhang X, Mumford SL, Cnattingius S, Schisterman EF, Kramer MS: **Reduced birthweight in short or primiparous mothers: physiological or pathological?** *Bjog* 2010, **117:**1248-1254.
- 267. Kozuki N, Lee AC, Silveira MF, Sania A, Vogel JP, Adair L, Barros F, Caulfield LE, Christian P,Fawzi W, et al: The associations of parity and maternal age with small-for-gestational-age,

preterm, and neonatal and infant mortality: a meta-analysis. *BMC Public Health* 2013, **13** Suppl 3:S2.

- 268. Wilcox MA, Chang AM, Johnson IR: **The effects of parity on birthweight using successive** pregnancies. *Acta Obstet Gynecol Scand* 1996, **75:**459-453.
- 269. Desai M, Gutman J, L'Lanziva A, Otieno K, Juma E, Kariuki S, Ouma P, Were V, Laserson K, Katana A, et al: Intermittent screening and treatment or intermittent preventive treatment with dihydroartemisinin-piperaquine versus intermittent preventive treatment with sulfadoxine-pyrimethamine for the control of malaria during pregnancy in western Kenya: an open-label, three-group, randomised controlled superiority trial. *Lancet* 2015, 386:2507-2519.
- Zucker JR, Lackritz EM, Ruebush TK, Hightower AW, Adungosi JE, Were JB, Campbell CC:
 Anaemia, blood transfusion practices, HIV and mortality among women of reproductive age in western Kenya. *Trans R Soc Trop Med Hyg* 1994, 88:173-176.
- 271. Aaron E, Bonacquisti A, Mathew L, Alleyne G, Bamford LP, Culhane JF: Small-for-gestationalage births in pregnant women with HIV, due to severity of HIV disease, not antiretroviral therapy. Infect Dis Obstet Gynecol 2012, 2012:135030.
- Ayisi JG, van Eijk AM, ter Kuile FO, Kolczak MS, Otieno JA, Misore AO, Kager PA, Steketee
 RW, Nahlen BL: The effect of dual infection with HIV and malaria on pregnancy outcome in
 western Kenya. Aids 2003, 17:585-594.
- 273. Chen JY, Ribaudo HJ, Souda S, Parekh N, Ogwu A, Lockman S, Powis K, Dryden-Peterson S,
 Creek T, Jimbo W, et al: Highly Active Antiretroviral Therapy and Adverse Birth Outcomes
 Among HIV-Infected Women in Botswana. J Infect Dis 2012, 206:1695-1705.
- 274. Ekouevi DK, Coffie PA, Becquet R, Tonwe-Gold B, Horo A, Thiebaut R, Leroy V, Blanche S,
 Dabis F, Abrams EJ: Antiretroviral therapy in pregnant women with advanced HIV disease
 and pregnancy outcomes in Abidjan, Cote d'Ivoire. *Aids* 2008, 22:1815-1820.
- 275. van der Merwe K, Hoffman R, Black V, Chersich M, Coovadia A, Rees H: Birth outcomes in South African women receiving highly active antiretroviral therapy: a retrospective observational study. J Int AIDS Soc 2011, 14:42.
- 276. Li N, Sando MM, Spiegelman D, Hertzmark E, Liu E, Sando D, Machumi L, Chalamilla G, Fawzi
 W: Antiretroviral Therapy in Relation to Birth Outcomes among HIV-infected Women: A
 Cohort Study. J Infect Dis 2015.
- 277. Kesho Bora Study G, de Vincenzi I: Triple antiretroviral compared with zidovudine and single-dose nevirapine prophylaxis during pregnancy and breastfeeding for prevention of

mother-to-child transmission of HIV-1 (Kesho Bora study): a randomised controlled trial. Lancet Infect Dis 2011, **11**:171-180.

- 278. Szyld EG, Warley EM, Freimanis L, Gonin R, Cahn PE, Calvet GA, Duarte G, Melo VH, Read JS, Group NPS: Maternal antiretroviral drugs during pregnancy and infant low birth weight and preterm birth. *Aids* 2006, **20**:2345-2353.
- 279. Marazzi MC, Palombi L, Nielsen-Saines K, Haswell J, Zimba I, Magid NA, Buonomo E, Scarcella P, Ceffa S, Paturzo G, et al: Extended antenatal use of triple antiretroviral therapy for prevention of mother-to-child transmission of HIV-1 correlates with favorable pregnancy outcomes. *Aids* 2011, 25:1611-1618.
- Harrington WE, Mutabingwa TK, Muehlenbachs A, Sorensen B, Bolla MC, Fried M, Duffy PE:
 Competitive facilitation of drug-resistant Plasmodium falciparum malaria parasites in
 pregnant women who receive preventive treatment. *Proc Natl Acad Sci U S A* 2009,
 106:9027-9032.
- 281. Moussiliou A, De Tove YS, Doritchamou J, Luty AJ, Massougbodji A, Alifrangis M, Deloron P, Ndam NT: High rates of parasite recrudescence following intermittent preventive treatment with sulphadoxine-pyrimethamine during pregnancy in Benin. *Malar J* 2013, 12:195.
- 282. Diakite OS, Kayentao K, Traore BT, Djimde A, Traore B, Diallo M, Ongoiba A, Doumtabe D, Doumbo S, Traore MS, et al: Superiority of 3 over 2 doses of intermittent preventive treatment with sulfadoxine-pyrimethamine for the prevention of malaria during pregnancy in mali: a randomized controlled trial. *Clin Infect Dis* 2011, 53:215-223.
- 283. Chico RM, Hack BB, Newport MJ, Ngulube E, Chandramohan D: On the pathway to better birth outcomes? A systematic review of azithromycin and curable sexually transmitted infections. Expert Rev Anti Infect Ther 2013, 11:1303-1332.
- 284. Luntamo M, Kulmala T, Cheung YB, Maleta K, Ashorn P: **The effect of antenatal monthly** sulphadoxine-pyrimethamine, alone or with azithromycin, on foetal and neonatal growth faltering in Malawi: a randomised controlled trial. *Trop Med Int Health* 2013, **18**:386-397.
- 285. Caughey AB, Nicholson JM, Washington AE: First- vs second-trimester ultrasound: the effect on pregnancy dating and perinatal outcomes. *Am J Obstet Gynecol* 2008, **198**:703.e701-705; discussion 703.e705-706.
- 286. Desai M, Gutman J, Taylor SM, Wiegand RE, Khairallah C, Kayentao K, Ouma P, Coulibaly SO, Kalilani L, Mace KE, et al: Impact of Sulfadoxine-Pyrimethamine Resistance on Effectiveness of Intermittent Preventive Therapy for Malaria in Pregnancy at Clearing Infections and Preventing Low Birth Weight. *Clin Infect Dis* 2015.

- 287. Kakuru A, Jagannathan P, Muhindo MK, Natureeba P, Awori P, Nakalembe M, Opira B,
 Olwoch P, Ategeka J, Nayebare P, et al: Dihydroartemisinin-Piperaquine for the Prevention
 of Malaria in Pregnancy. N Engl J Med 2016, 374:928-939.
- 288. Agger B: Critical Thoery, Poststructuralism, Post modernisation, Their Sociological Relevence. Annual Reviews Sociology. pp. 105-5311991:105-531.
- 289. Creswell JW: *Qualitative Inquiry and Research Design: Choosing among the five approaches.*3rd Edition edn. London: Sage Publication Inc.; 2013.
- 290. Delanty G: *Modernity and Post structuralism Part II: In: Harrington, A. ed. Modern Social Theory.* New York: Oxford University Press Inc.; 2005.
- 291. Creswell JW: *Research Design: Qualitative, Quantitative and mixed Method Approaches.* Thousand Oaks: Sage Publication Inc.; 2004.
- 292. Petersen A: Research on Men masculinities: Some implications of Recent Theory Future Work. In *Men masculinities*, vol. 52003.
- 293. Verhoeff FH, Brabin BJ, Chimsuku L, Kazembe P, Russell WB, Broadhead RL: An evaluation of the effects of intermittent sulfadoxine-pyrimethamine treatment in pregnancy on parasite clearance and risk of low birthweight in rural Malawi. Ann Trop Med Parasitol 1998, 92:141-150.
- Verhoeff FH, Brabin BJ, Chimsuku L, Kazembe P, Broadhead RL: Malaria in pregnancy and its consequences for the infant in rural Malawi. Ann Trop Med Parasitol 1999, 93 Suppl 1:S25-33.
- 295. Mnyika KS, Kabalimu TK, Rukinisha K, Mpanju-Shumbusho W: Randomised trial of alternative malaria chemoprophylaxis strategies among pregnant women in Kigoma, Tanzania: I. Rationale and design. *East Afr Med J* 2000, 77:98-104.
- Saute F, Menendez C, Mayor A, Aponte J, Gomez-Olive X, Dgedge M, Alonso P: Malaria in pregnancy in rural Mozambique: the role of parity, submicroscopic and multiple
 Plasmodium falciparum infections. *Trop Med Int Health* 2002, 7:19-28.
- 297. Mankhambo L, Kanjala M, Rudman S, Lema VM, Rogerson SJ: Evaluation of the OptiMAL rapid antigen test and species-specific PCR to detect placental Plasmodium falciparum infection at delivery. *J Clin Microbiol* 2002, **40:**155-158.
- 298. Montoya PJ, Lukehart SA, Brentlinger PE, Blanco AJ, Floriano F, Sairosse J, Gloyd S:
 Comparison of the diagnostic accuracy of a rapid immunochromatographic test and the rapid plasma reagin test for antenatal syphilis screening in Mozambique. Bull World Health Organ 2006, 84:97-104.

- 299. Brentlinger PE, Dgedge M, Correia MA, Rojas AJ, Saute F, Gimbel-Sherr KH, Stubbs BA, Mercer MA, Gloyd S: Intermittent preventive treatment of malaria during pregnancy in central Mozambique. Bull World Health Organ 2007, 85:873-879.
- Lujan J, de Onate WA, Delva W, Claeys P, Sambola F, Temmerman M, Fernando J, Folgosa E:
 Prevalence of sexually transmitted infections in women attending antenatal care in Tete
 province, Mozambique. S Afr Med J 2008, 98:49-51.
- 301. van den Broek NR, White SA, Goodall M, Ntonya C, Kayira E, Kafulafula G, Neilson JP: The
 APPLe study: a randomized, community-based, placebo-controlled trial of azithromycin for
 the prevention of preterm birth, with meta-analysis. *PLoS Med* 2009, 6:e1000191.
- Mayor A, Moro L, Aguilar R, Bardaji A, Cistero P, Serra-Casas E, Sigauque B, Alonso PL, Ordi J, Menendez C: How Hidden Can Malaria Be in Pregnant Women? Diagnosis by Microscopy, Placental Histology, Polymerase Chain Reaction and Detection of Histidine-Rich Protein 2 in Plasma. *Clin Infect Dis* 2012.
- 303. Meuris S, Piko BB, Eerens P, Vanbellinghen AM, Dramaix M, Hennart P: **Gestational malaria:** assessment of its consequences on fetal growth. *Am J Trop Med Hyg* 1993, **48**:603-609.
- 304. Lukuka KA, Fumie OS, Mulumbu MR, Lokombe BJ, Muyembe TJ: [Malaria prevalence at delivery in four maternity hospitals of Kinshasa City, Democratic Republic of Congo]. Bull Soc Pathol Exot 2006, 99:200-201.
- 305. Senga E, Loscertales MP, Makwakwa KE, Liomba GN, Dzamalala C, Kazembe PN, Brabin BJ:
 ABO blood group phenotypes influence parity specific immunity to Plasmodium falciparum
 malaria in Malawian women. *Malar J* 2007, 6:102.
- 306. Mwangoka GW, Kimera SI, Mboera LE: **Congenital Plasmodium falciparum infection in neonates in Muheza District, Tanzania.** *Malar J* 2008, **7:**117.
- 307. Vuylsteke B, Bastos R, Barreto J, Crucitti T, Folgosa E, Mondlane J, Dusauchoit T, Piot P, Laga
 M: High prevalence of sexually transmitted diseases in a rural area in Mozambique.
 Genitourin Med 1993, 69:427-430.
- 308. Rutgers S: Syphilis in pregnancy: a medical audit in a rural district. *Cent Afr J Med* 1993,
 39:248-253.
- 309. Mayaud P, Msuya W, Todd J, Kaatano G, West B, Begkoyian G, Grosskurth H, Mabey D: **STD** rapid assessment in Rwandan refugee camps in Tanzania. *Genitourin Med* 1997, **73:**33-38.
- 310. Mayaud P, Mosha F, Todd J, Balira R, Mgara J, West B, Rusizoka M, Mwijarubi E, Gabone R, Gavyole A, et al: Improved treatment services significantly reduce the prevalence of sexually transmitted diseases in rural Tanzania: results of a randomized controlled trial. *AIDS* 1997, **11**:1873-1880.

- Mayaud P, Uledi E, Cornelissen J, ka-Gina G, Todd J, Rwakatare M, West B, Kopwe L, Manoko
 D, Grosskurth H, et al: Risk scores to detect cervical infections in urban antenatal clinic
 attenders in Mwanza, Tanzania. Sex Transm Infect 1998, 74 Suppl 1:S139-146.
- 312. Taha TE, Dallabetta GA, Hoover DR, Chiphangwi JD, Mtimavalye LA, Liomba GN, Kumwenda NI, Miotti PG: Trends of HIV-1 and sexually transmitted diseases among pregnant and postpartum women in urban Malawi. AIDS 1998, 12:197-203.
- Taha TE, Gray RH, Kumwenda NI, Hoover DR, Mtimavalye LA, Liomba GN, Chiphangwi JD,
 Dallabetta GA, Miotti PG: HIV infection and disturbances of vaginal flora during pregnancy.
 J Acquir Immune Defic Syndr Hum Retrovirol 1999, 20:52-59.
- 314. Kambarami RA, Manyame B, Macq J: **Syphilis in Murewa District, Zimbabwe: an old** problem that rages on. *Cent Afr J Med* 1998, **44**:229-232.
- 315. Majoko F, Munjanja S, Nystrom L, Mason E, Lindmark G: Field efficiency of syphilis screening in antenatal care: lessons from Gutu District in Zimbabwe. *Cent Afr J Med* 2003, 49:90-93.
- 316. Romoren M, Rahman M: Syphilis screening in the antenatal care: a cross-sectional study from Botswana. *BMC Int Health Hum Rights* 2006, 6:8.
- 317. Central Statistical Office (CSO) MoHM, Tropical Diseases Research Centre (TDRC),, University of Zambia aMII: Zambia Demographic and Health Survey, 2007 Calverton, Maryland, USA:
 CSO and Macro International Inc.; 2009.
- 318. Kinoshita-Moleka R, Smith JS, Atibu J, Tshefu A, Hemingway-Foday J, Hobbs M, Bartz J, Koch MA, Rimoin AW, Ryder RW: Low prevalence of HIV and other selected sexually transmitted infections in 2004 in pregnant women from Kinshasa, the Democratic Republic of the Congo. *Epidemiol Infect* 2008, **136**:1290-1296.
- 319. Menendez C, Bardaji A, Sigauque B, Romagosa C, Sanz S, Serra-Casas E, Macete E, Berenguera A, David C, Dobano C, et al: A randomized placebo-controlled trial of intermittent preventive treatment in pregnant women in the context of insecticide treated nets delivered through the antenatal clinic. *PLoS One* 2008, 3:e1934.
- Mapingure MP, Msuya S, Kurewa NE, Munjoma MW, Sam N, Chirenje MZ, Rusakaniko S, Saugstad LF, de Vlas SJ, Stray-Pedersen B: Sexual behaviour does not reflect HIV-1
 prevalence differences: a comparison study of Zimbabwe and Tanzania. J Int AIDS Soc 2010, 13:45.
- 321. Taylor MM, Ebrahim S, Abiola N, Kinkodi DK, Mpingulu M, Kabuayi JP, Ekofo F, Newman DR, Peterman TA, Kamb ML, Sidibe K: **Correlates of syphilis seropositivity and risk for syphilis**-

associated adverse pregnancy outcomes among women attending antenatal care clinics in the Democratic Republic of Congo. *Int J STD AIDS* 2014, **25**:716-725.

- Chiduo M, Theilgaard ZP, Bakari V, Mtatifikolo F, Bygbjerg I, Flanholc L, Gerstoft J,
 Christiansen CB, Lemnge M, Katzenstein TL: Prevalence of sexually transmitted infections
 among women attending antenatal clinics in Tanga, north eastern Tanzania. Int J STD AIDS 2012, 23:325-329.
- Bonawitz RE, Duncan J, Hammond E, Hamomba L, Nambule J, Sambambi K, Musonda V,
 Calise A, Knapp A, Mwale J, et al: Assessment of the impact of rapid syphilis tests on
 syphilis screening and treatment of pregnant women in Zambia. Int J Gynaecol Obstet
 2015, 130 Suppl 1:S58-62.
- 324. Vuylsteke B, Laga M, Alary M, Gerniers MM, Lebughe JP, Nzila N, Behets F, Van Dyck E, Piot
 P: Clinical algorithms for the screening of women for gonococcal and chlamydial infection:
 evaluation of pregnant women and prostitutes in Zaire. Clin Infect Dis 1993, 17:82-88.
- 325. Romoren M, Rahman M, Sundby J, Hjortdahl P: **Chlamydia and gonorrhoea in pregnancy:** effectiveness of diagnosis and treatment in Botswana. *Sex Transm Infect* 2004, **80:**395-400.
- 326. Romoren M, Velauthapillai M, Rahman M, Sundby J, Klouman E, Hjortdahl P: **Trichomoniasis** and bacterial vaginosis in pregnancy: inadequately managed with the syndromic approach. *Bull World Health Organ* 2007, **85:**297-304.
- 327. Crucitti T, Jespers V, Mulenga C, Khondowe S, Vandepitte J, Buve A: Trichomonas vaginalis is
 Highly Prevalent in Adolescent Girls, Pregnant Women, and Commercial Sex Workers in
 Ndola, Zambia. Sex Transm Dis 2009.
- 328. Tolosa JE, Chaithongwongwatthana S, Daly S, Maw WW, Gaitan H, Lumbiganon P, Festin M, Chipato T, Sauvarin J, Goldenberg RL, et al: **The International Infections in Pregnancy (IIP) study: variations in the prevalence of bacterial vaginosis and distribution of morphotypes in vaginal smears among pregnant women.** *Am J Obstet Gynecol* 2006, **195**:1198-1204.
- 329. Shayo PA, Kihunrwa A, Massinde AN, Mirambo M, Rumanyika R, Ngwalida N, Gumodoka B, Kidola J, Magoma M: Prevalence of bacterial vaginosis and associated factors among pregnant women attending at Bugando Medical Centre, Mwanza, Tanzania. Tanzan J Health Res 2012, 14:175-182.
- 330. Mbopi Keou FX, Mbu R, Mauclere P, Andela A, Tetanye E, Leke R, Chaouat G, Barre-Sinoussi F, Martin P, Belec L: Antenatal HIV prevalence in Yaounde, Cameroon. Int J STD AIDS 1998, 9:400-402.
- 331. Gaydos CA: Nucleic acid amplification tests for gonorrhea and chlamydia: practice and applications. Infect Dis Clin North Am 2005, 19:367-386, ix.

- 332. Cook RL, Hutchison SL, Ostergaard L, Braithwaite RS, Ness RB: Systematic review:
 noninvasive testing for Chlamydia trachomatis and Neisseria gonorrhoeae. Ann Intern Med
 2005, 142:914-925.
- 333. Meehan M, Wawer M, Swerwadda D, Gray R, Quinn T: *Laboratory methods for the diagnosis* of reproductive tract infections and selected conditions in population-based studies. In: jejeeboy S, Koenig M, Elias C, eds. Reproductive Tract Infections and other Gynaecological Disorders. Cambridge: Cambridge University Press; 2003.
- 334. Madico G, Quinn TC, Rompalo A, McKee KT, Jr., Gaydos CA: **Diagnosis of Trichomonas** vaginalis infection by PCR using vaginal swab samples. *J Clin Microbiol* 1998, **36:**3205-3210.
- 335. Mockenhaupt FP, Ulmen U, von Gaertner C, Bedu-Addo G, Bienzle U: Diagnosis of placental malaria. J Clin Microbiol 2002, **40:**306-308.
- 336. Msuya SE, Mbizvo EM, Stray-Pedersen B, Uriyo J, Sam NE, Rusakaniko S, Hussain A: Decline in HIV prevalence among women of childbearing age in Moshi urban, Tanzania. Int J STD AIDS 2007, 18:680-687.
- 337. Larkin GL, Thuma PE: Congenital malaria in a hyperendemic area. Am J Trop Med Hyg 1991,
 45:587-592.
9.0 Appendices

Appendix 1.1a: Search terms for studies reporting malaria and/or STIs/RTIs and birth outcomes

Organism/infection	Medical subject heading terms	Free test terms
Treponema pallidum	Treponema pallidum OR syphilis AND "	Treponema pallidum and birth outcome
, ,	hirth outcome"	Treponema pallidum and pregnancy outcome
	birtir bateonie	Syphilis and birth outcome
		Syphilis and pregnancy outcome
		Treponema pallidum and stillbirth
		Treponema pallidum and IUGR
		Treponema pallidum and Preterm delivery
		Treponema pallidum and low birth weight
		Treponema pallidum and PROM
		Syphilis and stillbirth
		Syphilis and IUGR
		Syphilis and Preterm delivery
		Syphilis and low birth weight
Neisseria aonorrhoea	Neisseria gonorrhoege OR Gonorrhoea	Neisseria gonorrhoeae and birth outcome
	OB Concernhood AND " hirth outcome"	Gonorrhoea and birth outcome
	OR GOHOITHOEA AND DIT IT OULCOME	Gonorrhea and birth outcome
		Neisseria gonorrhoeae and pregnancy outcome
		Gonorrhoea and pregnancy outcome
		Gonorrhea and pregnancy outcome
		Neisseria gonorrhoeae and stillbirth
		Gonorrhoea and stillbirth
		Gonorrhea and stillbirth
		Neisseria gonorrhoeae and IUGR
		Gonorrhoea and IUGR
		Neisseria gonorrhoeae and Preterm delivery
		Gonorrhoea and Preterm delivery
		Neisseria gonorrhoeae and low birth weight
		Gonorrhoea and low birth weight
		Gonorrhea and low birth weight
		Neisseria gonorrhoeae and PROM
		Gonorrhoea and PROM
		Gonorrhea and PROM
Chlamydia trachomatis	Chlamydia trachomatis AND " birth	Chlamydia trachomatis and birth outcome
	outcome"	Chlamydia trachomatis and pregnancy outcome
	outcome	Chlamydia trachomatis and stillbirth
		Chlamydia trachomatis and IUGR
		Chlamydia trachomatis and Preterm delivery
		Chlamydia trachomatis and low birth weight
		Chlamydia trachomatis and stillbirth
		Chlamydia trachomatis and PROM

Α	pr	ber	dix	(1	.1a:	: Se	earc	h t	erm	s fo	or st	udie	es re	epo	ortir	lg I	mala	ria	and	/or	STIS	s/R	Γls	and	birt	th d	outc	omes	conti	nued
	ГГ			· - ·					•••••	• • •						. סי			~,	•••	• • • •					••••		0		

Organism/infection	Medical subject heading terms	Free text terms				
Trichomonas vaainalis	Trichomonas vaginalis and birth outcome	Trichomonas vaginalis and birth outcome				
5	5	Trichomonas vaginalis and pregnancy outcome				
		Trichomonas vaginalis and stillbirth				
		Trichomonas vaginalis and IUGR				
		Trichomonas vaginalis and Preterm delivery				
		Trichomonas vaginalis and low birth weight				
		Trichomonas vaginalis and stillbirth				
		Trichomonas vaginalis and PROM				
		Trichomoniasis and birth outcome				
		Trichomoniasis and pregnancy outcome				
		Trichomoniasis and stillbirth				
		Trichomoniasis and IUGR				
		Trichomoniasis and Preterm delivery				
		Trichomoniasis and low birth weight				
		Trichomoniasis and PROM				
Bacterial vaginosis	Bacterial vaginosis and birth outcome	Bacterial vaginosis and birth outcome				
		Bacterial vaginosis and pregnancy outcome				
		Bacterial vaginosis and stillbirth				
		Bacterial vaginosis and IUGR				
		Bacterial vaginosis and Preterm delivery				
		Bacterial vaginosis and low birth weight				
		Bacterial vaginosis and PROM				
Malaria	Malaria and " birth outcome"	Malaria and birth outcome				
		Malaria and pregnancy outcome				
		Malaria and stillbirth				
		Malaria and IUGR				
		Malaria and Preterm delivery				
		Malaria and low birth weight				

Organism/infection	Medical subject heading terms	Free text terms
Treponema	Treponema pallidum OR syphilis AND "pregnant women"	Treponema pallidum and pregnant and Africa
		Treponema pallidum and pregnancy and Africa
paillaum	AND AIrica	Syphilis and pregnant and Africa
		Syphilis and pregnancy and Africa
		Screen and syphilis and pregnant and Africa
		Screen and syphilis and pregnancy and Africa
		Test and syphilis and pregnant and Africa
		Test and syphilis and pregnancy and Africa
Neisseria	Neisseria gonorrhoeae OR Gonorrhoea OR Gonorrhoea AND	Neisseria gonorrhoea and pregnant and Africa
acharrhaad	Drognant woman AND Africa	Neisseria gonorrhoea and pregnancy and Africa
gonornoed	Pregnant women AND Anica	Gonorrhoea and pregnant and Africa
		Gonorrhea and pregnancy and Africa
		Screen and gonorrhoea and pregnant and Africa
		Screen and gonorrhea and pregnant and Africa
		Test and gonorrhoea and pregnant and Africa
		Test and gonorrhea and pregnant and Africa
Chlamvdia	Chlamydia trachomatis and Pregnant women AND Africa	Chlamydia trachomatis and pregnant and Africa
the share still		Chlamydia trachomatis and pregnancy and Africa
tracnomatis		Screen and Chlamydia and pregnant and Africa
		Screen and Chlamydia and pregnancy and Africa
		Test and Chlamydia and pregnant and Africa
		Test and Chlamydia and pregnancy and Africa
Trichomonas	Trichomonas vaginalis and Pregnant women AND Africa	Trichomonas vaginalis and pregnant and Africa
		Trichomonas vaginalis and pregnancy and Africa
vaginalis		Trichomoniasis and pregnant and Africa
		Trichomoniasis and pregnancy and Africa
		Screen and Trichomoniasis and pregnant and Africa
		Screen and Trichomoniasis and pregnancy and Africa
		Test and Trichomoniasis and pregnant and Africa
		Test and Trichomoniasis and pregnancy and Africa
Bacterial vaginosis	Bacterial vaginosis and Pregnant women AND Africa	Bacterial vaginosis and pregnant and Africa
		Bacterial vaginosis and pregnancy and Africa
		Screen and Bacterial vaginosis and pregnant and Africa
		Screen and Bacterial vaginosis and pregnancy and Africa
		Test and Bacterial vaginosis and pregnant and Africa
		Test and Bacterial vaginosis and pregnancy and Africa
Malaria	Malaria and Pregnant women AND Africa	Malaria and pregnant and Africa
in a la l		Malaria and pregnancy and Africa
1		Screen and malaria and pregnant and Africa
		Screen and malaria and pregnancy and Africa
1		Test and malaria and pregnant and Africa
1		Test and malaria and pregnancy and Africa

Appendix 1.1b: Search terms for studies reporting point prevalence for malaria and STIs/RTIs

Site	Reference	No.	No. tested	Prevalence	95% CI	Diagnostic method
Year study conducted		women				
		positive				
Chikwawa, Malawi (1993-94)	Verhoeff, 1998 [293]	109	575	18.9	15.7-22.1	Giemsa stain microscopy
Chikwawa, Malawi (1993-95)	Verhoeff, 1999 [294]	743	3913	19.0	17.8-20.2	Giemsa stain microscopy
Blantyre, Malawi (1997-98)	Rogerson, 2000 [141]	2034	4764	42.7	41.3-44.1	PCR
Blantyre, Malawi (1997-98)	Rogerson, 2000 [88]	85	339	25.1	20.5-29.7	Giemsa stain microscopy
Kigoma, Tanzania (NR)	Mnyika, 2000[295]	66	705	9.4	7.3-11.6	Giemsa stain microscopy
Manhica, Mozambique (1998)	Saute 2002 [296]	158	686	23.0	19.8-26.2	Giemsa stain microscopy
Blantyre, Malawi (2000)	Mankhambo, 2002[297]	70	135	51.9 ^{¥¥}	43.5-60.3	PCR
Sofala, Mozambique (2003-04)	Montoya, 2006 [298]	78	587	13.3	10.6-16.1	Giemsa stain microscopy
Morogoro, Tanzania (2001-02)	Wort,2006 [142]	493	1684	29.3	27.1-31.5	Giemsa stain microscopy
Multicenter, Mozambique (2003-04)	Brentlinger,2007 [299]	255	684	37.3	33.7-40.9	Giemsa stain microscopy
Multicentre, Mozambique (2004)	Lujan,2008 [300]	52	1117	4.7 [¥]	3.5-5.9	Giemsa stain microscopy
Southern Malawi (2004-05)	Van Den Broek, 2009[301]	571	2297	24.9	23.1-26.7	Giemsa stain microscopy
Mongochi, Malawi (2003-04)	Rantala, 2010 [95]	51	475	10.7	7.9-13.5	PCR
Blantyre, Malawi (1997-2001)	Feng, 2010 [1]*	798	7671	10.4	9.7-11.1	Giemsa stain microscopy
300 DHS geographic clusters, DRC	Taylor, 2011 [144]*	193	520	37.2	33.3-41.7	PCR
Sambizanga and Ingombota, Angola	valente, 2011[89]*	49	567	8.6	6.3-10.9	PCR
(2006-2007)						
Manhica, Mozambique (2003-04)	Serra-Casas, 2011 [143]	120	399	30.0	25.5-34.5	PCR
Manhica, Mozambique (2003-2005)	Mayor, 2012 [302]*	26	272	9.5	6.0-13.0	Giemsa stain microscopy
Luanda, Angola (2008)	Campos,2012 [145] *	74	679	10.9	8.6-13.2	Giemsa stain microscopy
Nchelenge, Zambia (2013)	Siame, 2015[115]*	83	375	22	17.8-26.2	PCR
Mansa , Zambia (2009-2010)	Mace, 2012 [90]*	21	419	5.0	2.9-7.1	Giemsa stain microscopy
Machinga, Malawi (2010-2011)	Gutman, 2015 [261]*	760	2,566	29.6	27.8-31.2	RDT
PCR: polymerase chain reaction		•		1		,
*Studies that did not contribute to the no	oled prevalence estimates in th	no mota-analy	sis hy Chico et al	1 2012 [6]		

Appendix 1.2a: Peripheral malaria point prevalence estimates among pregnant women in Zambia and surrounding countries

*Studies that did not contribute to the pooled prevalence estimates in the meta-analysis by Chico *et al.* 2012 [6]

[¥]Lowest prevalence estimate

Rapid diagnostic test

Site	Reference	No. women	No.	Prevalence	95% CI	Diagnostic method			
Year study conducted		positive	tested						
North Kivu, DRC (NR)	Meuris, 1993[303]	242	461	52.5	47.9-57.1	Placental histology			
Blantyre, Malawi (1997-99)	Rogerson, 2000[88]	77	232	33.2	27.1-39.3	Placental impression and blood smear with microscopy			
Blantyre, Malawi (NR)	Rogerson, 2003 [104]	124	464	26.7	22.6-30.7	Placental histology, impressions, smear with microscopy			
Kinshasa, DRC (2004)	Lukuka, 2006 [304]	41	196	21.0	15.3-26.7	Placental impression and blood smear with microscopy			
Chikwakwa, Malawi (2004-05)	Senga, 2007[305]	124	636	19.5	16.4-22.6	Placental histology and blood smear with microscopy			
Muheza, Tanzania (2002-05)	Mwangoka, 2008 [306]	100	1,022	9.8	8.0-11.6	Placental blood microscopy			
Muheza, Tanzania (2002-05)	Kabyemela, 2008 [146]	55	445	12.4	9.3-15.5	Placental impression and blood smear with microscopy			
Blantyre, Malawi (1997-2001)	Feng, 2010 [1]	1058	7671	13.8	13.0-14.4	Placental blood microscopy			
Luanda,Angola	Valente, 2011 [89]*	99	567	17.5	14.4-20.6	PCR			
Manhica, Mozambique (2003-05)	Serra-Casas, 2011 [143]	110	342	32.1	27.2-37.1	PCR			
Manhica, Mozambique (2003-05)	Mayor, 2012 [302]*	156	272	57.4 ^{¥¥}	51.5-63.3	Placental histology			
Mansa, Zambia (2009-2010)	Mace, 2012 [90] *	161	435	37.2	32.7-41.7	Placental histology			
Rufiji, Tanzania (2012)	Mosha, 2014 [147]*	29	175	16.6	11.1-22.1	PCR			
Moshi, Tanzania (2012)	Mosha, 2014 [147]	4	175	2.3 [¥]	0.1-4.5	PCR			
PCR: polymerase chain reaction									
*Studies that did not contribute to th	*Studies that did not contribute to the pooled prevalence estimates in the meta-analysis by Chico <i>et al.</i> 2012 [6]								
^{¥¥} Uighost provolongo ostimoto	- F F		,	[-]					

Appendix 1.2b: Placental malaria point prevalence estimates among pregnant women in Zambia and surrounding countries

** Highest prevalence estimate

[¥]Lowest prevalence estimate

Site	Reference	No.	No.	Prevalence	95% CI	diagnostic method
Year study conducted		women	tested			
		positive				
Vilanculos, Mozambique (1991-92)	Vuylsteke, 1993[307]	31	201	14.6 ^{¥¥}	9.7-19.5	RPR and TPHA
Umzigwana, Zimbabwe (1991)	Rutgers, 1993[308]	198	1433	13.8	12.0-15.6	RPR only
Zambiazai, Mozambique (1992-93)	Cossa, 1994[131]	211	1728	12.2	10.7-13.7	RPR and MHA-TP
Mwanza, Tanzania(1992-93)	Mayaud, 1995 [126]	97	964	10.1	8.2-12.0	RPR and TPHA
Dar es Salaam, Tanzania (1993)	Mwakagile, 1996 [129]	31	777	4.0	2.6-5.4	RPR and TPHA
Rwandan Camp, Tanzania (1994)	Mayaud, 1997[309]	2	100	2.0	0.7-4.7	RPR and TPHA
Mwanza, Tanzania (NR)	Mayaud, 1997 [310]	174	2380	7.3	6.2-8.4	RPR and TPHA
Mwanza, Tanzania (1994)	Mayaud, 1998[311]	45	628	7.2	5.2-9.2	RPR and TPHA
Blantyre, Malawi (1995)	Taha, 1998 [312]	98	808	12.1	9.8-14.7	RPR and TPHA
Blantyre, Malawi (1996)	Taha, 1999 [313]	1005	9309	10.8	10.2-11.4	RPR and TPHA
Murewa, Zimbabwe	Kambarami, 1998 [314]	36	366	9.8	6.8-12.9	RPR and TPHA
Murewa, Zimbabwe	Kambarami, 1998 [314]	28	308	9.2	6.0-12.4	VDRL and TPHA
Mwanza,Tanzania (2002)	Watson-Jones,2002 [10]	106	1809	5.9	4.8-7.0	RPR and TPHA
Gutu, Zimbabwe (NR)	Majoko, 2003 [315]	9	85	10.6	4.1-17.1	RPR only
Harare, Zimbabwe (NR)	Pham, 2005[132]	74	2969	2.5	1.9-3.1	RPR and TPHA
Sofala, Mozambique (2003-04)	Montoyo, 2006[298]	33	381	8.6	5.8-11.4	RPR and TPHA
Gaborone, Botswana (2000-01)	Romeron, 2006 [316]	35	703	5.0	3.4-6.6	RPR and TPHA
Multicentre, Tanzania (2003-04)	Swai, 2006 [136]	92	1,265	7.3	5.9-8.7	RPR only
ZDHS sites, Zambia	CSO, 2007 [317]	5	128	4.1	2.9-5.3	RPR and TPHA
Blantyre, Malawi (2000-04)	Kwiek, 2008 [137]	199	3824	5.2	4.5-5.9	RPR and TPHA
Kinshansa, DRC (2004)	Kanoshita-Moleka, 2008[318]	0	529	0.0 [¥]	0-0	RPR and TPHA
Multicentre, Mozambique (2004)	Lujan, 2008 [300]	52	1117	4.7	3.5-5.9	RPR and TPHA
Rural Tanzania (2003-04)	Yahya-Malima 2008 [133]	21	1296	1.6	0.9-2.3	RPR and TPHA
Manhica, Mozambique	Menendez 2008 [319]	122	1030	11.8	9.8-13.8	RPR only
Moshi, Tanzania (2002-04)	Msuya, 2009[221]*	24	2654	0.9	0.5-1.3	RPR and Determine Syphilis TP
Southern Malawi (2004-05)	Van Den Broek, 2009 [301]	163	2297	7.1	6.1-8.2	VDRL only
Manhica, Mozambique (2000)	Menendez, 2010 [139]	31	258	12.0	8.0-16.0	RPR and enzyme immunoassay
Harare, Zimbabwe (2002-03)	Kurewa, 2010 [128]	8	678	1.2	0.4-2.0	RPR and TPHA
Harare, Zimbabwe (2002-04)	Mapingure, 2010 [320]	8	662	1.2	0.4-2.0	RPR and Determine Syphilis TP
Sentinel sites, Congo DR (2011)	Taylor, 2014 [321]*	742	17,669	4.2	3.9-4.5	RPR and Determine Syphilis TP
Multiple, Tanzania (2008-2010)	Chiduo, 2012 [322]*	284	25,802	1.1	1.0-1.2	RPR and TPHA
Mwanza, Tanzania (2012)	Hokororo, 2015 [130]*	21	403	5.2	3.0-7.4	RPR and TPPA
Multicentre, Tanzania (2011)	Manyahi, 2015 [138]*	973	38,920	2.5	2.3-2.7	RPR only
Kalomo, Zambia (2012)	Bonawirz, 2015 [323]*	50	1868	2.7	2.3-3.4	Rapid syphilis test

Appendix 1.2c: Syphilis point prevalence estimates among pregnant women in Zambia and surrounding countries

Appendix 1.2c: Syphilis point prevalence estimate among pregnant women in Zambia and surrounding countries continued

RPR: Rapid plasma reagin

TPHA: *Treponema pallidum* haemagglutination assay

TPPA: Treponema pallidum particle agglutination

VDRL: Venereal disease research laboratory

FTA-Abs-Fluorescent treponemal antibody absorption

MHA-TP: Micro heamagglutination assay-*Treponema pallidum*

CSO-Central statistical Office.

*Studies that did not contribute to the pooled prevalence estimates in the meta-analysis by Chico et al. 2012 [6]

^{¥¥} Highest prevalence estimate

[¥]Lowest prevalence estimate

Site	Reference	No.	No.	Prevalence	95% CI	Diagnostic method
Year study conducted		women	tested			
		positive				
Vilancolus, Mozambique (1991-92)	Vuylsteke, 1993 [307]	11	141	7.8	3.4-12.2	Culture
Kinshasa, DRC (1990)	Vuylsteke, 1993 [324]	75	1,160	6.5	5.1-7.9	EIA
Mwanza,Tanzania (1992)	Mayaud, 1995 [126]	64	964	6.6	5.0-8.2	EIA
Mwanza, Tanzania (1994)	Mayaud, 1998 [311]	39	660	5.9	4.1-7.7	EIA
Harare, Zimbabwe	Latif, 1999 [125]*	88	1656	5.3 [¥]	4.2-6.4	EIA
Gaborone, Botswana (2001-01)	Romoren, 2004 [325]	42	557	7.5	5.1-9.5	LCR cervical swab
Multicentre, Mozambique (2004)	Lujan, 2008 [300]	51	835	6.1	4.8-7.7	LCR cervical swab
Kinshasa, DRC (2004)	Kanoshita-Moleka, 2008 [318]	9	521	1.7	0.6-2.8	PCR cervical swab
Moshi, Tanzania (2002-04)	Msuya, 2009 [221]*	183	1048	17.5 ^{¥¥}	15.2-19.8	ELISA
Manhica, Mozambique (2000)	Menendez, 2010 [139]	15	151	9.9	5.1-14.7	DNA ID-assay cervical swabs
Mwanza, Tanzania (2012)	Hokororo, 2015[130]*	48	403	11.4	8.2-14.4	PCR
PCR: polymerase chain reaction	-				•	
*Studies that did not contribute to the	e pooled prevalence estimates in the	meta-analysis	by Chico <i>et al</i>	. 2012 [6]		
^{¥¥} Highest prevalence estimate						
[*] Lowest prevalence estimate						

Appendix 1.2d: Chlamydia trachomatis point prevalence estimates among pregnant women in Zambia and surrounding countries

Site	Reference	No. women	No.	Prevalence	95% CI	Diagnostic method
Year study conducted		positive	tested			
Kinshasa, DRC (1990)	Vuylsteke,1993 [324]	19	1,160	1.6	0.9-2.3	Culture
Vilanculos, Mozambique (1991-92)	Vuylsteke,1993[307]	14	201	7.0	3.5-10.5	Culture
Mwanza, Tanzania (1992-93)	Mayaud,1995 [126]	20	964	2.1	1.2-3.0	Culture
Dar es Salaam, Tanzania (1993)	Mwakagile 1996 [129]	28	777	3.6	1.9-4.3	Culture
Mwanza, Tanzania (1994)	Mayaud, 1998 [311]	15	660	2.3	1.2-3.4	Culture
Blantyre, Malawi (1993)	Taha, 1998 [312]	54	2161	2.5	1.8-3.2	Culture
Blantyre, Malawi (1995)	Taha, 1998[312]	20	808	2.5	1.4-3.6	Culture
Harare, Zimbabwe	Latif 1999 [125]*	65	1656	3.9	3.0-4.8	Culture
Blantyre, Malawi (1995)	Taha, 1999 [313]	373	9104	4.1	3.2-5.0	LCR cervical swab
Gaborone, Botswana (2000-01)	Romeron, 2007 [127]	21	703	3.0	1.7-4.3	Culture
Multicentre, Mozambique (2004)	Lujan, 2008 [300]	21	835	2.5	1.4-3.6	PCR urine
Moshi, Tanzania (2002-04)	Msuya, 2009 [221]*	13	2555	0.5 [¥]	0.2-0.8	Culture and gram stain
Manhica, Mozambique (2000)	Menendez, 2010 [139]	27	145	18.6 ^{¥¥}	12.2-24.8	Culture and gram stain
Mwanza, Tanzania (2012)	Hokororo, 2015 [130]*	27	403	6.7	4.3-9.1	PCR
PCR: polymerase chain reaction	-					
*Studies that did not contribute to the	e pooled prevalence estimates i	in the meta-analysis b	by Chico <i>et al</i> .	2012 [6]		
^{¥¥} Highest prevalence estimate	-					

Appendix 1.2e: Neisseria gonorrhoeae point prevalence estimates among pregnant women in Zambia and surrounding countries

^{*}Lowest prevalence estimate

Site	Reference	No. women	No. tested	Prevalence	95% CI	Diagnostic method
Year study conducted		positive				
Kinshasa, DRC (1990)	Vuylsteke,1993 [324]	213	1,160	18.4	16.2-20.6	Wet-mount microscopy
Vilancolus, Mozambique (1991-92)	Vuylsteke 1993[307]	46	201	22.9	17.1-28.7	Wet-mount microscopy
Dar es Salaam, Tanzania (1993)	Mwakangile 1996[129]	176	777	22.7	19.8-25.7	Wet-mount microscopy
Rwadan camp, Tanzania (1994)	Mayaud, 1997[309]	31	100	31.0	21.9-40.1	Wet-mount microscopy
Mwanza, Tanzania (1994)	Mayaud, 1998 [311]	108	660	16.4	13.6-19.2	Wet-mount microscopy
Blantyre, Malawi (1993)	Taha, 1998 [312]	19	808	2.4 [¥]	1.3-3.5	Wet-mount microscopy
Harare, Zimbabwe	Latif, 1999 [125]	22	1656	13.0	11.4-14.6	Wet-mount microscopy
Blantyre, Malawi (1994)	Taha, 1999 [313]	215	9137	31.1	30.1-32.1	Wet-mount microscopy
Gaborone, Botswana (2000-01)	Romoren, 2007[326]	132	703	18.8	15.9-21.7	Wet-mount microscopy
Moshi, Tanzania (2002-2004)	Msuya, 2009 [221]*	128	2555	5.0	4.2-5.8	Wet-mount microscopy
Ndola, Zambia (NR)	Crucitti, 2010 [327]*	99	307	32.2 ^{¥¥}	27.0-37.4	PCR
Manhica, Mozambique (2000)	Menendez 2010 [139]	78	254	30.7	25.0-36.4	Wet-mount microscopy
Moshi, Tanzania (2002-04)	Mapingure, 2010 [320]	128	2555	5.0	4.2-5.9	Wet-mount microscopy
Harare, Zimbabwe (2002-04)	Mapingure, 2010 [320]	80	680	11.8	9.4-14.2	Wet-mount microscopy
Harare, Zimbabwe (2000)	Kurewa, 2010 [128]	80	678	11.8	9.4-14.3	Wet-mount microscopy
Mwanza, Tanzania (2012)	Hokororo, 2015 [130]*	54	403	13.4	10.1-16.7	Wet-mount microscopy
PCR: polymerase chain reaction						
*Studies that did not contribute to the p	ooled prevalence estimates in	the meta-analysis by	y Chico <i>et al</i> . 201	2 [6]		
^{¥¥} Highest prevalence estimate						

Appendix 1.2f: Trichomonas vaginalis point prevalence estimates among pregnant women in Zambia and surrounding countries

[¥]Lowest prevalence estimate

			<u></u>		r	
Site	Reference	No.	No.	Prevalence	95% CI	Diagnostic method
Year study conducted		women	tested			
		positive				
Rwadan camp, Tanzania (1994)	Mayaud, 1997 [309]	16	100	16.0	8.8-23.2	Clue cell count
Mwanza, Tanzania (1994)	Mayaud, 1998 [311]	159	660	24.1	20.8-27.4	Clue cell count
Harare, Zimbabwe	Latif, 1999 [125]*	71	1656	4.3 [¥]	3.3-5.3	Amsel
Blantyre, Malawi (1993)	Taha, 1999 [313]	640	2077	30.8	28.8-32.8	Amsel
Multicentre, Zimbabwe (1999-2004)	Toloso, 2006 [328]	51	209	24.4	18.6-30.2	Nugent
Gaborone, Botswana (2000-01)	Romoren, 2007 [326]	264	703	37.6 ^{¥¥}	34.0-41.2	Nugent
Tanzania (1999-2004)	Msuya, 2009 [221]*	534	2555	20.9	19.3-22.5	Amsel
Harare, Zimbabwe (2002-04)	Mapingure, 2010 [320]	195	598	32.6	28.8-36.4	Amsel
Moshi, Tanzania (2002-04)	Mapingure, 2010 [320]	534	2555	20.9	19.3-22.5	Amsel
Harare, Zimbabwe (2002-03)	Kurewa,2010 [128]	221	678	32.6	29.1-36.1	Amsel
Mwanza, Tanzania (2011)	Shayo, 2012[329]	81	284	28.5	23.2-33.6	Nugent
Mwanza, Tanzania (2012)	Hokororo, 2015 [130]*	102	403	25.3	21.1-29.5	Nugent
*Studies that did not contribute to the p	ooled prevalence estimates in	the meta-ar	alysis by Chic	o <i>et al</i> . 2012 [6]		
^{¥¥} Highest prevalence estimate						
[*] Lowest prevalence estimate						

Appendix 1.2g: Bacterial vaginosis point prevalence estimates among pregnant women in Zambia and surrounding countries

Infection	Diagnostic method	Sensitivity	Specificity
Syphilis	Treponema pallidum haemagglutination assay	_~ 100	_~ 100 [330]
	Rapid plasma regain (RPR) only	86-100	93-98 [116]
	Venereal disease research laboratory (VDRL)	78-100	98 [116]
	RPR or VDRL	_~ 100	_~ 100 [6]
	RPR and TPHA	_~ 100	_~ 100 [6]
	RPR and Determine Syphilis TP	_~ 100	_~ 100 [6]
	RPR and Microhemagglutination assay-Treponema pallidum (MHA-TP)	_~ 100	_~ 100 [6]
	RPR and Immunoglobulin G (ELISA test)	_~ 100	_~ 100 [6]
Neisseria gonorrhoeae	Gram stain	50-70	95-100 [331]
	Polymerase chain reaction (PCR) with cervical swab	36-75	98-100 [332]
	Culture	_~ 80	100 [117]
	Culture and Gram stain	_~ 80	100 [6]
	Ligase chain reaction (LCR) with cervical swab	95-100	98-100 [117]
	PCR with cervical swab	89-97	94-100 [117]
Chlamydia trachomatis	Culture cervical swab	74-90	98-99 [117]
	Enzyme immunoassay (EIA)	71-87	97-99 [117]
	Culture cervical swab or EIA (score based on the less precise method, i.e mean of EIA)	71-87	97-99 [6]
	PCR urine	78-89	99-100 [332]
	LCR cervical culture	90-97	99-100 [117]
	PCR cervical swab	99	99-100 [117]
	Deoxyribonucleic acid identification assay-cervical swab	_~ 100	_~ 100 [6]
Trichomonas vaginalis	Wet mount microscopy	38-82	100 [117]
	In-pouch culture	96	100 [333]
	PCR	97	98 [334]
Bacterial vaginosis	Clue cells ≥ 20% of epithelium	40	97 [121]
	Amsel criteria: 3 or 4 of 4	51	98 [121]
	Nugent score ≥ 7	86-89	94-96 [122-124]
Peripheral parasitaemia	Giemsa stain microscopy	50 -90	_~ 95 [91-93]
	PCR	91	91[95]
	Antigen assay	Reported as equal to	microscopy [6]
	Rapid diagnostic test	89	76
Placental parasitaemia	Placental blood microscopy	63	98-99 [104]
	PCR	93-99	73-76 [335]
	Placental histology	91	98-99 [104]
	Placental blood microscopy and tissue histology (score based on more precise method, i.e histology	91	_~ 100 [6]
	Placental histology, impression, smears with microscopy (score based on most precise method i.e. histology	91	_~ 100 [6]
	Placental impression and blood smear with microscopy (score based on more precise method, i.e. impression	91	_~ 100 [105]

Appendix 1.3: Diagnostic Methods and published Sensitivity and Specificity Measures

Infection	Site, country (year)	First Author, Year of publication	Risk factors for infection
Treponema pallidum	Vilanculos, Mozambique (1991-92)	Vuylsteke, 1993[307]	NR
	Umzigwana, Zimbabwe (1991)	Rutgers, 1993[308]	NR
	Zambiazai, Mozambique (1992-93)	Cossa, 1994[131]	The following factors were associated with infection; married status (aOR 2.5, 95% CI, 1.0-6.1), history of past STD (aOR5.6; 95% CI, 3.3-9.4), genital ulcers (aOR 3.7; 95% CI, 1.4-9.8), vaginal discharge (OR 1.5; 95% CI, 1.0- 2.1), genital warts (aOR 3.0; 95% CI, 1.1- 8.7), HIV infection (aOR 2.2; 95% CI, 1.0-5.0), and being from Nicoadala district (aOR 6.8; 95%CI, 1.7-58.1).
	Mwanza, Tanzania(1992-93)	Mayaud, 1995 [126]	NR
	Dar es Salaam, Tanzania (1993)	Mwakagile, 1996 [129]	Higher prevalence of syphilis, trichomoniasis and gonorrhoea in teenagers than older women (45.3% verses 29.2, <i>P</i> < 0.001)
	Rwandan Camp, Tanzania (1994)	Mayaud, 1997[309]	NR
	Mwanza, Tanzania (NR)	Mayaud, 1997 [310]	NR
	Mwanza, Tanzania (1994)	Mayaud, 1998[311]	NR
	Blantyre, Malawi (1995)	Taha, 1998 [312]	NR
	Blantyre, Malawi (1996)	Taha, 1999 [313]	NR
	Murewa, Zimbabwe	Kambarami, 1998 [314]	RPR positivity was not significantly associated with the women's age or parity.
	Mwanza,Tanzania (2002)	Watson-Jones,2002 [10]	NR
	Gutu, Zimbabwe (NR)	Majoko, 2003 [315]	NR
	Harare, Zimbabwe (NR)	Pham, 2005[132]	Increases in parity and gravidity were significantly associated with increased risk of syphilis infection. Prior stillbirths were associated with an increased risk of syphilis infection (odds ratio [aOR], 3.4; 95% CI, 1.61 to 7.37; $P = 0.001$).
	Sofala, Mozambique (2003-04)	Montoyo, 2006[298]	NR
	Multicentre, Tanzania (2003-04)	Swai, 2006	Women living in rural areas had higher prevalence than those in urban areas ($P < 0.0001$).Marital status did not appear to influence the prevalence of syphilis. There was a higher likelihood of having syphilis for women aged > 34 years compared to those less than 34 years. Women with no education were more likely to be infected with syphilis than were women with some education ($P < 0.0001$)

Infection	Site, country (year)	First Author, Year of	Risk factors for infection
Trananama nalllidum	ZDUS sites Zambia		ND
rreponema palmaam	ZDHS sites, Zambia	CSU, 2007 [317]	
	Noshi, Talizalila (2002-04)	Wisk 2007[550]	NR Sumbilie corproportivity was associated with anomia (200) 1.4.05% (1
	Biantyre, Malawi (2000-04)	KWIEK, 2008 [137]	Syphilis seroreactivity was associated with anemia (aOR: 1.4, 95% Cl 1.0, 1.9), previous diagnosis of STI (aOR: 5.3, 95%Cl: 3.5, 8.0), rural residence, multigravidity. Self-reported history of miscarriage or stillbirth (aOR: 2.3, 95%Cl: 1.6, 3.5) Syphilis seroreactivity was inversely associated with maternal education ($\chi^2_{(trend)}$ =17, <i>P</i> <0.0001). Syphilis was not associated with maternal employment, cost of housing material, being a Yao, or marital status.
	Kinshansa, DRC (2004)	Kanoshita-Moleka, 2008[318]	NR
	Multicentre, Mozambique (2004)	Lujan, 2008 [300]	No factors were independently associated with syphilis infection
	Rural Tanzania (2003-04)	Yahya-Malima 2008 [133]	Women with self-reported previous spontaneous abortion were more likely to have a positive syphilis serology (aOR 4.3, 95% CI: 1.52–12.02), infection with syphilis not associated with self-reported stillbirth (OR 2.4, 95% CI: 0.64–8.95).Odds were also higher for women who had more than one lifetime sexual partner, OR 5.4 (95% CI: 1.88–15.76).
	Manhica, Mozambigue	Menendez 2008 [319]	NR
	Moshi, Tanzania (2002-04)	Msuya, 2009[221]*	NR
	Southern Malawi (2004-05)	Van Den Broek, 2009[301]	NR
	Manhica, Mozambique (2000)	Menendez, 2010 [139]**	Being divorced/widowed (aOR 3.6; 95% Cl, 0.6-21.9) was associated with syphilis infection
	Harare, Zimbabwe (2002-03)	Kurewa, 2010 [128]	Risk factors for a positive serologic STI* were increasing age above 30 years (aOR 2.61; 95% Cl, 1.49-4.59), polygamy (aOR 2.16; 1.06-4.39), and multigravid (aOR 3.89: 1.27-11.98), number of lifetime sexual partners (aOR 2.8; 1.2-6.2) and partner taking alcohol (aOR 1.16; 1.01-1.33).

Infection	Site, country (year)	First Author, Year of publication	Risk factors for infection
Treponema pallidum	Harare, Zimbabwe (2002-04)	Mapingure, 2010 [320]	NR
	Sentinel sites, Congo DR (2011)	Taylor, 2014 [321]	Syphilis seropositivity was significantly higher among women attending rural clinics (5.0%) as compared to urban clinics (3.0%) and those tested in antenatal care clinics in the provinces of Equateur (7.6%) and Orientale (7.7%) as compared to other provinces $P < 0.001$).
	Mwanza, Tanzania (2012)	Hokororo, 2015 [130]	The presence of any STI^{α} was associated with being in a long-term (as opposed to short-term) relationship (aOR 2.6; 95% Cl 1.4 - 4.9, increasing age difference between the girl and her partner (aOR 1.1; 95% Cl 1.0 - 1.1 per year, and history of prior pregnancy (aOR 1.6; 95% Cl 1.0 - 2.6).
	Multicentre, Tanzania (2011)	Manyahi, 2015 [138]	The risk for syphilis infection was significantly higher among women attending semi-urban and rural clinics than those attending urban clinics and those having 3-4 (aOR 1.50; 95% CI, 1.17-1.92) and 5 (aOR 1.70, 95% CI, 1.27-2.27) previous pregnancies. Marital status and level of education were not statistically significant
Neisseria gonorrhoea	Kinshasa, DRC (1990)	Vuylsteke,1993 [324]	NR
	Mwanza,Tanzania (1992)	Mayaud, 1995 [126]	Chlamydia/gonorrhoea Age < 25 years (aOR=2.2, <i>P</i> < 0.003; Unmarried (aOR= 3.2, <i>P</i> < 0.003; in polygamous marriage (aOR = 2.3, <i>P</i> < 0.003; > one sexual partner over the previous year (aOR = 1.7, <i>P</i> < 0.06; previously delivered more than 5 years (OR=3.2, <i>P</i> < 0.01
	Dar es Salaam, Tanzania (1993)	Mwakagile 1996 [129]	As stated for this paper under syphilis (infections combined)
	Blantyre, Malawi (1995)	Taha, 1999 [313]	NR
	Harare, Zimbabwe	Latif 1999 [125]	Women with vaginal discharge: being separated from the partner for a month or more ($P = 0.002$), having had sex with a new partner in the previous three months ($P = 0.002$), current use of condoms (P = 0.011 presence of a purulent vaginal discharge ($P = 0.004$). increasing educational level was inversely associated with cervical infection ($P = 0.007$).

Infection	Site, country (year)	First Author, Year of publication	Risk factors for infection
Neisseria	Gaborone, Botswana (2000-01)	Romeron, 2007 [127]	Compared to Women aged 30 and above, women aged 20-29 (aOR
gonorrhoeae			4.0; 1.60-10.10) and below 20 (aOR 10.5; 3.59-30.72) had a higher
			risk of infection with Chlamydia and/or gonorrhoea.
	Multicentre, Mozambique (2004)	Lujan, 2008 [300]	No factors were independently associated with infection
			gonorrhoea
	Moshi, Tanzania (2002-04)	Msuya, 2009 [221]*	NR
	Manhica, Mozambique (2000)	Menendez, 2010 [139]	Women with more than one lifetime sexual partner were at a
			higher risk of being seropositive for gonorrhoea than women with
			one lifetime sexual partner (aOR 2.8; 1.2-6.3)
	Mwanza, Tanzania (2012)	Hokororo, 2015 [130]	As stated for this paper under syphilis (infections combined)
Chlamydia trachomatis	Vilancolus, Mozambique (1991-92)	Vuylsteke, 1993 [307]	NR
	Kinshasa, DRC (1990)	Vuylsteke, 1993 [324]	NR
	Mwanza,Tanzania (1992)	Mayaud, 1995 [126]	As stated for this paper under gonorrhoea (the two infections
			were combined)
	Harare, Zimbabwe	Latif, 1999 [125]*	As stated for this paper under gonorrhoea (the two infections
			were combined)
	Gaborone, Botswana (2001-01)	Romoren, 2004 [325]	NR
	Multicentre, Mozambique (2004)	Lujan, 2008[300]	No factors were independently associated with infection
			Chlamydia
	Kinshasa, DRC (2004)	Kanoshita-Moleka, 2008 [318]	NR
	Moshi, Tanzania (2002-04)	Msuya, 2009 [221]*	NR
	Manhica, Mozambique (2000)	Menendez, 2010 [139]	Infection with chlamydia was not significantly associated with any
			social or behavioral risk factor
	Mwanza, Tanzania (2012)	Hokororo, 2015 [130]	As stated for this paper under syphilis (infections were combined)
Trichomonas vaginalis	Kinshasa, DRC (1990)	Vuylsteke,1993[324]	NR
	Vilancolus, Mozambique (1991-92)	Vuylsteke 1993 [307]	NR
	Dar es Salaam, Tanzania (1993)	Mwakangile 1996 [129]	Prevalence was lower in pregnant women in monogamous
			marriages
	Rwadan camp, Tanzania (1994)	Mayaud, 1997[309]	NR
	Mwanza, Tanzania (1994)	Mayaud, 1998 [311]	NR

Appendix 1.4: Risk factors for malaria and STIS/RTIS continu	ıed
--	-----

Infection	Site, country (year)	First Author, Year of publication	Risk factors for infection
Trichomonas vaginalis	Blantyre, Malawi (1993)	Taha, 1999 [313]	NR
	Gaborone, Botswana (2000-01)	Romoren, 2007[326]	NR
	Ndola, Zambia (NR)	Crucitti, 2010 [327]*	NR
	Manhica, Mozambique (2000)	Menendez 2010 [139]	Women with anogenital warts were at increased risk of having trichomoniasis (aOR 7.1, 95% CI, 1.4-36.5)
	Multicentre	Mapingure, 2010 [320]	NR
	Harare, Zimbabwe (2000)	Kurewa, 2010 [128]	For vaginal infections** it was age at sexual debut; sexual debut before 20 years of age compared to whose sexual debut was after 20 th birthday (aOR 1.60; 95% CI, 1.06-2.42) Being in a polygamous relationship was significantly associated with having <i>Trichomonas vaginalis</i> ; (aOR 2.24, 95% CI, 1.09-4.56).
	Mwanza, Tanzania (2012)	Hokororo, 2015 [130]	As stated for this paper under syphilis (infections were combined)
Bacterial vaginosis	Rwadan camp, Tanzania (1994)	Mayaud, 1997 [309]	NR
	Mwanza, Tanzania (1994)	Mayaud, 1998 [311]	NR
	Harare, Zimbabwe	Latif, 1999 [125]	NR
	Multicentre, Zimbabwe (1999-2004)	Toloso, 2006 [328]	NR
	Harare, Zimbabwe	Taha, 1999 [313]	NR
	Gaborone, Botswana (2000-01)	Romoren, 2007 [326]	NR
	Tanzania (1999-2004)	Msuya, 2009 [221]*	NR
	Moshi, Tanzania (2002-04) and Harare, Zimbabwe (2002-04)	Mapingure, 2010 [320]	NR
	Harare, Zimbabwe (2002-03)	Kurewa,2010 [128]	As stated for this paper under <i>T. vaginalis</i> (infections were combined)
	Mwanza, Tanzania (2012)	Hokororo, 2015	As stated for this paper under syphilis (infections were combined)
Peripheral Malaria	Macha, Zambia (1989)	Larkin, 1991[337]	NR
	Chikwawa, Malawi (1993-94)	Verhoeff, 1998 [293]	NR
	Chikwawa, Malawi (1993-95)	Verhoeff, 1999 [294]	NR

Infection	Site, country (year)	First Author, Year of publication	Risk factors for infection
Peripheral malaria	Blantyre, Malawi (1997-98)	Rogerson, 2000 [141]	first attendance at a clinic in the dry season relative to the wet season (OR 0.70; 95% Cl, 0.62–0.79) and a younger age, < 20 years, (OR 0.97; 95% Cl, 0.95–0.99) were associated with the presence of parasitaemia but there was no significant difference in malaria prevalence with gravidity as a continuous variable (OR 0.96, 95% Cl, 0.90–1.01)
	Blantyre, Malawi (1997-98)	Rogerson, 2000 [88]	Gravidity had a significant influence on malaria infection. Peripheral parasite prevalence was similar in primigravidae and secundigravidae, 24.0% and 26.1% respectively but was lower in multigravidae (17.2%)
	Kigoma, Tanzania (NR)	Mnyika, 2000[295]	NR
	Blantyre, Malawi (2000)	Mankhambo, 2002[297]	NR
	Sofala, Mozambique (2003-04)	Montoya, 2006 [298]	NR
	Morogoro, Tanzania (2001-02)	Wort,2006 [142]	Adolescent primigravidae had higher parasite prevalence than adolescent secundigravidae (41.3% versus 28.1%, <i>P</i> > 0.05) and adult primigravidae (41.3% versus 31.5%, <i>P</i> = 0.007).
	Multicenter, Mozambique (2003- 04)	Brentlinger,2007 [299]	NR
	Southern Malawi (2004-05)	Van Den Broek, 2009[301]	NR
	Mongochi, Malawi (2003-04)	Rantala, 2010 [95]	NR
	Blantyre, Malawi (1997-2001)	Feng, 2010 [1]*	Bed net use protected from peripheral parasitaemia (aOR 0.47; 95% Cl, 0.37, 0.60)
	300 DHS geographic clusters, DRC	Taylor, 2011 [144]*	In a multivariate model only lower wealth (all categories of wealth below the highest quintile) was significantly associated with higher prevalence of parasitaemia ($P < 0.001$). No association between age or gravidity or area of residence and malaria infection was found.
	Sambizanga and Ingombota, Angola (2006-2007)	Valente, 2011[89]*	Women ≤ 18 years old were at a higher risk of infection than women ≥ 25 years (aOR 2.1; 95% CI, 1.1-4.2). Women who had self-reported malaria infection earlier in pregnancy (aOR 2.0; 95% CI, 1.0-3.9) were at a higher risk of infection. No association was found between parity or area of residence or the level of education and placental malaria infection.

Infection	Site, country (year)	First Author, Year of publication	Risk factors for infection
Peripheral malaria	Manhica, Mozambique (2003-04)	Serra-Casas, 2011 [143]	Multiparous women were at a reduced risk of peripheral infection
			compared to primiparous women (aOR 0.26; 95% CI, 0.13-0.53) and
			women who had received SP were had reduced risk of infection
			compared to women who had received placebo (aOR 0.30; 95% Cl, 0.14-
			0.63)
	Manhica, Mozambique (2003-	Mayor, 2012 [302]*	Prevalence of <i>P. falciparum</i> malaria, both by microscopy and RT-PCR,
	2005)		was equal across age and parity groups
	Luanda, Angola (2008)	Campos,2012 [145]	Age, parity and gestational age were not associated with malaria
			infection. Women who had previous malaria during pregnancy also had
			a higher risk of current malaria (aOR 4.86; 95% Cl, 1.45-16.25)
	Nchelenge, Zambia (2013)	Siame, 2015 [115]*	Ownership of at least one net in their household was not associated
			with malaria. Age was the only factor associated with malaria; women
			aged 30-34 were 64% less likely to have malaria compared to those in
			the reference group 14-18 years old (AOR 0.36; 95% CI, Not reported).
	Mansa , Zambia	Mace, 2012 [90]	NR
Placental malaria	North Kivu, DRC (NR)	Meuris, 1993[303]	NR
	Blantyre, Malawi (1997-99)	Rogerson, 2000[88]	Gravidity had a significant influence on malaria infection. Placental
			parasite prevalence was similar in primigravidae and secundigravidae,
			2.5% and 36.2% respectively but was lower in multigravidae (16.0%)
	Blantyre, Malawi (NR)	Rogerson, 2003 [104]	NR
	Kinshasa, DRC (2004)	Lukuka, 2006 [304]	No association between parity or IPTp-SP and placental malaria infection
	Chikwakwa, Malawi (2004-05)	Senga, 2007[305]	NR
	Muheza, Tanzania (2002-05)	Mwangoka, 2008 [306]	NR
	Muheza, Tanzania (2002-05)	Kabyemela, 2008 [146]	Iron deficiency (aOR 0.20; 95% CI 0.11-0.36) and multigravidity (aOR
			0.32; 0.15-0.66) were independently associated with a decreased risk of
			placental malaria. The following factors were not independently
			associated with placental malaria infection; Use of intermittent
			presumptive therapy, insecticide-treated bed nets and delivering during
			seasons of low malaria transmission

Infection	Site, country (year)	First Author, Year of publication	Risk factors for infection
Placental malaria	Blantyre, Malawi (1997-2001)	Feng, 2010 [1]	Number of sulphadoxine-pyrimethamine doses received correlated
			inversely with placental parasitaemia (aOR 0.79; 95% Cl, 0.68, 0.91).
			Bednet use protected from placental parasitaemia (aOR 0.41; 95% CI,
			0.31, 0.54).
	Luanda, Angola	Valente, 2011 [89]	Women ≤ 18 years old were at a higher risk of placental malaria
			infection than women \geq 25 years (aOR 2.2; 95% Cl, 1.1-4.5). Women
			who had self-reported malaria infection earlier pregnancy (aOR 1.7; 95%
			Cl, 1.1-2.8) and those residing in the peripheral areas of Luanda (aOR
			1.7; 1.0-30) were at a higher risk of infection. No association was found
			between parity or the level of education and placental malaria
	Manhica, Mozambique (2003-05)	Serra-Casas, 2011 [143]	Multiparous women were at a reduced risk of placental infection
			compared to primiparous women (aOR 0.30; 95% CI, 0.16-0.57) and
			women who had received SP had a lower risk of infection compared to
			women who had received placebo (aOR 0.42; 95% Cl, 0.22-0.63)
	Manhica, Mozambique (2003-05)	Mayor, 2012 [302]	NR
	Mansa , Zambia	Mace, 2012 [90]	NR
	Rafiji and Moshi, Tanzania	Moshi, 2015	Primigravidity (aOR 2.4; 95% Cl, 1.1-5.0; P = 0.025) and residing in a high
			transmission area (aOR 9.4; 95% C,I 3.2-27.7; P < 0.001) were significant
			risk factors for placental malaria. IPTp was associated with a lower risk
			of placental malaria (aOR 0.3; 95% C,I 0.1-1.0; P = 0.044).
			1
aUR; Adjusted odds rati	on; BV: Bacterial Vaginosis; HSV: herp	bes simplex virus; NR: Not reported	
Vaginal infection: Triche	moniosis bestorial vaginosis and car	and syphilis	
$\alpha_{\Delta nv}$ STI: HSV-type 2 tri	ichomoniasis, Dacteriai Vaginosis anu car	synhilis and HIV	
Any 511. 113V-type 2, th	chomoniasis, chiamyula, gonornoea	, sypiniis and the	

Appendix 2.1: Information Sheet

You are being invited to take part in a research study. It is important for you to understand why the research is being done and what it will involve before you decide. Please take time to read the following information carefully and discuss it with others if you wish. Feel free to ask us if there is anything that is not clear or if you would like more information. Take time to decide whether or not to take part. Please understand that participation in this study is voluntary and that you are free to withdraw at any time, without giving any reason and without your medical care or rights being affected.

About the study

The study is on the malaria and reproductive tract infections in pregnancy. The principle investigator in this study is a lecturer of Parasitology at the University of Zambia and a PhD student at the London school of Hygiene and Tropical medicine. The study will be conducted under the supervision of experienced epidemiologists from the both the London School of Hygiene and Tropical Medicine and The University of Zambia.

Malaria, curable sexually transmitted infections (STI) and non-sexually transmitted infections of the reproductive system of women put the health and lives of pregnant women and their babies at risk. All of these diseases are present in our communities. In Zambia women receive pills at antenatal care that are meant to prevent malaria. Women are also screened for syphilis and treated if necessary. This is all in the interest of protecting them and their babies. However, these are not the only diseases that are common among pregnant women. Other diseases also affect the lives and health of women and their babies. Through this study we hope to find out the burden of malaria and the other infections that affect pregnant women when they occur together.

Purpose of the project

We hope to use the information from the study to contribute to improved health policies to ensure that women receive appropriate testing and/or treatment of other infection that affect them and their babies. This study will also examine to what extent malaria and reproductive tract infections occur together among pregnant Zambian women. The Ministry of Health can use this information to make more effective recommendations for management of common infections that occur in pregnant women.

Procedure

The nurses/research assistants will collect some personal information and collect a blood sample from you by pricking your finger and another one by drawing blood from your arm. They will also collect a vaginal sample from you and you will be given the first dose of sulphadoxine-pyrimethamine for protection of you and your baby against malaria. You will be asked to return to the clinic after four weeks. It is very important that you return after four weeks for a second malaria screening and the second dose of sulphadoxine-pyrimethamine.

When you come back to have your baby, the nurses will get a small piece of tissue from your placenta (the extra tissue that comes out after the baby) and ask you a few more questions. A 2cm by 2 cm by 1 cm piece will be sampled from the placenta. The baby will be weighed only; no blood will be taken from the baby.

All blood, vaginal swabs and tissue samples will be labelled with an identification number, not your name. Your name and ID number will be recorded on a list kept only by the study coordinator. This list will be kept only to help us to report back to you on your results and to follow up with you if you are found to have any of the infections that we will test you for. After the study is over, this list will be destroyed.

Number of women required

We need about 1100 pregnant women to take part in this study and women will be recruited consecutively as they come for their first antenatal visit. We are inviting anyone who is willing to take part provided that:

- 1. The person is attending antenatal for the first time.
- 2. She has not taken antibiotics or antimalarial agents in the previous 4 weeks.
- 3. She is less than 32 weeks pregnant.

Risks or discomforts

We expect the risks for you being in this study to be rare. You may feel a sting when a vein is punctured or when your figure is pricked to obtain blood from you. If enough blood is not taken from one prick, a second prick may be done. We may also ask you a few personal questions and you may feel uncomfortable answering these questions. You may choose to not answer any questions that you do not feel comfortable answering.

Benefits

This study will benefit pregnant women in Zambia, their children. We intend to share the results of this study with the Ministry of Health so that they can have evidence for the need to improve their recommendations on management and control of malaria and other infections that put the lives of pregnant women and their unborn babies at risk. This study has the potential to contribute to the improvement of health care which pregnant women and babies born to them receive.

Confidentiality

Any information or samples that leave the health centre will have an identification number (ID) not your name. Your name and ID number will be recorded on a list kept only by the study coordinator. We intend to use this list to report back to you on your results so that you can get the required treatment. After the study is over, all forms and your blood samples will be destroyed. Only the final results of the study, without names, will be shared with others.

Cost/Payment:

Your participation in this study will not cost you any money. The only cost to participating in this study is your time. This should take up to an extra hour (in addition to period you will spend for the normal ANC activities) of your time today in order for us to collect samples and information about you. The project has no money to compensate you for the time you will spend for the study.

Persons to Contact

If you have any questions about this study, please contact the principle investigator at P. O. Box 33528, Lusaka or call mobile: "Mobile number for the principle investigator was inserted here". If you feel that you have been harmed as a result of being part of this study, please contact the University of Zambia Biomedical Research Ethics Committee at the following address:

University of Zambia Ridgeway Campus P.O. Box 50110, Lusaka, Zambia Telephone 260-1-250753

If you agree to take part in this study you will be asked to sign two copies of the consent form and you will keep one together with this information sheet.

Appendix 2.2: Eligibility screening form

If a potential participant's gestational age by last menstrual period and fundal height determination is less than 32 weeks please fill out this form.

- 1. Have you taken any antimalarial treatment in the past one month No [] Yes []
- 2. Have you taken any antibacterial medication in the past one month No [] Yes []
- 3. Are you willing to allow a member of our team to record your HIV test results. No [] Yes []

Indicate to a potential participant that the results will not be recorded with their name but with an identification number that they will be given if they decide to take part in the study

If the response to question 1 and to question 2 is "NO" and that to question 3 is "Yes" please ask for consent using the consent form.

Appendix 2.3: Consent Form

Patients ID Number for study:

(for official use)

Title of project: The Epidemiology of malaria and sexually transmitted infections and reproductive tract infections in pregnant women in Ndola district, Zambia

Name of researcher: Mrs. Enesia Chaponda Ngulube

Please tick appropriately

- 1. I confirm that I have read and understand the information sheet for the above study and have had the opportunity to ask questions. Yes [] No []
- 2. I understand that my participation is voluntary and that I am free to withdraw at any time, without giving any reason and without my medical care or rights being affected. Yes [] No []
- **3.** I understand that a vaginal samples and blood form my arm and from my finger will be collected from today. **Yes** [] **No** []
- I understand that a piece of my placenta will be taken from me at delivery Yes [] No []
- I understand that sections of any of my medical notes that are of interest to the project may be read by the researchers. I give permission for these individuals to have access to my records.
 Yes [] No []
- 6. I agree to take part in the above study Yes [] No []

Participant's signature Date
Thumb print (if unable to sign)

Name of person taking consent form (If different from principle investigator)

Signature

1 for participant; 1 for researcher

Appendix 2.4: Questionnaire Participant's ID Number: /___/__//___/

Demographics and Socio-economics

1. Date of interview:.....(dd/mm/yyyy)

Village/area of residence

Name of Health centre

- HIV status
 [0] Negative [1] Positive
- Are you on antiretroviral therapy?
 [0] No [1] Yes
- 4. What is your age?

Measure and record weight, height and blood pressure at first antenatal visit

- 5. Weight:
- 6. Height:
- 7. Systolic blood pressuremmHg
- 8. Diastolic blood pressuremmHg
- 9. What is your marital status?
 - [1] Single
 - [2] Married
 - [3] Separated/divorced
 - [4] Widowed
 - [5] Other, Please specify _____
 - [99] Not answered
- 10. What is the highest level of schooling you completed? [Indicate number of years as 1, 2, 3... etc. count should include repeated years and years spend in tertiary education where applicable]

11. What is your main source of income?

- [1] Working for wages
- [2] Selling items
- [3] Self-employed (other than selling items)
- [4] None
- [5] Other, please specify____
- [99] Unknown/not answered

12. What is your husband's main source of income?

[1] Working for wages

[2] Selling items

[3] Self-employed (other than selling items)

[4] None

[5] Other, please specify_____

[6] Not applicable (single/widowed)

Does your household have any of the following?

[1] Yes

[1] Yes

13. Television

[0] No [1] Yes

14. Electricity [0] No

15. Radio

[0] No [1] Yes 16. Freezer/Fridge

[0] No

17. What type of fuel does your household mainly use for cooking?

- [1] Firewood
 [2] Charcoal
 [3] Gas
 [4] Electricity
 [5] Other, please specify______
 [99] Not known
 18. What type of floor is in the house?
- [1] Earth/sand
 [2] Wooden/Bamboo
 [3] Vinyl/Tiles
 [4] Cement
 [5] Other, ______
 [99=] Not known
 19. What type of roof is on the main house?

Plastic/tarpaulin
 Thatch/grass
 Iron sheets/tiles/Asbestos
 Other _____
 Not known

B. Indoor residual spraying and bed net ownership

20. Do you have a bed net? (If not, go to question 23)

[0] No [1] Yes

21. Is your net factory treated (long lasting net)? If 'yes' go to question 22

[0] No [1] Yes [99] Unknown

22. If your net is not factory treated, when was the last time it was treated?

[1] Treated in the last 6 months
 [2] Treated 6-12 months ago
 [3] Treated more than 12 months ago
 [4] Never been treated
 [99] Unknown

23. Did you sleep under the bed net last night?

[0] No [1] Yes

24. Has your house been sprayed for malaria control in the last 12 months?

[0] No [1] Yes [99] Unknown

C. Medical History

- 25. Is this your first pregnancy? (If yes, enter 0 in question 26 and go to question 30) [0] No [1] Yes
- 26. If no, how many previous pregnancies have you had? (include all pregnancies resulting in live births, stillbirths, abortions/miscarriages) Number of previous pregnancies (1, 2, 3... etc) Pregnancies with twins count as 1 pregnancy.



- 27. Have you ever experienced a miscarriage (lost a baby before it could survive independently)? (Based on assessment by a medical practitioner)[0] No[1] Yes
- 28. Have you ever experienced a stillbirth (given birth to a dead child)? (Based on assessment by a medical practitioner)[0] No[1] Yes
- 29. Have you ever given birth to a premature baby? (Based on assessment by a medical practitioner)

[0] No [1] Yes

30. Have you ever experienced itching or/and burning session when urinating and/or vaginal discharge and/or had sores on your private parts during this pregnancy? (if 'no' go to 33)

[0] No [1] Yes

31. If your answer to 25 is 'yes', did you receive treatment? [0] No [1] Yes

- 32. Are you currently experiencing any of the symptoms mentioned in 30? [0] No [1] Yes
- 33. Have you been treated for malaria during this pregnancy so far?[0] No[1] Yes
- 34. Have you ever been treated for a sexually transmitted or reproductive tract infection this pregnancy

[0] No [1] Yes

D. Sexual behaviour

35. How old were you when you started your period (menstruation)? *Enter age in years. If unanswered/unknown enter 99*



36. How old were you when you had your first sexual encounter? Enter number of years. If unanswered enter 99



37. How many sexual partners have you had in your life time?

Enter number of partners. If unanswered enter 99



To be administered after delivery.

Patients ID Number: /___//___/

E. Birth characteristics and medical History

- 38. Have you been treated for malaria during this pregnancy?[0] No[1] Yes[99] Not sure
- 39. Did you take iron supplementation for blood building? [0] No [1] Yes [99] Not sure
- 40. Have you taken or used any medicines/gel for the treatment of an STI or itchiness during this pregnancy? If no go to 41[0] No[1] Yes[99] unknown/not sure
- 41. If yes in which trimester¹ was the medicine taken?
 - [0] First Trimester
 - [1] Second trimester

[2] Third trimester [99] Unknown

¹1st trimester = in the first 3 months of pregnancy

2nd trimester = in the 4th to 6th month after the last menstruation

3rd trimester = in the 7th to 9th month after the last menstruation

42. Indicate the name(s) of the drug(s) taken for STI/RTI treatment as recorded on the ANC card.

.....

Questions 43-60 to be filled in by or with the help of the attending nurse and patients records

43. Date of delivery (dd/mm/yyyy) /___/__//__/__/ /___/

- 44. Place of delivery
 - [1] Hospital
 - [2] Health centre/clinic
 - [3] Home
 - [4] Other.....

45. Who performed delivery?

- [1] Family member
- [2] Traditional birth attendant (TBA)
- [3] Midwife
- [4] Doctor
- [5] Other
- 46. Delivery induced/spontaneous
 - [1] Induced
 - [2] Spontaneous
 - [99] Not known

47. Type of delivery?

- [1] Vaginal
- [2] C-section
- [3] Assisted
- [4] Other

48. How was the baby born?

- [1] Alive
- [2] Dead
- [3] Unknown

49. Sex of baby

- [0] Female [2] Male
- 50. Birth weight:grams.
- 51. Delivery gestational age:weeks [For official use, to be recorded based earlier assessment by ultrasound]
- 52. Number of babies delivered
 - [0] Singleton
 - [1] Twins
 - [2] Triplets
- 53. Birth outcome
 - [0] Live birth [1] Stillbirth [99] Unknown

(If live birth, go to F)

- 54. For stillbirths, was the baby moving at the start of labour?
 - [0] No [1] Yes [99] Unknown *If yes go to F*
- 55. If the baby was not moving before delivery, how many days before delivery did the foetal movements stop?(*Enter number of days 1, 2, 3, 8 etc*)



F. Drug History

- 56. HIV status [0] Negative [1] Positive
- 57. Has the participant been on ART? [0] No [1] Yes [99] Unknown
- 58. Has the participants taken cotrimoxazole (commonly known as pyriton) during this pregnancy?

[0] No [1] Yes [99] unknown

- 59. Has the Participant taken Sulphadoxine-pyrimethamine (Fancidar) for malaria prevention during this pregnancy?
 - [0] No [1] Yes [99] Unknown
- 60. If yes to question 59 indicate the number of doses as indicated by participant/recorded on the participants ANC card/record.



Participant's ID number:	\\	
1. Date of enrolmer	t dd/mm/yyyy	
 Date of 28 day follow-up 	dd/mm/yyyy	
 Did the participa attend the 28 da follow-up 	t [0] No [1] Yes	//
 Was the participant lost to follow-up 	[0] No [1] Yes	//
5. If yes give reasor	Reasons [1] Moved [2]cannot be traced [3] withdrew consent/refused [4]Other (To give reason go to 6)	
6. If other reason	Other	·

Appendix 2.5: Study completion or drop out record form

Appendix 2.6: Laboratory Record/Reporting forms

F1: Blood smear record form

Participant's ID number: _______\

Date of collection: dd/mm/yy ____

Sample collected at day 0 Day 28 (Tick appropriately as indicated on slide)

ID number	Result	No. of parasites	No. of white blood cells	Parasites/µl of blood

F2: Record form for Chlamydia, trichomoniasis and gonorrhoea test results.

Participant's ID number: ________\

Organism	Test result
Chlamydia trachomatis	Positive Negative
Neisseria gonorrhoea	Positive Negative
Trichomonas vaginalis	Positive Negative

F3: Record form for bacterial vaginosis diagnosis.

Record form based on Nugent criteria

ID Number	Gram variable coccobacilli	Gram negative curved bacilli	Lactobacilli	Total score	Classification

F4: Placental Histology Results Reporting Form

ID NO	Parasite	Pigment	Inflammation	Pathology class

Organism	Primor na	mo	Sequence			
Organisin	Filler na	lile	Sequence			
Chlamydia	KL1		5'-TCCGGAGCGAGTTACGAAGA-3'			
trachomatis	KL2		5'AATCAATGCCCGGGATTGGT 3'			
Neisseria	Ngu1		5'-CAA CTA TTC CCG ATT GCG A-3'			
gonorrhoeae	Ngu2		5'-GTT ATA CAG CTT CGCCTG AA-3'			
	T\/2					
Trichomonas	103		5 - ATTGTCGAACATTGGTCTTACCTC -3			
vaginalis	TV7		5' – TCTGTGCCGTCTTCAAGTATGC –3'			
Placmodium	1 ST	rDluc E				
falciparum	reaction	TPIUS 5	5-CETUTUTUCETTAAACTTE-S			
jaloparati	reaction	rPlus 6	5'-TTAAAATTGTTGCAGTTAAAACG-3'			
	and	" E A L 1				
	Z					
	reaction	rfalz	5'-ACACAAIGAACICAAICAIGACIACCCGIC-3			
Sources of primer sequences:						
C. trachomatis- Schachter 1997 [193]						
N. gonorrhoeae- Chaundry 2002 [192]						
<i>T. vaginalis</i> - Kengne <i>et al.,</i> 1994 [197]						
P falciparum- Snounou et al., 1993 [201]						

Appendix 2.7: Names and sequences of primers that were used for detection of chlamydia, gonorrhoea, trichomoniasis and malaria infection

Appendix 2.8: Polym	nerase chain	reaction	parameters
---------------------	--------------	----------	------------

Organism	Reaction	Initial	Denaturation	Annealing	Extension	Final	No. of cycles
(gene)		Denaturation ⁰ C, Time	⁰C, Time	⁰C, Time	⁰ C, Time	Extension ⁰ C, Time	
<i>Chlamydia trachomatis</i> Chlamydial ribosomal plasmid	Single PCR reaction	95, 10 min	95, 10 min	62, 2 min	72, 1:30 min	72, 7 min	45
Neisseria gonorrhoea (Orf1)	Single PCR reaction	95, 10 min	95, 1 min	55 <i>,</i> 2 min	72, 1:30 min	72, 7 min	45
Trichomonas vaginalis (repeat DNA sequence)	single PCR reaction	95, 10 min	95, 1min	60, 2 min	72, 1:30 min	72, 7 min	45
Plasmodium falciparum (MSP2)	Primary and secondary	94, 2 min	94 for 45 sec	61.1, 45 sec	65, 1 min	65, 2 min	25
Plasmodium falciparum (Dhfr and Dhps)	Primary and secondary	94, 2 min	94, 45 sec	43.4, 45	65,1 min	65,2 min	25
Plasmodium falciparum small subunit ribosomal	primary	94,3 min	94,1 min	60, 2min	72, 2min	72, 7min	30
RNA	secondary	94,3 min	94, 1min	55,2min	72, 2min	72,7 min	30
MSP2-Merozoite surface protein 2 DHFR-Dihydrofolate reductase DHPS-Dihydropteroate synthase DNA- Deoxyribonucleic acid RNA-Ribonucleic acid							

Polymorphism		Primer	Sequence	
Pf DHFR		M1	5'-TTTATGATGGAACAAGTCTGC-3'	
	1 st	M5	5'-AGTATATACATCGCTAACAGA-3'	
	reaction			
<i>Pf</i> DHPS	reaction	R2	5'-AACCTAAACGTGCTGTTCAA-3'	
		R1	5'-AATTGTGTGATTTGTCCACAA-3'	
	2 nd	M3	5'-TTTATGATGGAACAAGTCTGCGACGTT-3'	
<i>PJ</i> DHFR	reaction	F/	5'-AAATTCTTGATAAACAACGGAACCTttTA-3'	
		F	5'-GAAATGTAATTCCCTAGATATGgAATATT-3'	
		M4	5'-TTAATTTCCCAAGTAAAACTATTAGAgCTTC-3'	
		К	5'-TGCTAGTGTTATAGATATAGGatGAGcATC-3'	
Pf		к/	5'-CTATAACGAGGTATTgCATTTAATgCAAGAA-3'	
DHPS		Ĵ	5'TGCTAGTGTTATAGATATAGGTGGAGAAagC-3'	
		L	5'-ATAGGATACTATTTGATATTGGAccAGGATTcG-3'	
		L/	5'-ATTACAACATTTTGATCATTCgcGCAAccGG-3'	
Source of primer sequence: Duraisingh 1998 [203], Pf-Plasmodium falciparum				

Appendix 2.9: Primer names and sequences for detection of DHFR and DHPS mutations
Appendix 2.10: Restriction fragment length polymorphisms analysis of DHFR and DHPS polymorphisms

Appendix 2-	10a RFLP analy	sis of DHFR po	olymorphisms	
Substrate	Codon	Enzyme	RFLP	Residue
M3-F/	DHFR 16	NlallI	93bp, 376bp	Ala
			146bp, 376bp	Val
	DHFR 51	Tsp509I	120bp, 154bp	Asn
			120bp, 218bp	lle
	DHFR 108	Alul	196bp, 326bp	Ser
			522bp (no cut)	Other
		Bsrl	190bp, 322bp	Asn
			522bp	Other
		BstNI	196bp, 326bp	Thr
			522bp	Other
F-M4	DHFR 59	Xmnl	137bp, 189bp	Cys
			137bp, 163bp	Arg
	DHFR 108	Alul	119bp, 180bp	Ser
			299bp	Other
		Bsrl	146bp, 180bp	Asn
			326bp	Other
		BstNI	145bp, 181bp	Thr
			326bp	Other
Appendix 2-	10b RFLP analy	sis of DHPS p	olymorphisms	
Substrate	Codon 436	Enzyme Mnll	RFLP	Residue
К-К/			121bp, 278bp	Ser
			121bp, 317bp	Other
		MspA1I	406bp	Ala
			438bp (no cut)	Other
	437	Mwol	387bp	Ala
		Avall	419bp	Other
			404bp	Gly
	5.40		438bp	Other
	540	FOKI	85bp, 320bp	Glu
	420	(1):	4050p	Other (Lys)
J-K/	436	Hindili	4060p	Phe
			4380p	Other
		Hnai	4060p	Ala
L-L/	581	BstUI BslI	4380p	Other
			1050p	Ald
			1380p	Other
			1280p	Gly
	613	Mwol	10100 (110 cut)	Other
			1280p	Ald
		BsaWI	131pp	Cor
			161bp	Othor
		Agel	128pp	
			1200µ	Othor
			τοτομ	Uther

Appendix 2.11: Sequences of oligonucleotide primers that were employed for the differentiation of recrudescence from re-infection

Merozoite Surface Protein 2: msp2				
Reaction	Primer name	Sequence		
First reaction:	M2-OF	5'- ATGAAGGTAATTAAAACATTGTCTATTATA -3'		
	M2-OR	5'- CTTTGTTACCATCGGTACATTCTT -3'		
Second reaction:				
FC27-type	M2-FCF	5'- AATACTAAGAGTGTAGGTGCARATGCTCCA -3		
	M2-FCR	5'- TTTTATTTGGTGCATTGCCAGAACTTGAAC -3'		
IC3D7-type	M2-ICF	5'- AGAAGTATGGCAGAAAGTAAKCCTYCTACT -3'		
	M2-ICR	5'- GATTGTAATTCGGGGGGATTCAGTTTGTTCG -3		
Source: Snounou et al, 1999 [205]				

Appendix 2.12: Letter of notification of Ethical Clearance



THE UNIVERSITY OF ZAMBIA

BIOMEDICAL RESEARCH ETHICS COMMITTEE

Telephone: 260-1-256067 Telegrams: UNZA, LUSAKA Telex: UNZALU ZA 44370 Fax: + 260-1-250753 E-mail: unzarec@unza.zm Assurance No. FWA00000338 IRB00001131 of IORG0000774 Ridgeway Campus P.O. Box 50110 Lusaka, Zambia

4th September 2013

Your Ref: 004-02-13

Enesia Chaponda Ngulube, University of Zambia Department of Biological Sciences P.O. Box 32379 Lusaka

Dear Ms Ngulube,

RE: APPROVAL OF RESEARCH PROPOSAL: "THE EPIDEMIOLOGY OF MALARIA, SEXUALLY TRANSMITTED AND REPRODUCTIVE TRACT INFECTIONS AND THEIR CO-INFECTION AMONG PREGNANT WOMEN IN A CATCHMENT AREA IN NCHELENGE DISTRICT, ZAMBIA."

The above research proposal was resubmitted to the University of Zambia Biomedical Research Ethics Committee on 21^{st} June, 2013 with recommended changes. The proposal is approved.

CONDITIONS:

- This approval is based strictly on your submitted proposal. Should there be need for you to modify or change the study design or methodology you will need to seek clearance from the Research Ethics Committee.
- If you have need for further clarification please consult this office. Please note that it is mandatory that you submit a detailed progress report of your study to this Committee every six months and a final copy of your report at the end of the study.
- Please note that when your approval expires you may need to request for renewal. The request should be accompanied by a Progress Report (Progress Report Forms can be obtained from the Secretariat).
- Ensure that a final copy of the results is submitted to this Committee.

Yours sincerely, in Dr. J.C Munthali

CHAIRPERSON

Date of approval:

4th September, 2013

Date of expiry: 3rd September 2014